Coventry University



DOCTOR OF PHILOSOPHY

Development of a Conceptual Framework for Enhancing Payment Practices in the UK Construction Industry

Swai, Laura

Award date: 2022

Link to publication

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- · Users may download and print one copy of this thesis for personal non-commercial research or study
- This thesis cannot be reproduced or quoted extensively from without first obtaining permission from the copyright holder(s)
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Development of a Conceptual Framework for Enhancing Payment Practices in the UK Construction Industry



by

Laura Lazaro Peter Swai

PhD, Civil Engineering, Architecture and Building

August 2021

Development of a Conceptual Framework for Enhancing Payment Practices in the UK Construction Industry

A thesis submitted to the School of Energy, Construction and Environment, Faculty of Engineering, Environment and Computing, Coventry University in partial fulfilment of the requirements for the degree of Doctor of Philosophy in Civil Engineering, Architecture and Building

August 2021





Certificate of Ethical Approval

Applicant:

Project Title:

Laura Swai

Development of a Framework to Minimise Unfair Payment Practices in the UK Construction Industry

This is to certify that the above named applicant has completed the Coventry University Ethical Approval process and their project has been confirmed and approved as Medium Risk

Date of approval:14 Dec 2020Project Reference Number:P115954



v

ACKNOWLEDGEMENTS

First, I want to thank God for his loving kindness, grace and wisdom to complete this journey. Second, thanks to my parents Arch Lazaro Peter and Domina Lazaro for supporting me financially and morally in all my inspirations and pursuits to achieve my dreams. Third, to my one and only sister Dr Linda Lazaro Peter, thanks for being an exemplary model and mentor in my life. You have always believed and pushed me to never let my guard down nor settle for less. Thank you, Dada.

Also, my heartfelt thanks go to my dearest husband Martin George Kinyaha, for his love and support in this journey. You have been my number one supporter and problem solver in whatever situation or task I experienced during this course. Thanks love.

Finally, I wish to express my sincere thanks to my PhD Director of Study (DoS) and other members of my supervisory team. I want to specifically thank my DoS Dr Andrew Oyen Arewa for his constructive feedback and unwavering support during the study. Indeed, you challenged my thinking, sharpened my learning skills, inspiration and guided me toward completion of my PhD study. Thank you very much and keep on doing God's work.

Abstract

Globally, unfair payment practices remain a major commercial issue for many industries. The problem is exacerbated in the construction industry with huge negative impacts on contractors and other supply chains. For example, in 2016 there were over £30 billion worth of unpaid invoices to Small and Medium-Sized (SMEs) construction contractors in the UK. Also, recent report revealed that in the last three years, £7.8 billion of retention monies deducted by construction clients were not returned. A separate UK government report claimed that 72% of construction contractors are often compelled to sign up to contracts with disparities in rates of items, prolonged delays to payment periods, the imposition of rates and other prejudicial payment practices. Indeed, the industry's multi-tiered structure together with the commercial bargaining position of clients makes contractors and other supply chains in the industry susceptible to unfair payment practices. Moreover, the problem is endemic and chronic in the construction sector despite various government regulations and private initiatives designed to alleviate the problem. Yet, there is little research on how to enhance fair payment practices in the construction sector. Therefore, the aim

how to enhance fair payment practices in the construction sector. Therefore, the aim of this study is to develop a framework for enhancing fair payment practices in the UK construction industry.

The key research questions are: How endemic are unfair payment practices in the UK construction industry? How effective is the proposed framework for enhancing fair payment practices in the construction industry? The study adopted concurrent mixed methods design; involving the use of archival data, questionnaire surveys and interviews with construction stakeholders in the UK. Data obtained were analysed and merged to provide better understanding of unfair payment practices in the construction industry. Data analysis techniques employed include descriptive statistics, Cronbach's alpha reliability, relative importance index, Pearson's correlation analysis, and content analysis.

Findings from the study show that Tier 2 clients accounted for 82% of unfair payment practices, while 13% and 5% of cases were linked to Tier 3 and Tier 1 clients, respectively. Indeed, Tier 2 has and does exert a strong commercial influence over their supply chain. Other findings reveal that the use of cash flow strategy, business model and the culture in the construction industry are the major causal factors of lingering unfair payment practices. The study also found that payment provisions in standard forms of contracts are often ignored and impaired for various reasons including payer attitudes, legal loopholes, current payment processes, the

use of adhesion contracts and weak bargaining powers. Moreover, the study also discovered that subcontractors and suppliers often find it difficult to challenge current unfair payment practices, because of client-contractor's relationships.

The study also reveals that unfair payment practices have snowball effect on construction supply chain; with direct consequences for business profit margins, wide ranging insolvencies and indirect effects leading to mental illness, stress and reputational damage for businesses.

Lastly, the study developed a framework to enhance fair payment practices for the construction supply chain. Validation of the framework by construction industry professionals revealed that it has the potential to enhance prompt payment of invoices from Tier 1 clients (private/public) to main contractors, subcontractors and other suppliers.

Glossary of Terms and Phrases

Cash flow	The word is used to denote money in and out of a business account.
Delay payment	Postponement of payment to a recipient or gross deviation from contractually agreed payment time frame.
Late Payment	Overdue or behind schedule payment to a recipient.
Payment	Remittance of what is due to a person or entity
FIDIC	International Federation of Consulting Engineers
JCT	The Joint Contract Tribunal
NEC	The New Engineering Contract
Project Bank Account	A ring-fenced account with the sole purpose of facilitating payment in construction project
ROCE	Return on Capital Employed
SFoC	Standard Forms of Contract
SPSSR	Special purpose payment system regulator
Subbies	Subcontractors
Supply chain	Chain of firms linked or formed by contracts
Tier 1 clients	Major construction clients (private or public)
Tier 2 clients	Main contractors acting as client to subcontractors or suppliers
Tier 3 clients	Subcontractors acting as clients to other sub subcontractor/suppliers
Trade credit	Deliver now and then pay later
Unfair payment practices	Terms or practices that deemed to be 'grossly unfair'

List of Appendices

Appendix A	Questionnaire Survey 1
Appendix B	Questionnaire Survey 2
Appendix C	Interview Survey
Appendix D	Participant Information Sheet
Appendix E	List of large (main) contractors' business performance in the UK construction industry
Appendix F	List of small and medium sized contractors (SME'S) business performance in the UK construction industry
Appendix G	Quantitative data analysis
Appendix H	Certificate of ethical approval
Appendix I	Questionnaire Validation Survey

List of Abbreviations

-	Asset Based Finance Association
-	British Woodworking Federation
-	The Department for Business, Innovation and Skills
-	The Department for Business, Energy and Industrial Strategy
-	Business-to-business
-	Business to customer
-	Crown Commercial Service
-	Construction Leadership Council
-	European Payment Report
-	Federation of Small Business
-	Federation of Master Builders
-	National Specialist Contractors Council
-	National Federation of Builders
-	Office National Statistics
-	Interim Valuation Date
-	Standard Forms of Contract
	- - - - - - - - - - - -

Table of Contents

Abstract	3
Glossary of Terms and Phrases	5
List of Appendices	5
List of Abbreviations	7
List of Figures	1
List of Tables	5
CHAPTER 1	B
Introduction	8
1.1 Introduction	8
1.2 Background to the research problem18	B
1.3 Context and Rationale of the Research 22	2
1.3.1 Government Context	4
1.3.2 Industrial Context	5
1.3.3 Academic Context2	B
1.4 Aim & Objectives	1
1.5 Research Questions. 3: 1.5.1 Study Hypothesis 3:	1 2
1.6 Research Method	2
1.7 Contribution to Knowledge	2
1.8 Scope of the Study	4 4 6
1.9 Summary	9
2.1 Introduction	D
2.2 Composition of the UK Construction Industry	D
Figure 2.2: Composition of the UK Construction Sector (adapted Rhodes 2019)4	1
2.2.1 Size and Distribution of Firms4	1
2.2.2 SMEs42	2
2.2.3 The Construction Supply Chain44	1
2.3 Insolvency Risk	5
2.4 Payment Forms and Related Issues 4	5
2.4.1 Significance of Payments to Contractors4	7
2.4.2 Payment Mechanisms44	B
2.5 Types of Unfair Payment Practices	D

2.6 Eu	ropean Perspective – Unfair Payment Practices	52
2.7 His	storical Perspective – Unfair Payment Practices	54
2.8 Ca	uses of Unfair Payment Practices	55
2.9 Cu	ltural (or Attitudinal) Factors	57
2.10 In	ndustrial Factors	60
2.11	Technical factors	62
2.12	Regulator Factors	64
2.13	Role of Cashflow in Unfair Payment Practices	65
2.14	Effects of Unfair Payments Practices	67
2.15	Unfair Payments and Construction Business Performance	68
2.16	Summary	71
СНАРТ	TER 3	72
Contra	actual, Legislative and Regulative Measures to Payment Practices	72
3.1	Introduction	72
3.2 Ov	erview of Contract theories	72
3.2. 1	1 Historical Perspectives on Standard forms of Contracts	.74
3.2.2	2 Payment provisions in JCT Standard Building Contracts	.77
3.2.3	3 Payment Provision in NEC 4	.78
3.2.4	4 Payment provision in FIDIC	.79
3.2.5	5 Retention and Unfair payment practices in Construction	.81
3.2.6	5 Procurement and Payment Practices	.82
3.3 Leg	gislative Provisions	82
3.3. 1	1 Payments under the Housing Grants, Construction and Regeneration Act (1996)	.85
3.3.2 (LDE	2 Payments under the Local Democracy, Economic Development and Construction Act EDC)	.86
3.3.3 2011	3 Payments under the Scheme of Construction Contracts (England and Wales) Regulations	; .87
3.3.4	4 Public Contracts Payment Regulation 2015	.88
3.4	Construction Payment Charters and Codes	88
3.4. 1	1 The Prompt Payment Code	.88
3.4.2	2 The Fair Payment Charter	.89
3.4.3	3 The Construction Supply Chain Payment Charter	.90
3.5	Alternative Measures of Enhancing Fair Payments in Construction	91
3.6	Potentiality of Emerging Technologies in Minimising Late Payments	92
3.7	Contractual Relationships and Unfair Payment Practices	93

3.8	Summary	96
СНА	APTER 4	
Res	search Methodology	
4.1	Introduction	
4.2	Research Philosophy	
4.3	Research Paradigms	
4	I.3.1 Post-positivist Paradigm	99
4	I.3.2 Interpretivist Paradigm	
4	I.3.3 Pragmatic Paradigm	
4	I.3.4 The Transformative Paradigm	
4.4	Research Approach	
4.5	Research Strategy and Time Horizons	
4.6	Research Method	105
4	I.6.1 Quantitative and Qualitative Methods	
4	I.6.2 Mixed Method	
4.7	Justification for the Chosen Method	
4	I.7.1 Quantitative Method of the Main Study	
	4.7.1.1 Questionnaire Design	
	4.7.1.2 Piloting of the Questionnaire	
	4.7.1.4 Participant selection and sample	
	4.7.1.5 Data collection and procedures	
4	I.7.2 Qualitative Method for the main study	
	4.7.2.1 Interview design and measure	
	4.7.2.2 Participant selection and the sample	
	4.7.2.3 Procedure and data collection	
	4.7.2.4 Equipment	
4	I.7.3 Archival data method for the study	
4.8	Data Analysis	
4	I.8.1 Quantitative data analysis	
4	I.8.2 Qualitative data analysis	
4	<i>I.9</i> Reliability and Validity	
4.10	0 Ethical Considerations	119
	4.10.1 Consent of the Participants	
	4.10.2 Privacy and Confidentiality	
4.1	1 Summary	120
Qua	antitative Data Analysis	121
5.1	Introduction	121

5.2 Questionnaire Description	121
5.3 The Wilcoxon Signed Rank Test for Data Homogeneity	122
5.4 Test for reliability (Cronbach's Alpha)	125
5.5 Descriptive statistics	126
5.6 The Skewness Test	128
5.7 Profile of the study participants	129
5.8 Quantitative Analysis of the study's objectives	131
5.8.1 Objective 1- To evaluate the robustness of payment procedures in various standar of contract used in the construction industry	d forms 131
5.8.2 Objective 2- To determine the magnitude (degree) of unfair payment practices am tiers of construction clients	iong 132
5.8.3 Objective 3 -To investigate the contemporary causes of unfair payment practices in construction industry	n the UK 135
5.8.4 Objective 4- To ascertain whether unfair payment practices influence contractor business performance	rs' 138
5.8.5 Relationship between Unfair Payment practices and Construction Business Performance	139
5.8.6 Objective 5- To evaluate the role of the client-contractor relationship regarding payment practices	unfair 144
 5.8.7 Measurement of Supplementary factors about unfair payment practices	146 146 147 148
5.9 Results and findings	149
5.10 Summary	151
CHAPTER 6	152
Archival Data Analysis	152
6.1 Introduction	152
6.2 Archival data collection for construction clients' payment performance and o studies	case 152
6.3 Payment performance of different tiers of construction clients and contractors	153
6.4 Contractors' business performance using Archival Data	158
6.4.1 Operating Profit Margin of a Contractor's Business (Tier 3 And 4 Clients')	159
6.4.2 Return on capital employed for a contractor's business (Tier 3 and 4 clients)	161
6.4.3 Liquidity ratio of a contractor's business (Tier 3 and 4 clients)	
6.4.4 Debt ratio for a contractor's business (Tier 3 and 4 clients)	162 c
o.4.5 Statistical analysis for the time frame in nonouring involces and construction busines	s 162
6.4.6 Relationship between time frame in honouring invoices and construction business perfo (Tier 2 clients)	rmance
	167

6.5 Case Studies	172
6.6 Results and Findings	176
6.7 Summary	176
CHAPTER 7	177
Qualitative Data Analysis - Semi-Structured Interviews	177
7.1 Introduction	177
7.2 Overview of the participants' profile and the Population Sample	177
7.3 Brief description of the interview data and analysis	178
7.3.1 Validity of the findings	178
 7.4 Qualitative findings 7.4.1 General views of study participants concerning unfair payment practice 7.4.2 Perceptions of study participants to unfair payment issues in the construction industry. 	179 179 180
7.4.3 Views of study participants on clients' indifference to unfair payment practices	182
of contracts (SFoCs) and other legislative measures	182
construction clients 7.4.6 Participants views on unfair payment practices effects on a contractor's business performa	183 ince
7.4.7 Participants' views regarding the role of the client-contractor relationship on unfair payn practices in the UK construction industry	184 nent 185
7.5 Results and findings	186
7.6 Summary	187
CHAPTER 8	188
Discussion	188
8.1 Introduction	188
8.2 Research Findings	188
8.2.1 Robustness of payment procedures in various standard forms of contracts	189
8.2.2 The Magnitude of Unfair Payment Practices	192
8.2.3 Contemporary Causes of Unfair Payment Practices	194
8.2.4 Unfair payment practices influence a contractor's business performance	196
8.2.5 Role of the Client-Contractor Relationship	198
CHAPTER 9	201
Development of the Framework	201
9.1 Introduction	202
9.2 Functionality of the proposed framework	202
9.3 Benefits of the proposed framework	203
9.4 Validation of the proposed framework	206

9.5 Administration of the proposed framework	209
9.6 Limitations of the framework	209
CHAPTER 10	211
Conclusion and Recommendations	211
10.1 Conclusion	211
10.2 Limitations	214
10.3 Recommendations	215

List of Figures

Figure 2.2 Composition of the UK Construction sector	41
Figure 2.2.1: Number of Construction firms in UK 2018	42
Figure 2.2.3 Schematic Representation of Construction Supply Chain	45
Figure 2.3: Total new company insolvencies in England and Wales	46
Figure 2.4.1 Contractor's incoming and outgoing payments	48
Figure 2.6 Percentage of agreement to longer payment terms by UK companies	53
Figure 2.13 Typical S-Curves of Contractors' income versus expenditures	67
Figure 3.2.1 Most used SFoCs in UK construction Industry	76
Figure 3.2.2 Payment Timeline – JCT Design and Build Contract 2016	78
Figure 3.2.3 Payment Timeline – NEC4 Contracts	79
Figure 3.2.4 Payment Timeline – FIDIC Contracts	80
Figure 3.6 Nature of Client-Contractor Relationship	96
Figure 4.1 Research Onion Concept	97
Figure 4.6.2: Mixed-method design	107
Figure 5.6a Pictorial view of skewness for Questionnaire 1	128
Figure 5.6b Pictorial view of skewness for Questionnaire 2	128
Figure 5.8.1: Perception of study participants on SFoC mechanisms in protecting contractors against unfair payment practices	132
Figure 5.8.2.1 Magnitude of unfair payment practices among tiers of construction clients	133
Figure 5.8.2.2 Percentage commonality of unfair payment practices among construction client	134
Figure 5.8.2.3 Average duration in days to receive payment from client's	135
Figure 5.8.4 Unfair payment practices influence contractors' business performance	138
Figure 5.8.5.a: Scatter diagram comparing values of percentage agreement of study participants on unfair payment practices to average company's profit margin (Column C v Column B)	141
Figure 5.8.5.b: Scatter diagram comparing values of percentage agreement of study participants on unfair payment practices to average contractor's ROCE (Column D v Column B)	142
Figure 5.8.5c: Scatter diagram comparing values of percentage agreement of study participants on unfair payment practices to average contractor's liquidity ratio (column (Column E v Column B)	143
Figure 5.8.5d: Scatter diagram comparing values of percentage agreement of study participants on unfair payment practices to average contractor's debt ratio (Column F v Column B)	144

Figure 5.8.6 Role of client-contractors' relationship on unfair payment practices	145
Figure 5.8.6b Clients' perception of unfair payment practices	146
Figure 5.8.7.2 Disparity in commercial bargaining power between parties	148
Figure 5.8.7.3 Potentiality of emerging technologies in minimising unfair payment practices	149
Figure 6.4.5a: Scatter diagram showing relationship between delayed time frame in honoring invoices Vs average business's operating profit margin (Column B v Column C)	164
Figure 6.4.5b: Scatter diagram showing relationship between delayed time frame in honoring contractors' invoices and average business's capital employed (Column B v Column D)	165
Figure 6.4.5c: Scatter diagram showing relationship between delayed time frame in honoring contractors' invoices and average business's liquidity ratio (Column B v Column E)	166
Figure 6.4.5d: Scatter diagram showing relationship between delayed time frame in honoring invoices and average business's debt ratio (Column B v Column F)	167
Figure 6.4.6a Scatter diagram showing relationship between delayed time frame in honouring contractors' invoices Vs average business's operating profit margin (Column B v Column C)	170
Figure 6.4.6b: Scatter diagram showing relationship between delayed time frame in honoring invoices and average business's capital employed (Column B v Column D)	171
Figure 6.4.6c: Scatter diagram showing relationship between additional days of honoring invoices and average business's liquidity ratio (Column B v Column E)	171
Figure 6.4.6d: Scatter diagram showing relationship between delayed time frame in honoring invoices and average business's debt ratio (Column B v Column F)	172
Figure 9.1 Proposed Conceptual framework for enhancing Payment Practices in the UK construction industry	205
Figure 9.2 Modified Conceptual framework for enhancing Payment Practices in the UK construction industry	208

List of Tables

Table 1.5: Research questions identified in the study	32
Table 2.5: Types of Unfair Practices and their commonality to construction clients at risk	52
Table 2.6: Average payment terms across European countries (adapted from European	
Payment Report 2019)	53
Table 3.3: Different payment legislations and acts enacted on different countries	84
Table 4.3.4a: Paradigm language commonly used in research	101
Table 4.3.4b: Research paradigm, methods and tools	102
Table 4.5: Research strategies and their corresponding purpose and method	104
Table 4.7: Flowchart of the procedures in implementing concurrent mixed method in the study	108
Table 4.7.1.3: Questionnaire measurement	111
Table 5.2: Total number of valid questionnaires received from each method	122
Table 5.3.1: Total perception of study participants to questionnaire 1	123
Table 5.3.1.1: Wilcoxon signed rank test on questionnaire data collected- survey1	124
Table 5.3.2: Total perception of study participants to questionnaire survey 2	124
Table 5.3.2.1: Wilcoxon signed rank test on questionnaire data collected- survey 2	125
Table 5.4.1a: Cronbach alpha test on questionnaire 1 data collected	125
Table 5.4.1b: Cronbach alpha test on questionnaire 2 data collected	125
Table 5.4.2: Interpretation of Cronbach's alpha reliability test values	126
Table 5.5a Descriptive Statistics for Questionnaire 1	126
Table 5.5b: Descriptive Statistics for Questionnaire 2	127
Table 5.7: Profile of study participant	130
Table 5.8.1: Percentage agreement to payment procedures and provisions in	
standard forms of contracts	131
Table 5.8.3.1: Group ranking and mean of causes of unfair payment practices	136
Table 5.8.3.2: Relative Importance Index of contemporary causes of Unfair payment practices	137
Table 5.8.4: Percentage agreement to whether unfair payment practices influence contractors'business performance	138
Table 5.8.5: Summary between participant's agreement on unfair payment practices contractor's business performance over five years period	and 140
Table 5.8.7.1: Effect of unfair payment practices	147
Table 6.3.1: Payment performance for Tier 1 Main (private) construction clients to contractors (adapted from Build UK 2018/2019)	153

Table 6.3.2: Payment performance; Tier 1 public clients acting main construction clients to contractors (adapted from Build UK 2018/2019)	154
Table 6.3.3: Payment Performance; Tier 2 Clients (Main contractors) to subcontractors (adapted from Build UK 2018/2019)	156
Table 6.3.4 Payment Performance; Tier 3 Clients (Subcontractors) to sub-subcontractors suppliers (adapted from Build UK 2018/2019)	and 157
Table 6.4.1: Summary of time frame for honouring invoices and average construction contractors' business performance over five-year period	161
Table 6.4.5 Interpretation of correlation coefficient (adapted from Higgins 2003)	167
Table 6.4.6 Summary of time frame for honouring contractors' invoices and average construction contractors' business performance over five-year period	169

Table 6.5 Summary of Construction Clients: nature, turnover, outstanding payments, businessperformance, average number of days for payment and causes for going out of business175

CHAPTER 1 Introduction

1.1 Introduction

This chapter highlights the background to the study problem. The chapter provides an overview of the research problem, its background and context, the methodology used and its contributions to knowledge. The chapter is structured into seven sections: section one presents the context and rationale of the study. Section two presents the aim, objectives and research questions of the study, the third, fourth, fifth and six sections present the methodology used, the contribution to knowledge, the scope of the study and the structure of the thesis, while section seven present definitions of key terms and concepts.

1.2 Background to the research problem

Globally, unfair payment practices such as disparities in rates for items, delayed payments to suppliers and unpaid retentions to contractors, etc, remain a lingering commercial issue in the construction industry. The problem is exacerbated by many known and unknown factors; with huge negative consequence for contractors and other supply chains in the construction sector (The National Audit Office (NAO) 2018). In the UK, issues of unfair payment practices cut across strata and hierarchies of supply chains. For example, the Euler report (2015) asserted that there had been a substantial increase from 18% to 27% in the number of latepayment cases in the UK construction industry; with over £30 billion in unpaid invoices to Small and Medium-Sized (SMEs) contractors. Moreover, 82% of the total unpaid invoices were monies owed to 'subbies' (subcontractors) from different tiers of construction clients. The Federation of Small Businesses (FSB 2018) claimed that "over the past two years, a third of subbies in the UK construction industry have experienced longer payment periods of six weeks and beyond". Arguably, small and medium-sized (SMEs) contractors (also known as subbies) are disproportionately affected by unfair payment practices; despite various industry and government initiatives designed to curb them.

For example, the Department for Business, Innovation and Skills (BIS 2015) report titled "*Challenging Grossly Unfair Payment Terms and Practices*" identified a catalogue of common unfair payment practices in the UK construction industry to include late payments, "pay when/if paid", disparities in rates for items, the exclusion of provisional remedies, delays to payments, longer payment terms, flat fees, discounts for prompt payments, the imposition of rates on subcontractors and unpaid retentions to contractors. Sadly, the identified unfair payment practices stated above are typical and endemic in the construction industry.

Peter and Arewa (2018) argued that late-payment predicament is not peculiar to SMEs alone; the main contractors in the industry also experience similar problems. Recent payment data published by the Department of Business, Energy and Industrial Strategy in the UK revealed that most private clients had standard payment terms of 90 days and a maximum of 120 days compared to the industry standard of 28 to 30 days payment agreements. The implication of such a protracted payment duration is that both main contractors and subcontractors are exposed to high business costs due to unfair payment practices that go beyond the agreed payment terms.

Late payment is perhaps the most common type of unfair payment practices for construction contractors. It is worsened by the nature and culture of the construction sector; characterised by fragmentations, complexities, multi-layered hierarchy contracts, diverse range of specialist skills and multifaceted works. Moreover, the use of a cascade system of payments where the payment flows from the upper tiers (client and contractors) to the supply chain seems to compound the deep-rooted payment problems in the industry (Latham Report 1994).

Apart from late payment quandary, there are a host of other unfair payment practices that are subtle to detect yet widely utilised by some clients to defraud their contractors and suppliers. For example, disparities in rates for items, the exclusion of provisional remedies and the discounting of items are all gross terms usually imposed on contractors and suppliers to exclude them from utilising any sort of remedy in the contract. Consequently, these practices compel them to provide lower rates of payments to their clients. The Federation of Small Businesses (FSB 2018) claimed that 12% of construction contractors gave discounts to encourage prompt payments, 7% gave retrospective discounts to clients to facilitate their payments, clients paid 6% of their fees to remain on their chain in a preferred list of contractors and a 3% discount was provided to clients after goods and services had been supplied. While there is an overwhelming belief that unfair practices are unethical; they remain prevalent with little or no research on how to enhance payment practices in the construction industry.

Fundamentally, early payments to construction contractors is the lifeblood of the construction industry. Ameer (2005b) affirmed that early payments to contractors in the construction sector are important because of the large amount of money spent on purchasing materials, labour, plants and credit payments on a regular basis. Okeyo et al. (2015) opined that regular disbursements or timely payments are crucial to the supply chain, particularly for subbies, to ensure continuity and the completion of projects on time. Without regular payments for work done, it is practically impossible for most contractors, especially small and medium sized enterprises (SMEs), to complete works due to low capital investment. Besides, this category of contractors depends mainly on consistent interim payments to finance their operations and invest in future business growth.

Indeed, the greatest danger of unfair payment practices (especially late payment), is the existential risk to both large and small construction companies. For example, a recent survey by TSheets (2018) asserted that one in five contractors become insolvent annually as a result of cashflow problems which are directly linked to unfair payment practices. The Insolvency Service Report (2019) revealed that the bulk of construction insolvencies involving 3,013 companies registered in the UK emerged from unfair payment practices. The National Specialist Contractors Council (NSCC 2014) affirmed that "most construction firms are struggling to expand and invest in future development because of late and other unfair payment plagues particularly SMEs contractors". Ye and Rahman (2010) argued that the

timeliness of payment is important to circumvent failure risks that could endanger a project's viability, as well as the profitability of the business.

Outside the UK, the European Payment Report (2019) revealed that delayed payments continue to soar throughout European countries' economies; with twothirds (62%) of construction companies having their payments delayed due to unfair payment arrangements; thereby putting businesses at risk of insolvency. The report is similar to the European Commission Payment Directive (2011/7) that claimed "... tens of thousands of small and medium-sized businesses collapse annually in Europe as a result of late payments and other unfair commercial practices among businesses...". Similarly, the European Payment Report (2016) stated that "though signs of recession and debt risk are imminent in Europe; ... unfair payment practices are the most problematic commercial issue to companies". The demise of Carillion Plc in 2018 as the UK's second-largest construction and facilities contractor, illustrates the scale of the late payment epidemic as an ongoing and worsening issue among large and small businesses in the construction sector.

Moreover, Carillion Plc's recent demise is unique to understanding the scale of unfair payment practices in the UK construction industry. For instance, the company was a signatory to the UK government's prompt payment code, yet it had the reputation of being a notorious late payer in the industry. Cherry (2018), the Federation of Small Businesses chairman, noted that Carillion had its suppliers wait for 120 days to receive payments for monies owned to them. He affirmed that "...I wrote to Carillion back in July 2017 to express concern after hearing from FSB members that the company was making small suppliers wait 120 days to be paid...". Carillion's demise is estimated to leave a debt of £2 billion in outstanding payments owed to 30,000 small and medium-sized subbies and suppliers (Loxley 2018a). Indeed, lessons learnt from Carillion's collapse have highlighted the impact of unfair payments in the UK construction industry. The recent payment data released by Build UK (2019) under the new rules of "benchmarking industry payment performance "revealed that none of the construction industry's biggest contractors paid within an average of 30 days and on time. For instance, payment

data published by Build UK illustrate that Murphy Group Plc had the longest record of paying their supply chain with an average of 66 days, whilst Willmott Dixon had a minimal of 33 days (Price 2018).

Besides, within the sphere of payment practices in the construction industry, there seems be scepticism regarding clients/contractor's commitment to various fair payment initiatives (such as existing codes, charters and regulations etc.) designed to alleviate the problem. Although various studies about late payment exist; these studies often look at the causes and effects of unfair payment practices. Yet, there is little research concerning how to enhance fair payments within the construction supply chain.

1.3 Context and Rationale of the Research

Unfair payment practices to businesses is undeniably a global phenomenon and the issue is not limited to the UK alone. The Federation of Small Businesses (FSB 2019) claimed that "late payment is endemic to construction companies whether big or small". For instance, a recent publication by the Asset Based Finance Association (2016) illustrates "a sharp rise of 22% in late-payment related cases, with most construction companies having to wait 82 days for their invoices to be paid compared to 28 days stipulated in their contracts".

The findings further revealed that a contingent payment factor often associated with long supply chains within the sector usually intensifies unfair payment practices in the construction industry. The Construction Excellence Board (2016) stressed that construction consists of a high level of disaggregation and fragmentation within its supply chain, therefore constraining the contractors' ability to receive timely payments.

A survey conducted by NSCC (2014) claimed that "archaic payment practices contribute significantly to late-payment" dilemmas in construction. For example, the NSCC (2014) survey showed that 700 construction firms wrote-off £250 million in late payment and retentions debts in one year. This amount consists of 6.6% of

their turnover. The report further reveals that 4 out of 10 specialist contractors faced various degrees of unfair payment practices annually. There is reason to believe that small contractors bear a greater burden of late payment compared to large contractors. Connell (2014) argued that from an entrepreneur's perspective, SMEs tend to suffer tighter financial conditions and economic losses when exposed to late payments. For example, small scale construction businesses spend on average 130 hours each year chasing payments, at an average cost of £1,500 per business (Construction Excellence 2016). The report further stressed that this category of businesses (SMEs) incurred £180 million in debt regarding interest charges on loans used to execute work for clients.

Another dimension to unfair payment practices in the UK construction industry is unpaid retentions. For example, research by the British Woodworking Federation (BWF 2018) affirmed that £10.5 billion of an overall construction turnover of £220 billion is held in retention by clients as part of the contractors' funds. Furthermore, the BWF (2018) report asserted that over the last three years, £7.8 billion in retention monies had not been paid to subbies and suppliers within the supply chain. Parallel research by Pye Tait Consulting (2017) on behalf of the Department for Business, Energy and Industrial Strategy (BEIS) affirmed that half of contractors' payments for completed work had been held in retentions for the past three years, up to 2019. Recent findings by the Construction Enquirer (2019) confirmed that adverse payment practices were common in construction businesses with a total amount of £21.6 billion in retentions illegally withheld by Tier 1 clients from their supply chain.

On the other hand, the Federation of Master Builders (FMB 2017) revealed excessive payment terms and practices were hindering construction businesses. The report affirmed that 70% of small and medium-sized contractors were exposed to 'cowboy clients', that would seek for longer payment terms or delay payments to recipients on spurious grounds. Sarah McMonagle (2017) claimed that "nearly a quarter of construction SMEs usually wait for more than four months to receive payments from their clients". The European Payment Report (EPR 2019)

23

acknowledged that the UK had longer payment terms compared to other European countries and that there was no evidence to suggest the problem was abating.

1.3.1 Government Context

In the last two to three decades, the UK government has enacted various legislations, rules and acts to alleviate unfair payment practices in the construction industry. For instance, the Housing Grants Construction and Regeneration Act 1996 (HGCRA) and the Scheme for Construction Contracts Regulations 1998 were designed to enhance payment practices and outlaw conditional clauses in the UK construction industry.

Kumaraswamy (2011) described HGCRA as a benchmark and catalyst for spearheading the development of different legislations and acts such as the Local Democracy Economic Development (LDEDC 2009), the Late Payment of Commercial Debts Regulation (2013), the Small Business, Enterprise and Employment Act (2015) and the Public Contract Regulation (2015). Furthermore, the Construction Excellence (2016) report examined 19 separate payment initiatives (charters, codes and regulations) launched by the government to curb unfair payment practices and discovered that despite various government legislations and private initiatives, there was a significant increase in the number of cases relating to unfair payment practices in the construction industry. For example, the Construction Supply Chain Payment Charter, a privately managed agreement, set out 11 fair payment commitments for their members to adhere to; such as enhanced payment practices including a collaborative culture and a sustainable supply chain for projects. An investigation by Construction Excellence (2016) showed that cases of unfair payment practices were widespread among members of the Construction Supply Chain Payment Charter. These are clear indications that the problem of unfair payment practices defies various government and private initiatives designed to curb them.

Additionally, most standard forms of contracts (SFoCs) are designed to provide adequate mechanisms to ensure payments are made promptly within the

construction supply chain. Standard forms of contracts such as JCT and NEC consist of strict payment timelines and the provision of a 28 days payment period (that is from the date a contractor submits their interim payment application to the project's quantity surveyor or client's representative for payment).

Apart from rules and regulations, some UK government departments have the right to pursue legal claims against organisations that default in paying their business partners. For example, the UK Department for Business Innovation and Skills (BIS) has the power to prosecute organisations and individuals that practice late and other unfair payment practices across in public and private sectors. The BIS works with the Crown Commercial Service (CCS) to enhance prompt payment policies for governments and public sectors. Griffiths et al. (2017) claimed that one of the most significant features of government-specific initiatives to control unfair payment practices was the introduction of project bank accounts (PBAs). PBAs was introduced by the National Audit Office (2005) as a second security for payment measures following a raft of late payment problems and the initiation of different legislations by many local and international jurisdictions.

On the other hand, the UK government and the leaders of notable top private companies have voiced their concerns against the problems of unfair payment practices in the construction industry. For example, Kelly Tolhurst, the Minister for Small Businesses, Consumers and Corporate Responsibility (2018), orated that "though we have introduced a package of measures to build a culture change in payment practices ..." the recent collapse of Carillion has highlighted the continuing prevalence of unfair payments affecting small and large companies".

Rachel Reeves, the Chair of the Business, Energy and Industrial Strategy Committee (BEIS 2018) claimed that *"the problems caused by late payments from large contractors are well publicised and we know they can often have a devastating impact on small businesses … It is high time to get a better understanding of payment practices that have potential to enhance the old construction industry commercial problem … as well as understand how big firms* *treat their small suppliers who are the lifeblood of the UK economy.*" This statement and other government initiatives mentioned above are direct confirmation that the UK government has a strong desire to find sustainable solutions to deep-rooted unfair payment practices in the construction sector.

1.3.2 Industrial Context

Historically, the construction industry has a chequered history of unfair payment practices. Construction Excellence (2016) claimed that "most construction companies spend on average 130 hours each year chasing payment; at an average cost of £1,500 per business". Available statistics show that SMEs doing business in the construction sector are often victims of unfair payments (Connell 2014). Moreover, many industry stakeholders have spoken openly concerning grossly unfair payment terms, acknowledging the need to improve endemic unfair payment practices in the construction industry. For example, it could be argued that the scale of the late payment problem in the construction industry led to the formation of the Construction Leadership Council (CLC) in 2014; consisting of major UK clients and contractors, suppliers and other stakeholders. The CLC's primary objective was to proffer long-lasting solutions to late payments in construction. The CLC (2014) came up with a payment charter backed by the UK government stating that all contractors under the umbrella of the Council must be paid for work done within 30 days.

Though the intent and purpose of CLC (2014) is undeniably a superb idea; there is no substantial evidence to prove that payment practices have improved considerably in the construction industry. Richard Beresford (2018) claimed that "… late payment and retention of funds remain the biggest problem to the UK construction industry, despite widespread recognition that late-payment practices are unacceptable; … there is no sign that the problem is likely to subside in the nearest future …".

Conversely, Suzannah Nichols (2016) argued that although there is a likely temptation to blame clients for not paying contractors on time; most late payments cases in the construction industry are usually contractors' faults that stem from the technicalities of payment procedures or poor valuation exercises. Indeed, payment practices in the industry show that some contractors, especially the SMEs, rarely use experienced quantity surveyors or cost consultants to prepare their periodic valuations. Yet there have been attempts to ratify technical mistakes in some contractors' valuations; normally adding a week or more to the overall payment duration. Perhaps there is a need for industry practitioners to engage qualified cost and quantity surveying professionals to cross-check periodic valuations before onward processing by the clients.

Plimmer (2013) claimed that some industry practitioners and even stakeholders seemed to believe that unfair payment practices, particularly late payments, had become an acceptable norm in the construction industry. Arguably, one of the greatest challenges facing stakeholders is cynicism concerning the effectiveness and implementation of these government measures in addressing unfair payments.

Moreover 80% of business to business (B2B) transactions conducted within the UK are made under trade credit practices; with the business supplier agreeing to extend the payment period for goods and services (Paul and Wilson 2006). Trade credit is known as deferred payment on the basis of "deliver now and then pay later". However, the consensus argument concerning the impact of unfair payment practices on small firms is often centred towards whether government interventions through legislations and credit management practices help to alleviate the problem.

Moreover, the Association of Chartered Certified Accountants (ACCA 2015) are of the view that although most trade credit is an important source of short-term funding for SMEs compared to bank lending or other formal finances; nonetheless, not all credit-based transactions proceed smoothly given that often, the timing of the payment does not match the expectations of either suppliers or buyers, resulting in further delayed payments. In addition, there is little industry research about the role of the client-contractor relationship *vis-à-vis* unfair payment practices. Hinze and Tracey (1994) argued that little is known about the actual negotiated terms and amicable working relationships that exist between clients and contractors. Ashworth (2006), Russell (2007) and Chalker et al. (2016) all stressed that there was a need for further investigation into the client-contractor relationship, particularly between main contractors and subcontractors (MC-SC), asserting that MC-SC relationships were riddled with biased commercial intentions. They argued that this was based on the fact that most main contractors take advantage of subcontractors and suppliers by drafting contracts that are one-sided or non-negotiable 'take-it or leave-it' agreements.

1.3.3 Academic Context

There are notable academic works about unfair payment practices. For example, Cotter (2005) affirmed that unfair payments issues could be inadvertent; although the ripple effects were likely to affect the entire supply chain. Kenley (2002) suggested that unfair payment practices were deliberate strategies that could best be described as "collect the money early and delay passing the cash to the supplier chain". Most clients and contractors use this strategy to boost their liquidity or working capital. The Australian Procurement and Construction Council (1996) opined that multi-tiered hierarchical structures, together with cascade payment obligations, influenced unfair payments in the industry. Besides, Pettigrew (2005) argued that the industry culture of 'work first, get paid later' was another dimension to unfair payments.

Hughes' et al. (1998) publication about payment defaults in the construction sector asserted that the 'cannot' and 'would not' pay attitudes of payers largely influenced the existence of unfair payments to construction contractors. The authors further claimed that unpaid retentions had worse detrimental effects on a subcontractor's cashflow; although this is largely considered a risk protection mechanism for clients.

28

Furthermore, there are numerous academic works about commercial risks regarding insolvency due to unfair payment practices. For example, Latham's report (1994) titled 'Constructing the Team' stated that "the cascade system of payment in the construction industry normally follows a pattern of client to main contractor, main contractor to subbie and subbie to supplier or vendor ... this type of payment mechanism is a recipe for delayed payment and has potential to expose small contractors involved in the supply chain to insolvency". In terms of productivity and performance, Abdul-Rahman et al. (2009), Jiang (2012) and Auydha (2012) opined that timely payments to contractors contributed significantly to desirable project completion quality, cash flow positioning and the continuity of works on site.

Moreover, there is little research on whether timely payments influence the profitability and positive cash flow of contractors. From an academic point of view, there is little emphasis about contemporary causes of unfair payment practices, the scale (that is the magnitude) of the problem within the construction sector, the impact of unfair payments on contractors' profitability or development of an initiative with potentials to enhance fair payment practices.

Besides, many academic studies have proffered various solutions in the past; yet the problem of unfair payment practices seem to be increasing by leaps and bounds. Moreover, unfair payment practices are often generalised as a form of payment problem, late payments, under payments, retentions and nonpayments. For instance, Azman et al. (2013) stated that irregular cash flow influences the finances of most organisations, but the exact effect of irregular payments on a contractor's financial performance is yet to be established.

Kenyatta *et al.* (2016) affirmed that payment defaults in the form of late or under payments has adverse effects on contractors and their supply chains. Ramachandra (2013), in a published thesis, argued that the problem persisted because late payment risks adversely affected lower tiers in the construction supply chain. Mbachu (2011) and Ansah (2011) suggested that causes of unfair payment practices could be a self-made problem; that is attributed to client or contractor error. Hence, there is a need to examine the role of different construction clients, the robustness of standard forms of contract and the client-contractor relationship in persistent unfair payment issues in the construction sector.

Moreover, there is a need to investigate the magnitude of unfair payment issues in the construction industry because the literature differs considerably regarding the scale of the problem in the lower and upper structures of the construction supply chain. For example, the BIS Research Paper (2013) categorised the contracting relationship and its structure within the construction supply chain using "tiers classification". The term "Tier", as defined within the report, is a commercial relationship between clients, contractors, subcontractors and suppliers engaged in the delivery part of construction works. The literature review shows that most studies about unfair payment problems tend to generalise the client-contractor relationship from a single perspective, that is from the Tier 1 client (public or private) to the main contractor. However, Kenyatta et al (2016) argued that there is a need to understand unfair payment issues from different clients' perspectives to enhance fair payment practices.

Therefore, for better understanding of the research problem, this study seeks to examine issues of payments practices from three distinct construction clients' perspectives. **Tier 1 clients:** major clients that pay for high value construction deliverables e.g. government authorities, property developers, private institutions and other funders. **Tier 2 clients**: main contractors in the construction sectors acting as clients to subcontractors. **Tier 3 clients**: subcontractors working with main contractors/Tier 1 clients acting as client to other subcontractors.

1.4 Aim & Objectives

The aim of the study is to develop a conceptual framework for enhancing payment practices in the UK construction industry. To achieve the aim of the study, the following objectives were considered.

- 1. To evaluate the robustness of payment procedures in various standard forms of contract used in the UK construction industry.
- 2. To determine the scale of unfair payment practices between different tiers of construction clients.
- 3. To investigate contemporary causes of unfair payment practices in the UK construction industry.
- 4. To ascertain the relationship between unfair payment practices and construction companies' business performance.
- 5. To evaluate the role of the client-contractor relationship on chronic unfair payment practices in the UK construction industry.
- 6. To develop a conceptual framework for enhancing payment practices in the UK construction industry.

1.5 Research Questions

Based on the study aim and objectives, below are the key research questions:

- 1. What are the contemporary causes of unfair payment practices in the UK construction industry?
- 2. How effective is the proposed framework for enhancing payment practices in the UK construction industry?
1.5.1 Study Hypothesis

Table 1.5 below illustrate two sets of null and alternative hypotheses for the study.

S/No.	Null Hypothesis	Alternative Hypothesis
H1	There is no significant relationship between construction companies' business strategies and unfair payment practices.	There is a significant relationship between construction companies' business strategies and unfair payment practices.
H2	There is no significant relationship between unfair payment practices and contractors' business performance in the construction industry	There is a significant relationship between unfair payment practices and contractors' business performance in the construction industry

Table 1.5 Research hypotheses identified in the study

1.6 Research Method

The study intends to adopt concurrent mixed methods to collect both quantitative and qualitative data. The plan is to collect the data separately, analyse and then merge results to obtain a more comprehensive understanding of the research problem. Moreover, a documentary review of archival data and case studies will be retrieved by the researcher to produce new insights through subscribing to reputable commercial and construction payment sources such as Build UK and Companies House website. The study adopts an explanatory type of research to provide a meaningful and accurate picture of industry payment practice in the UK.

1.7 Contribution to Knowledge

This study contributes significantly to the commercial, finance and construction project management body of knowledge. The study explores the commercial strategies adopted by construction businesses and clients, the realities of unfair payment issues and their influence on business performance in construction. The study developed a novel framework to enhance efficient payment arrangements in the construction industry. In specific terms, the study contributes to the enhancement of "commercial best practices" between clients, contractors and other supply chains. This contribution is based on the study's novel drive to improve fair payment practices between client, main contractor and other supply chain.

The proposed framework has the potential to build a congenial atmosphere; with potentials to curb systematic commercial abuse that exists between clients, main contractors and subcontractors in the UK. Moreover, the study suggests ways to improve the current construction payment provisions and legislations by proposing strict legal measures to help curb the problem of unfair payment in the construction sector. Hopefully, the proposed framework will address chronic unfair payment practices ingrained in the construction sector.

Culturally, the study's contribution will reinvigorate the industry's poor image and practices that have existed for years. The prevalence of an unfair payment culture in construction will be minimised by the proposed framework that encourages the prompt and adequate payment of invoices from Tier 1 clients to subcontractors and suppliers in the project. Thus, the current cascade payment culture will be revolutionised towards a direct payment method to the entire supply chain. Moreover, the proposed framework seeks to remove conditional contract clauses such as "pay-when-paid" or "pay-if-paid" to improve the industry's culture.

The economic benefit of the study cannot be overemphasised because the proposed framework will enhance contractor cashflow, reduce long payment periods and inject needed transparency into the payment culture of the construction industry. Consistent cashflow improves financial health and business performance, with potential for job creation, more business opportunities, productivity, investment, innovation and economic stability. Moreover, sufficient cash drives a good and entrepreneurial spirit that enables both the diversification of business and a multiplier effect to other sectors. Moreover, the proposed framework will provide adequate cashflow to construction contractors that will help businesses generate more wealth and boost economic growth. Overall, the viability of construction businesses will help to alleviate chronic insolvency that is common to businesses in the sector.

The study is sustainable because the developed framework is capable of reducing time wastage (time organisations spend) in chasing payments and costs incurred due to unfair payment practices.

1.8 Scope of the Study

The study's scope covers the UK construction industry; commercial management, contract and construction project management, payment regulations, acts, payment mechanisms, business performance and relationship management between clients and supply chains. The researcher is aware that many studies have been carried out on late payment problems rather than unfair payment practices, which broadens the generalised context. Also, the study will provide a robust evaluation of contemporary factors that influence unfair payment practices and will contribute to the development of a holistic approach to enhance fair payment practices.

1.8.1 Structure of the Study

The study is divided into ten chapters. The structure of each chapter is summarised as follows:

Chapter 1 sets out the introduction and background of the study problem. The chapter provides the research aim and objectives, the research questions and hypotheses, insights into the research method, contribution to knowledge, scope of the study and brief definitions of key terms and phrases.

Chapter 2 provides an overview of the literature and fundamental knowledge and concepts related to the research study. This entails a review of the UK construction industry, payment trends, forms of payments, types of unfair payment practices, their causes and business performance in the construction industry.

Chapter 3 is an extension of the literature review chapter that examines existing legislations and regulative measures designed to curb unfair payment practices in the UK. The chapter also reviews contractual theories and clauses related to payment practices.

Chapter 4 provides details about the research methodology adopted for the study. The chapter provides the research philosophy, research design, population sample, justification for the chosen research method, data collection techniques and analysis, reliability, validity and ethical consideration of the study.

Chapter 5 presents quantitative data analyses obtained via questionnaire surveys and archive data from UK Companies House.

Chapter 6 presents the results of archival analyses based on documentary evidence obtained from Build UK, UK Companies House, case studies and other reliable sources.

Chapter 7 presents the qualitative data analyses and summary of the interviewees' perceptions of the study problem. All interviews in this chapter were based on designed semi-structured interview questions.

Chapter 8 provides a robust discussion concerning the findings from the quantitative, qualitative and archival analyses. The chapter also compared findings from the study with the reviewed literature.

Chapter 9 entails the development of a framework to enhance fair payment practices between clients, contractors and other supply chains in the UK construction industry. This chapter is divided into two sections: the development of the framework and its validation by construction professionals.

Chapter 10 highlights the conclusions and recommendations of the study based on the study's set objectives in Chapter 1.

1.8.2 Terms, Concepts and Definitions

Payment: refers to monies used for remuneration or compensation in exchange for services, goods, work done or performance for an obligation. Kokkola (2010) defines 'payment' as a transfer of funds which discharges an obligation on the part of a payer *vis-à-vis* a payee. Payment to a construction contractor is a monetary consideration for the performance of works or services given in the contract.

Late payment: the phrases "late" or "delay" payments are often used interchangeably in many literatures and studies. Late payment is defined as the failure of the paymaster to pay within the period of honouring the certificates as provided in the contract (Harris and McCaffer 2003). Late payment normally occurs when payment is delayed after its contractual due date or is paid outside of the agreed payment period.

Unfair payments: refer to anything that is a gross deviation from good commercial practice and is contrary to good faith and commercial dealings. It also includes the nature of the goods or services in question; and whether the purchaser has any objective reason to deviate from the stipulated/contractual payment terms (BIS 2017).

SMES: these are generally defined as enterprises with less than 250 employees. The European Commission affirms that for an enterprise to be considered as small or medium-sized, two preconditions must be met. The number of employees must be between 49-250 with an annual turnover less than £30 million.

Cash flow: the movement of money into and out of the business or project. Money going into a construction project comes from payments received for work completed, payments for progressing work, payments for materials on site and payments for services rendered to other organisations. Money going out is payment for materials, subcontract payments, staff salaries and repayment for loans as well as the purchase of capital equipment. The difference between money going in and out is called net cashflow. As Lord Denning once orated, "...cash flow is the lifeblood of the construction industry..." and regular flow is essential to a project's success as well as the contractors' business performance.

Valuation: the process of determining a fair price for a product, property or even service rendered by individuals or organisations. In a construction context, valuation is the measurement of the value of the construction work installed or erected on site during the commencement of works.

UK Construction industry: the segment of the UK economy that is mostly concerned with the provision of construction materials and products, building services, manufacturers, professionals, installers of construction related work and construction clients. Basically, the construction industry consists of organisations that are related to the design, build, operation and refurbishment of buildings.

Standard forms of contract (SFoCs): pre-drafted forms of contracts issued by construction professional bodies and trade organisations (such as the Royal Institute of British Architects RIBA, the Royal Institution of Chartered Surveyors RICS, the Institute of Civil Engineering ICE, the Construction Confederation and the National Specialist Contractors Council, etc.) to codify contract administration practices, allocate risks and regulate the economic transactions of parties within a contract. According to Powell (2012), SFoCs are simply a standardised set of terms designed to suit different types of project and various types of procurement arrangements

Adhesion Contract: readymade terms or contracts drafted by the party with greater bargaining power, leaving the weaker party with one choice; either to accept the contract's terms or reject it. Most adhesion contracts are non-negotiable but enforceable at times.

Subbies: this abbreviation is commonly used in the construction industry to denote or refer to subcontractors.

Tier: denotes the chain and level of firms linked through the series of contracts in the project. The UK Construction Industry Strategy report (2013) titled "Supply Chain Analysis into the Construction Industry" describes the construction supply chain as tiers, whereby Tier 1 is termed the main contractors; Tier 2, specialist contractors and Tier 3, labour-only sub-contractors.

Tier 1 clients: in the context to this study, a Tier 1 client is deemed to be the main client of a construction project (for instance public authorities, developers, private individuals, institutions and funders) that pays for high value construction deliverables.

Tier 2 clients: refer to principal contractors with a high turnover of billions of pounds operating and procuring high value construction works. For example, Balfour Beatty Limited, Carillion Plc (before its insolvency in 2018), Laing Rouke Plc, Costain Limited, Kier, Vinci, Interserve, Willmott Dixon, Galliford Try and other large construction companies listed among the top 100 construction companies in the UK from 2015 to 2020.

Tier 3 clients: refer to subcontractors and specialist contractors with a direct commercial relationship with Tier 1 and Tier 2 clients.

Tier 4 clients: refers to subcontractors and suppliers employed by Tier 3 clients.

Project Bank Account: often abbreviated to 'PBAs' in the construction industry. A PBA is a ring-fenced bank account that transfers payment directly and simultaneously to main the contractor, subcontractors, suppliers and other members of a construction supply chain (Cabinet Office 2012a).

Business performance: this term is often used interchangeably to denote the financial outcome of a business. According to Smith and Reece (1999), business performance can be defined as the operational ability of a company to satisfy the desires of major stakeholders and must be measured to an organisation's accomplishments. In construction, business performance hinges on the profit and

turnover of contractors. Profits are residual amount of monies obtained by a contractor on settling project expenses like preliminaries, subcontractors' accounts rendered, payments for materials, labour, overheads and many more. Turnover is the total income accrual for a company's accounts per year.

1.9 Summary

This chapter provided introduction to the studied problem. The chapter examined the background and context of the research and its rationale, aim and objectives. Subsequently, the chapter explained the scope of the study, the research questions, the study's contribution to knowledge and some brief definitions. The next chapter seeks to review the existing literature in the area of construction costs and project management.

CHAPTER 2

Unfair Payment Practices in Construction

2.1 Introduction

This chapter reviews the existing body of literatures surrounding unfair payment practices in the UK construction industry. The chapter reviews the key features and trends of unfair payment practices. The chapter commences with an overview and composition of the UK construction industry, forms of payment and types of construction organisations. The background to unfair payment practices from the global to UK context is also reviewed. The chapter also appraises contemporary causes and effects of unfair payments and their causal sequence for business performance in the construction industry.

2.2 Composition of the UK Construction Industry

The construction industry is considered the largest economic sector in the UK and a key driver for national growth. The Office of National Statistics (ONS 2019) affirmed that the industry contributes about £117 billion to the UK economy with 2.4 million jobs; equivalent to 7% of the total UK workforce.

According to the Department for Business, Energy and Industrial Strategy (BEIS 2017), the UK construction industry is classified into three main sectors: construction contracting, services and related products. This is further broken down into nine sub-categories that provide a range of activities, such as building construction, civil engineering, specialised construction activities, architectural services, quantity surveying, whole/after-sale of products and materials, renting and leasing, installation, manufacturing and other services. Nevertheless, traditional contracting businesses are the largest sub-sector in the UK construction industry accounting for 70% of the total value added and sectoral jobs, as illustrated in the figure below:





2.2.1 Size and Distribution of Firms

Figure 2.2.1 presents a pictorial view of construction firm categorisation. The data show that majority of firms that make up the construction industry consist of less than 50 people and that less than 20,000 firms employ 250 or more persons. Moreover, 80% of construction firms are sole proprietorships owed by single individuals or are family businesses.

Construction companies that employ more than 250 persons are normally referred to as large contractors and they are most susceptible to using unfair payment practices against subcontractors and suppliers in the industry.



Figure 2.2.1: Number of Construction firms in the UK in 2018 (adapted from Statista 2020)

2.2.2 SMEs

In the UK, traditional contracting businesses are heavily reliant on SMEs. The UK Department for Business Energy and Industrial Strategy (2019) asserted that 99% of businesses operating in the UK construction industry are SMEs. The definition of SMEs in construction varies from one country to another, therefore the real meaning is often difficult to state in a simple or straightforward manner. Arocena and Nunez (2010) classify SMEs as including artisans, ground works firms, painting and decorating firms, joinery firms, brick work firms, mechanical firms and plumbing firms, etc.

Brisoce et al (2001) categorised SMEs as including most subcontractors and suppliers within the construction industry. Conversely, the UK Company Act 2006 asserted that the definition of SMEs sets a company threshold of less than 250 employees, a balance sheet of less than £12.9 million and an annual turnoverof less than £25.9 million. Therefore, in this study, the term SME's denotes construction companies and all subcontractors, suppliers and tradesmen offering an array of construction work and services.

BERR (2008a) argued that as the majority of traditional contracting business are SMEs, their involvement within the supply chain is inevitable. For instance, a

typical supply chain for a project consists of a huge number of SMEs that play a significant role in the successful delivery of construction. Yet, this category of companies is mostly prone to unfair payment practices.

Euler (2017) report acknowledged that very often, SMEs encounter many challenges and complexities in running and growing their businesses, with deferred payments and unfair payments practices on goods and services deemed to be their greatest risk and concern. In addition, the BEIS report (2018) articulated that "most SMEs in construction are adversely affected by a range of payment problems when dealing with large companies or forming part of their supply chain".

The FSB report (2018) revealed that in the last five years, two-thirds of suppliers had experienced lengthened payment terms imposed on them by main contractors. For example, Carillon Plc, the UK's largest construction company before its insolvency, had standard payment terms of 90 days with a maximum of 120 days. Kier, another large company, had standard payment terms of 60 days with a maximum of 150 days, while John Sisk had standard payment terms of 75 days with a maximum of 120 days.

Furthermore, the report asserted that 12% of SMEs surveyed had been asked for discounts for prompt payment, with flat fees of 6% to remain in a large contractor's supply chain and 3% discounts on good and services that had already been supplied. This is clear evidence that SMEs are exposed to an array of unfair payment practices, including longer payments terms and unfavourable practices imposed by main contractors.

Wheeler (2016), the Managing Director of Construction Finance at Bibby Financial Services claimed that "....all too often we see the battle of David versus Goliath in the construction industry, whereas larger contractors and clients often wield the power to impose some sort of terms or practices ...; and SMEs are usually reluctant to negotiate or challenge unfair payment terms imposed on them for fear of losing future work and reputation..."

2.2.3 The Construction Supply Chain

SMEs form a significant part of the construction supply chain and this category of companies is described as complex and fragmentary in nature. Briscoe et al. (2001) argued that whether a small or large project is undertaken in construction, a number of hundreds if not thousands of small and medium-sized contractors cut across the supply chain spectrum.

BIS (2013) claimed that the best way to describe the construction supply chain is through relationships developed by all parties to a contract. For example, Tier 1 contractors consist of main contractors with a direct commercial relationship with their clients. Tier 2 contractors are predominantly subcontractors and suppliers with a direct commercial relationship with the main contractors. Meanwhile, Tier 3 contractors are normally at the lower end of the spectrum in the contracting chain. This category of contractors is made up of subcontractors and suppliers with a direct relationship with Tier 2 subcontractors. Arguably, this study seeks to view the structure of the construction supply chain from three distinct construction clients' perspectives: Tier 1 clients (private or public) to Tier 2 clients (main contractors), Tier 2 clients to Tier 3 clients (subcontractors), Tier 3 clients to Tier 3 clients (subcontractor-subcontractors) and Tier 4 clients and Tier 5 clients (suppliers or vendors).

Figure 2.2.3 illustrates a hierarchical structure of the construction supply chain, where payments flow from the client to main contractor, then to subcontractor and down to the other parties within the supply chain. If the flow of payment is delayed or disrupted by one party, the consequences will then cascade to all parties within the supply chain.



Schematic representation of construction supply chain Figure 2.2.3 Schematic Representation of the Construction Supply Chain

2.3 Insolvency Risk

Key challenges originating from financial, economic and political crises always impede the value of construction output and in most cases, lead to insolvency risks. Unfair payments practices (especially late and unpaid retentions) have left many construction firms on the brink of cashflow problems and liquidation, such that it has become the nature of how firms operate in the industry.

Recently, a government report by ONS (2019) confirmed a sudden 12% increase in insolvency cases in construction companies in 2018 which had been increasing since 2013. Figure 2.3 illustrates companies' insolvencies by industry, with construction having 3,106 cases; followed by administrative and support services with 2,585; and then wholesale and retail with 2,409. The data further suggested that out of 1,857 firms that had collapsed, 62% of construction firms had specialised in different activities such as groundwork, demolition, electrical and plumbing installations, while 32% had constructed building projects and civil engineering works. Peter and Arewa (2018) alluded that insolvencies experienced by most firms were attributed to lack of cash due to thin profit margins and financial pressures. Kelly (2020) confirmed that despite a firm's unique traits towards insolvency, most companies experienced cashflow problems as a result of unfair payments. Besides, different researchers have argued that the "issue of late, delayed and non-payment has left many construction companies grappling with cash flow problems" (Ramachandra 2015).





2.4 Payment Forms and Related Issues

Payment is a key metric of the exchange and transfer of funds for all transactions, whether they involve the acquisition of goods, services or financial assets. In construction, payments to contractors are often spelt out in contract arrangements and are normally monetised. Payment is monetary consideration for work done or work that is yet to be performed within a stipulated time frame.

Amer-Ali (2006) argued that unlike other industries, payment is a major concern for construction companies due to contractors' large outgoings and longer payment durations, the expensive nature of the works to be carried out and most importantly, contract payment clauses based on the completion of the construction works. For a better understanding, contractors are in most cases required to perform for up to one month before applying or receiving payments from clients. This payment method contrasts to what is seen in other industries such as aviation, tourism, pharmaceuticals etc, where payments are made before services are rendered. Moreover, the longer duration and huge capital outlay involved in the construction process creates financial burdens and cashflow pressures for many construction contractors, as most of them operate on poor margins with a strong dependency on payments.

2.4.1 Significance of Payments to Contractors

Regular payment to construction contractors is a significant element that allows the contracting organisation to pay for expenses incurred in carrying out works on site, thereby permitting completion of a project. Jiang (2012) and Hasmori et al. (2012) posited that the importance of payments cannot be overemphasised in projects, because the cashflow position of any contracting firm has a greater deal to do with performance. Figure 2.4.1 illustrates a typical example of a contractor's cashflow, with payments coming from interim/stage payments, payments for completed works, retentions, final payments, bank loans, sale of company assets and shareholders' funds, etc.

While the outgoing payments in figure 2.4.1 denote money going out of a contractor's account such as office running costs, management staff salaries, payments to subcontractors or suppliers, payments for plant hire and site labourers. The difference between the cash inflows and outflows is termed as net cashflow. The industry norm however, is that the contractors spend a huge amount of monies in carrying out works before receiving payments from the client. This means project cashflow is dependent on the contractor's working capital or on receiving funds from banks, such that any delayed or later payments could lead to contractors completing with a negative balance in their accounts.

Indeed, most contractors rely on interim payment as a major source of income for the successful running of a construction business. This aligns with Lord Denning's famous phrase: "payment is a lifeblood of the construction industry". The National Construction of Sri Lanka (NCSL 2008) argued that timely payments to contractors are pivotal in the construction business because it's help boost working capital, drive efficiency and optimise the construction process.





2.4.2 Payment Mechanisms

Traditionally, payment mechanisms from clients to contractors are based on work done and through instalments. Payments in instalments are referred to as interim or progress payments and are mostly affected by the issuance of an 'interim certificate'. Judi and Rashid (2010) affirmed that depending on the type of contract, the weekly or monthly disbursement of payments to contractor is based on the valuation of work carried out by a quantity surveyor.

Basically, the purpose of an interim payment is to ensure that the contractors are paid for work carried out on site and that they have sufficient funds to progress with the remainder. Eugenie (2016) described progress payment as a disbursal portion of construction monies from client to contractor after the completion of a certain stage of works in a project.

Generally, most construction projects take a substantial amount of time to complete and a large amount of money, which is greater than the financial strength of the contractor (or subcontractor). Therefore, regular payment by instalments curbs the need for a contractor to seek for additional funding as well as the cost of borrowing (Murdoch and Hughes 2000). Halpin and Woodhead (1998) argued that often there is a time lag between when contractor incurs expenses and getting paid for the work done, meaning not receiving regular payment on time tends to drive contractors to brink of insolvency.

Similarly, a retention payment is an amount withheld by a client to ensure that the work is properly executed and to mitigate the risk of defect liabilities. Hughes et al. (2000) were of the opinion that a retention payment is an act of performance catalyst and a deterrent against defective work. In the UK, most SFoCs provide an adequate mechanism for pre-agreed amounts of retention payments (usually 3-5%) and for their disbursement at intervals in the project.

An advance payment is also a common type of payment in the construction industry. It is a payment agreement that allows clients to pay contractors in advance for work not yet performed up to a limit of 15% to 20% of the contracted sum. This type of payment gives contractors a financial boost to commence the works, without seeking for additional funding or external borrowing. However advance payments are common in developing countries due to insufficient financial outlays and are often perceived to encourage corruption (Arewa and Farrell 2015).

In a nutshell, common types of payment mechanisms in construction can be categorised based on time and performance factors as follows:

49

- a) Periodic payments: these are based on a length or portion of time. It is a situation where the contracting parties agree to payment terms based on a certain length of time. The frequency of periodic payments can vary from monthly to yearly, depending on the contractual arrangement.
- b) Phase/Stage: this refers to a situation when payments are made at specific phases or stages of the works as agreed in the contract conditions.
- c) Performance payment: a payment system based on the total value of the works executed by the contractor at intervals or on a periodic basis. The frequency of periodic payments could vary from fortnightly to monthly, depending on the contractual arrangement. In reality, quantity surveyors are usually engaged to measure the work performed on an interim basis, for which contractor will be paid accordingly.
- d) Advance payment: this refers to a situation where a sum of monies is paid to a contractor before the execution of works on site.
- e) Final payment or payment after completion: this refers to another form of payment triggered by the completion of all activities scheduled before issuing payment to a contractor.

Apart from the types of payment mechanisms stated above, other forms or methods of payment that are commonly used in construction include target cost, reimbursable, cost-plus fee, fixed fee, guaranteed maximum price; lump sum, open book accounting, mechanic lien, incentive contracting and trade credits, etc. These types of payment are normally prescribed by collaborative forms of contract such as NEC, ACE PPC 2000, CIOB Complex Construction Contract 2013, etc. Generically, regardless of the types of payment in the construction sector, certain project conditions and client circumstances usually influence when contractors are paid.

2.5 Types of Unfair Payment Practices

The UK Department for Business Innovation and Skills (2017) report titled "Challenging Grossly Unfair Payment Terms and Practices" defines "unfair payment" based on the Late Payment Act (1998) provision, Section 4 subsection (7A), under three specific aspects such as: (i) anything that is a gross deviation from good commercial practice and contrary to good faith and fair dealing; (ii) the nature of goods or services in question; and (iii) whether the purchaser has any objective reason to deviate from the stipulated/contractual payment term.

This denotes that the word "grossly unfair" covers all contractual terms and practices that relate to late payments or unfair terms drawn up for general use. The report subsequently identified common grossly unfair commercial practices in the UK construction industry: late payment to contractors/suppliers, "pay when paid", disparities in rates for items, exclusion of provisional remedy, discounts, the imposition of rates on subcontractors and unpaid retentions to contractors.

Peter and Arewa (2018) expounded that the phrase "late" or "delayed payment" are often used interchangeably to mean payment not received as when due. However, this study recognises delayed payments as a failure to honour payments within the stipulated time frame and in accordance with the contract's conditions. A late payment is delayed or missed within the time stipulated in the contract documents. As previously discussed in Chapter 1, the issue of late payment is deep rooted and chronic in nature for construction contractors.

'Pay-when-paid' or 'pay-when-certified' stems from conditional clauses whereby payment by one party is dependent upon one party receiving payment from another. This type of payment is mainly synonymous with contractors in charge of settling subcontract or supplier payments initiated by the client. Although the conditional clause is prohibited by section 113 of Construction Act 1996, the practice is still widely adopted and used in the construction industry.

Disparities in rates for items, the exclusion of provisional remedies and discounts for items are gross terms usually imposed on subcontractors and suppliers to exclude them from utilising any sort of remedy in the contract or providing lower rates of payments. For instance, the FSB (2018) reported that 12% of suppliers had given discounts for prompt payments, 7% for retrospective discounting and

6% for a fee to remain in a contractor's supply chain and 3% discount applied after goods and services had been supplied. Table 2.5 below represents common types of unfair payment practices, nature of clients and parties at risks.

No	Nature of Clients	Types of Unfair payment practices	Parties at risk
1	Tier 1 Client (Public or Private)	Late payment, disparities in rates for items, discounts and lower interest	Tier 2, 3, 4 and 5 clients
2	Tier 2 clients (Main Contractors)	Late payment, disparities in rates for items, discounts, lower interest, unpaid retentions, longer payment terms, pay-when-paid, flat fees	Tier 3, 4 and 5 clients
3	Tier 3 clients (Subcontractors)	Late payment, unpaid retentions, discounts, flat fees	Tier 4 and 5 clients
4	Tier 4 client (Sub- subcontractor)	Late payment, unpaid retentions, disparities in rates for items, discounts, flat fees	Tier 5 clients
5	Tier 5 clients (Suppliers)	Late payment, unpaid retentions, disparities in rates for items, discounts, flat fee	Tier 5 clients

Table 2.5 Types of Unfair Practices and their Commonality to Construction Clients at Risk.

2.6 European Perspective – Unfair Payment Practices

According to the European Payment Report (EPR 2019), late payment in the construction industry is a deep-rooted problem for both small and large businesses in Europe. The report affirmed that a range of payment issues adversely affects most businesses and there is evidence that bad payment habits of large and public sectors are deteriorating. The EPR indicated that on average, large businesses paid their invoices after 40 days, up from 34 days in 2018, while the public sector settled invoices after 42 days, up from 40 days in 2018.

Findings from table 2.6 reveal that in 2019, most businesses in the UK offered longer payment terms compared to other European countries whereby business-to-business (B2B) had average payment terms of 45 days, an increase of 16 days compared the 2016 report. Business-to-customers (B2C) had an average of 25 payment days to settle their invoices. Comparing these figures to Europe as whole, average payment terms were 34 days B2B and 25 days B2C, respectively.

Furthermore, the report acknowledged that 56% of UK businesses were asked to accept longer payment terms than they were comfortable with, by both large and small companies as well as the public sector, as seen in figure 2.6.

Overall, the report acknowledged that the UK had longer payment terms compared to other countries and that there was no evidence to suggest that the problem was abating.

Country	Average business to	Average time business	Dublic Sector new terms
Country	Average business-to-	Average time business-	Public Sector pay terms
	business payment terms	to-business customers	(day)
	(uays)	(days)	
Austria	22	20	42
Belgium	38	20	60
Bosnia and Horzogovina	33	16	42
Bulgaria	35	10	42
Dulgaria	20	21	12
	41	21	38
	31	17	31
Denmark	24	16	26
Finland	23	14	23
France	48	21	48
Germany	28	16	27
Hungary	25	11	39
Ireland	33	31	23
Italy	24	24	67
Latvia	20	12	14
Lithuania	25	11	13
Netherlands	45	37	40
Norway	23	18	27
Poland	26	23	25
Portugal	63	25	75
Romania	41	23	45
Switzerland	34	27	42
United Kingdom	45	17	28

Table 2.6 Average payment terms across European countries (adapted from EuropeanPayment Report 2019)



Figure 2.6 Percentage of agreement to longer payment terms by UK Companies 2019 (adapted from the European Payment Report 2019)

53

2.7 Historical Perspective – Unfair Payment Practices

In the last 50 years, unfair payments have remained a reoccurring issue in the UK construction industry. Banwell (1964) opined that "...the operation of the payment system is not always smooth... payments to the main contractor by the clients are often slow and uneven, with consequential delays in payments to suppliers and subcontractors. This has an adverse effect on the efficiency and stability of the whole industry ...". Similarly, Latham's (1994) report titled 'Trust and Money' stated that payment difficulties in the construction industry was one of the greatest difficulties faced by contractors, as most businesses in the sector were usually afraid that they would not be properly paid by clients when due, or as a result of errors in certifying monthly payments. The report claimed "... the cascade system of payment in the industry, from client to main contractor, main contractor to subcontractor and down the supply chain exposes different ventures to insolvencies...."

The National Audit Office Report (2005) titled 'Improving Public Services through Better Construction' concluded that "in general unfair payment practices such as unduly prolonged or inappropriate cash retention have undermined the principle of integrated team working and the ability and motivation of specialist suppliers to invest in innovation and capacity".

Moreover, the report by the Business and Enterprise Committee 'Construction Matters' (2008), stated that "...there remains a deep-seated culture among main contractors of delaying, reducing or simply avoiding paying their subcontractors for many years." These findings may not be completely unsurprising, as both Banwell and Latham's reports highlighted the existence of payment problems in the construction industry.

Undoubtedly, different legislations and initiatives have been enacted over the years to tackle unfair payment practices, yet to date, the problem remains deeprooted in the construction industry. Recent data released by Build UK under the Duty to Report on Payment Practices and Performance (2019), asserted that some

of the biggest UK contractors still settled their invoices within 50 days and beyond, despite growing pressure, legislation and initiatives to alleviate poor payment practices.

2.8 Causes of Unfair Payment Practices

A thorough understanding of the causes of unfair payment practices will help to identify and implement measures towards alleviating the occurrence of this problem in projects. Hughes et al. (1998) stated that there were two primary causes of late or unfair payments based on clients' circumstance: **'cannot'** and **'would not'** pay attitudes. The **'cannot pay'** situation refers to a payer's financial difficulties due to insufficient capital and improper cash flow management. A **'would not'** pay situation however is when a payer refuses to pay a contractor or delays the payment for reasons such as incomplete work, mistakes in valuation and disputes from claims, etc.

The '**would not**' pay attitude is common among Tier 1 and 2 clients, thereby delaying payment to subcontractors and suppliers in order to sustain their cash flow for other projects and reduce overdraft facilities. Some school of thought believe that the **"would not**" pay attitude can be a deliberate act or perhaps the reluctance (unwillingness) of the subcontractor to protect their client-contractor relationship by any means. Besides, there is little or no empirical research to corroborate the scale of clients' deliberate acts of **"would not**" pay as opposed to genuine reasons for unfair payments. Danuri et al. (2006) mentioned that a client's financial incapability, local culture/or attitude, the use of conditional clause 'pay when paid', failure to implement good governance in business and delays in certification are the main causes for late payments in the construction industry.

The authors stressed that the industry practice of most consultants and even clients was to withhold or delay the certification of payments for biased reasons, even when the contractor has submitted an appropriate claim for work done. Euginie (2006) and Sin (2006) opined that this reason and other factors fuelled by

SFoC's used in construction permit the client/contractor to withhold or to refuse outright to pay monies owed to their supply chain.

On the other hand, there seem to be technical problems associated with unfair payment practices. For example, Abdul-Rahman et al. (2006) claimed that the following were major reasons for unfair payments in the construction industry: poor technological know-how concerning contract management; clients' lack of payment knowledge; payment modalities; delays in preparing valuations; certification of interim payments by client representatives (e.g. QS); insufficient documentation; poor data management; inaccurate valuations for work done; involvement of too many parties in honouring certificates; and conflicts between parties and misinterpretations of clients' variation orders. Frimpongs et al. (2003) are of the view that unnecessary bureaucracies in some public departments were responsible for the chronic unfair payments that are deep rooted in the sector.

Additionally, there appear to be strong cultural attributes to late payment issues in construction. For example, Abdul-Rahman et al. (2008) avowed that project culture significantly influenced late payments. The authors argued that in countries where the rule of law was ultimate; late payment was significantly minimised and vice versa. Ayudhya (2012) identified 24 causes of late payment in Thailand's building projects. The first five factors were identified as owners' financial problems, delays in work approval, major accidents, inaccurate bills of quantities and sub-standard workmanship.

Similarly, Yao (2014) affirmed that in poor financial management, delays in certification, contractual provisions, use of conditional 'pay-when-paid' practices, industry's culture and technical problems were some factors that influenced late payments. On the other hand, there is reason to believe that the nature of construction ventures has altogether contributed to unfair practices, such as 'all-comers' businesses; stringent competition, low business capital/asset-base, reliance on cash flow to sustain operations and lack of effective use of Information and Communication Technologies (ICT). For example,

56

Gibson (2000) and Pettigrew (2005) argued that the nature of traditional contracting enabled companies or individuals with little investments and limited experience to set up construction businesses. As a result, most of them suffered from financial difficulties when the settlement of their payment claims became irregular.

Besides, the industry has a unique cascade system of payment, whereby payment flows from the upper to the lower tiers; together with a prevalent condition of payment culture of 'paid if or when paid' that have made late payments and other similar practices deep rooted in the industry (Ramachandra and Rotimi, 2015). In summary, the literature review regarding causes of unfair payments can be largely attributed to the following factors: cultural (or attitudinal), regulations, technical and industrial processes. For clarity, the author wishes to recapitulate these factors into four main categories.

2.9 Cultural (or Attitudinal) Factors

a) Pay when paid

The historical inclination of the 'pay when paid' notion has been described as one of the worst factors for Tier 3 and Tier 4 clients in the UK construction industry. The 'pay when paid' notion instils a conditional payment provision for the main contractor in respect to the subcontracted works carried out. In other words, 'pay when paid' operates on the principle that a main contractor should not be liable to pay his subcontractors until payments have been received from the client. McCann (1999) ascertained that the rationale for use of such a draconian measure in the construction industry was to release the contractor's liability from payments and transfer the risk of the client defaulting to subcontractors further down the supply chain.

Arguably, due to the disorder caused by the use of 'pay when paid' in the construction industry, the Housing Grants, Construction and Regeneration Act (HCGRA 1996) worded a clause to counteract and prohibit this unlawful practice. Section 113 (1) of HGCRA states that 'a provision making payment under a

construction contract conditional on the payer receiving payment from a third person is ineffective unless that third party or person under contract is insolvent'. This means that if the client is solvent, the main contractor is prohibited to withhold payment or impose the 'pay when paid notion' on the subcontractor.

Moreover, it can be argued that the conditional payment notion cannot be completely rooted out from the industry, as most main contractors are still enforcing the use of such practices on subcontractors and even suppliers to date. For instance, in an event of a client withholding payments from a main contractor, the subcontractors and other members further down the supply chain are also exposed to risks of delayed payments or 'pay when paid' practices.

b) Client-contractor relationships

In the construction industry, relationships are often criticised for being adversarial and dynamic in nature. Client-contractor relationships, in particular client to main contractor (C-MC), main contractor to subcontractor (MC-SC), subcontractor to sub-subcontractor (SC-SCC) and sub-subcontractor to suppliers or labourers (SCC-S/L's), are subject to a partnering ethos, ranging from trust comportments, long term-commercial gains and 'who you know' in the construction industry.

Regardless of contractual connections, most of these relationships are deemed to be more cordial and long-term, often established with elements of familiarity and protectionism. For instance, many subcontractors and suppliers may not chase late payments or reject any unfair practices that are imposed on them becauseof the desire to protect their client relationship. This intuition has empowered some main contractors such as Carillion and many others to instil longer payment terms of 120 days and beyond. Bresnen and Marshall (2000) argued that whilst traditionally adversarial working relationship are deep-seated in construction there are elements of relationship trust and 'cosiness' that have evolved as well.

c) "Cannot" pay attitude of clients

The "cannot" pay attitude of clients refers to a situation when their ability and commitment to pay on time is impeded. A client as a singular person or entity in construction will commission the building of a project under a particular budget, that is sourced through the client's own monies or a consortium of investors interested in the project.

However, most clients are irregular purchasers of construction work and may be unaware of industry risks that tend to allocate a certain amount of money for a project. When the allocated budget runs into deficit, the client experiences financial difficulties and cash pressures in running the project, which results in late payments to the parties involved in the project.

Arguably, it has been affirmed that the 'cannot pay' attitude of many construction clients is influenced by financial and economic problems such as a shortage of funds, difficulties in obtaining loans, inflation, decline in bank lending, insufficiencies, market instability, poor credit and the mismanagement of financial resources, etc.

d) "Would not" pay attitude of clients

In contrast with the 'cannot pay' attitude, the typology of 'would not' relies on a client's willingness and intention to pay. Clients are obliged in most forms of contracts to take sole responsibility for carrying out all payments for completed works on site. However, a common practice for clients is to invent or seek a spurious claim that will eventually delay or withhold payments. Kenley (2002) argued that some construction clients deliberately withhold payments from their main contractors inorder to pay for other ongoing projects or personal benefits.

Griffiths et al. (2017) further argued that main contractors also exert considerable commercial pressure on their subcontractors and suppliers by withholding or extending payment terms in order to boost their working capital position and thereby gain cheap finance. Withholding payment as legally enforced in SFoCs, is a condition precedent of a 'pay less' notice issued by a client to a contractor within five days of the final payment.

Conversely, it could be argued that the 'would not' pay attitude of a client may be influenced by the diversion of project funds, corruption, client behavior, disputes, unsubstantiated claims, errors in valuation, failure to comply with an architect's instructions and so forth.

2.10 Industrial Factors

a) 'Cowboy' bullying

The UK construction industry is generally perceived to have 'cowboy' clients and contractors in its rank and file. The Department of Environment, Transport and the Regions (DETR 1998a) defines cowboys as persons being mendacious, incompetent and tax evaders within construction. Typically, cowboys are said to be clients and major contractors who seek to instigate or influence bullying tactics on SMEs for their own commercial gains or individual agendas.

For instance, the Federation of Master Builders report (FMB 2018) affirmed that cowboys are a serious threat to three quarter of construction firms and that they are often involved in bullying tactics; as most of these clients deliberately delay or withhold payments without any good reasons or on spurious grounds. The report further revealed that a quarter of SME's claimed that they have to wait more than four months to receive payments from clients or large contractors. Of late, the BEIS (2018) reported that over the past two years, a third of small suppliers and subcontractors had witnessed drastically increased payment terms due to cowboy bullying.

b) Corruption

Construction contractors are generally undercapitalised and any major interruption to their working capital may result in ripple effects for the entire supply chain. Arguably, the reliance on subcontracting that dominates the entire construction industry has stemmed an increase in corrupt and fraudulent practices by many contractors. It can be common practice for a main contractor to take advantage of their subcontractors and suppliers by inducing cowboy bullying, aggressive accounting strategies and lengthening payment terms. Humphreys et al. (2003) argued that main contractors have realised that their greatest potential for saving costs lies with their use of subcontractors' payments. Thus, when a contractor is paid for completed works on site by a client, the contractor shores up the cash to boost their cash position and liquidity whilst exposing their subcontractors and suppliers to late payments. This practice is perhaps the leading cause of why late payments persist in the construction industry.

c) Low barriers to entry and exit

As previously discussed, the UK construction industry is populated with a large number of SMEs and a small number of large companies. SMEs, which are mostly Tier 3 and 4 clients, are predominant in the provision of building construction and specialised activities such that to date, it is inevitable that main contractors cannot carry out projects without engaging them.

For this reason, the demand for interdependent SMEs in the construction industry has escalated to a point whereby any sole trader or individual can set up a smallsized firm with narrow expertise and poor capital to meet the industry's varied demands. Hence, when assessed in terms of barriers to entry and exit, the industry is proven to constitute low (no) barriers that permit the entry of small, undercapitalised companies with a heavy reliance on credit (Hughes et al. 1998). Low barriers to entry and limited capital results in cash flow difficulties that result in a contractor extending payment periods or even paying later than agreed.

Perhaps, it could be argued that majority of contractors replenish their working capital by taking advantage of payments for completed works either by deferring payments or paying their small subcontractors and suppliers later than agreed date.

d) Supply Chain finance

Fundamentally, access to working capital and finance for small to medium-sized enterprises (SMEs) remains a key challenge to the UK construction industry. Of late, the global economic financial crisis, regulatory changes and tightened lending standards have resulted in limited access to finance and cash flow pressures for many construction companies. Evidence from the RSM report (2017) shows that 63% of SMEs (mostly subcontractors and suppliers) had experienced financial difficulties for their business that forced most of them to rely on supply chain finance (SCF).

Supply chain finance (or reverse factoring) as an alternative source of finance, is mainly adapted by Tier 1 clients to improve and maintain their suppliers' working capital and cash flow through accelerated payments. Furthermore, SCF operates with Tier 2 clients working with third party funders mostly in financial institutions like Santander, Lloyd's Bank, NatWest and many others, to carry out supplier payments in conditions precedent to a specified date with interest being charged. However, it could be argued that despite its advantages, the SCF tool has become a tool of exploitation by main contractors whereby suppliers or subcontractors are forced to agree to longer payment dates or even being paid later that agreed. For instance, Carillion had adopted 120 days payments on its SCF to suppliers and subcontractors that exposed most of them to insolvency risks.

Although the SCF initiative was deemed to be a good idea to relieve SMEs from financial debt, recent findings have shown that contractors that signed up for the initiative became trapped, because the main contractor took advantage of the initiative by eventually paying their subcontractors over longer periods.

2.11 Technical factors

a) Defective works

As stated in most SFoCs, the contractor is obliged to carry out and complete works in a proper and workmanship-like manner, as stated in the contracts. The employer then ought to pay for the value of works that have been completed. Typically, the main contractor will be held liable for defective works done or materials carried onto a site by a nominated subcontractor or supplier.

In the case of a client QS notifying the contactor of any defects, the subcontractor will be responsible for rectifying them before payment has been made by the main contractor. In practice, any defective works done by a subcontractor will constitute a reasonable excuse or claim for a main contractor to withhold or delay paying the subcontractor.

b) Delay in approval of works

Delaying approval for works done on site is likely to cause late payment problems for contractors and other members within the contractual payment chain. Ansah (2011) ascertained that a consultant usually delays the approval of completed works either due to faults or for other reasons; for example, failure to comply with an architect's instructions, disputed work, defects, errors in claims and delays in project resources. These reasons will likely cause the client to refuse to initiate payments which will result in later payments to the contractor and their suppliers.

c) Errors in submitting claims and the valuation of work

Reeves (2003) argued that errors in submitting claims and valuation of completed works on site are among the main causes of late payments. Errors including incorrect calculations, wording of contracts, deductions to previous payments or retentions, omission of variation orders and failure to comply with an architect's instructions will result in later payments. When a contractor submits an interim application with errors, the quantity surveyor is forced to reject the application through the issuance of a payment notice that will require the contractor to resubmit, hence resulting in later payments.

d) Insufficient documentation and information

According to Hamzah et al. (2014), insufficient documentation and information submitted by a contractor will cause the client QS to withhold payments or even pay later that agreed. Insufficient documentation will give rise to the disruption of the entire payments process, whereby both contractor and QS will be forced to carry out proper checks on documents before submitting them to the main client. Thus, delayed payments are likely to occur to project.

2.12 Regulator Factors

a) Complex payment legislation

In recognition of the pervasiveness of late payments in construction, the UK government has enacted a number of different legislations and initiatives to address the problem. For example, the Housing Grants Construction and Regeneration Act (HGCRA 1996) stipulates a 28 days payment timeframe to any construction contractors with a total delay period of 45 days, after which interest on the delayed payment will apply. In addition, the Local Democracy, Economic Development and Construction Act (2009), the Scheme of Construction Contract Act (1998), the Late Payment of Commercial Debts Regulation (LCDR 2013) and the Construction Supply Chain Payment Charter are believed to include some stringent measures in tackling late payments.

Griffiths et al. (2007) argued that despite the positive effects of legislative interventions, late payments have continuously plagued the construction industry. Sood (2016) claims that complexities and lack of knowledge surrounding most payment legislations have enabled few to benefit, whilst difficulties remain for many in applying them in real practice.

b) Complexities and ambiguities within legislations provisions

Peter and Arewa (2018) argued that all SFoCs constitute payment mechanisms that enable parties to deal with payments in an efficient and timely manner. The payment mechanism advocates on the right to periodic payments, procedures and time-frame in triggering payments and role of the payer (certifier) under contract.

Certainly in real construction practice it rare for parties to adhere to the payment mechanisms in SFoCS due to the following: deliberate withholding of payment, the

use of 'pay when paid', local cultural attitudes, the 'cannot' or 'would not' pay attitudes of clients, insufficient documentation, poor communication, etc.

c) Use of adhesion contracts

Sharkey et al. (2014) argued that frequent amendments or modifications to SFoCs help to distribute and allocate risk to parties with weak bargaining power, thus leaving one party at a disadvantage. Adhesion contracts termed 'back to back' are mostly drafted by Tier 1 and 2 clients with the purpose of favouring themselves over other tiers as a result of their commercial bargaining power and predominance in the supply chain.

Normally, the practice is for the client or contractor to draft SFoCs by stipulating terms and conditions which are mostly deemed to be ambiguous and unfair to lower tiers, for instance the inclusion of longer payment terms, discounts for prompt payments, supply chain fees/flat fees, exclusion of provisional remedies and right to withhold payment, etc.

d) Multi-tiered contractual frameworks

The multi-tiered, or in other words hierarchical structure, of the industry's contractual frameworks places construction at a susceptible risk of late payment. Contractors within the contractual supply chain receive payments from clients above them and thus responsible for delivering those funds to the subcontractors and suppliers below their chain. This tiered or 'water flow payment structure' constrains the ability to receive timely payments and increases the receipt of later payments.

2.13 Role of Cashflow in Unfair Payment Practices

One of the most crucial yet difficult aspects of construction is to ensure that all members in the supply chain have sufficient cash to fund their businesses. Most project cash-inflows are designed to be on a regular basis for all parties, though often this does not happen as planned. The reason is that there is a significant time lag from the point at which a contractor receives payments to the point when they have to pay their supply chain. 'Work first then get paid later' is an old adage of construction practices, as the contractor will have to commence works at a site by forgoing incoming payments.

Cashflow refers to the flow or movement between the incoming and outgoing payments of a business (Cooke and Jepson 1986). According to this school of thought, incoming payments are usually termed as positive cashflow and are credited to company balances. Meanwhile, outgoing payments are termed as negative cashflow and are debited from company accounts during the project's execution.

Figure 2.13 illustrates a typical relationship between income (cash in), expenditure (cash out), time and the cumulative costs of a contract (Hendrickson 2008). The cash time curve, referred to as the S-curve, represents the cashflow forecast of a construction project. A key deduction from both the statutory and staircase points in the S-curve depicts a huge amount of expenditure (as cash out) spent by the contractor from the beginning of the project (due to site mobilisation and the erection of expensive fixtures on the ground) and the minimal amount of income received into the project. The difference between total income and total expenditure is termed as 'project deficit', covering short-term borrowings, bank loans, trade credits, short sales and other measures that will boost cashflow for a project.

It is often reported that construction is fundamentally undercapitalised; most contractors do not have sufficient capital to run their businesses and execute projects on site. Moreover, many measures are subject to high interest rates and scrutinies that force contractors to seek additional ways to boost their cashflows through inputting unpropitious measures or terms to their supply chain, for instance longer payments terms, delayed payments, etc.



Figure 2.13 Typical S-Curves of Contractors' income versus expenditure on projects

2.14 Effects of Unfair Payments Practices

Undoubtedly, unfair payments lead to untold consequences for contractors, particularly SME's. The imbalance between company's revenue and outgoing payments, proliferation of longer payment terms and unfair practices have impacted businesses' ability to manage their cashflow and plans for further investment. Amoako (2011) claimed that late payments create financial hardship for contractors; negatively effecting the supply chain, delaying project completion, causing disputes and cashflow problems, finally coercing businesses into liquidation (or insolvency).

Ramachandra and Rotomi (2011); Abdul-Rahman et al (2014) Danuri et al. (2006) all argued that the gravest effects of late payments include cashflow problems, increased financial difficulties and stress for contractors. Other authors such as Howorth and Wilson (1998), Munaaim et al. (2007) and Akinsiku and Ajayi (2016) all argued that late payments erode the profitability of companies, increase the risk of insolvency and strain client-contractor relationships.

Furthermore, the BEIS report titled "Late Payments, Retentions and Government Procurement" (2018) asserted that knock-on effects of both late and unfair payment practices on small businesses in the UK include cashflow problems,
failing profit performance, delayed payments to suppliers and wider impacts on the entire supply chain and the economy as well. The report further highlighted late payment's impact on business growth, future investment and administrative burden, as valuable amounts of time and money are spent in chasing payments owed to clients or main contractors. For instance, the BACS report (2019) confirmed that on average, small to medium companies spent 1,060 hours per year chasing late payments, with approximate costs of £9,000 per business.

Besides, there is little research concerning direct and indirect relationships between unfair payments and construction business performance. Moreover, it would be necessary to advance research on the effects of unfair payment practices on construction; such as business continuity, reputation, quality of construction deliverables, health and safety, productivity, staff retainment and even people themselves.

Conversely, it is incumbent to measure clients' views or attitudes to the effects of unfair payments on construction businesses because the root causes of late payments stem from them. There would likely be a temptation to focus on causes and effects of unfair payments without looking at clients' attitudes to the problems. Understanding whether clients' attitudes to unfair payments are deliberate or inadvertent will help proffer a flexible solution to the problem.

2.15 Unfair Payments and Construction Business Performance

Notably, there is an overwhelming body of literature that acknowledges how unfair payments constrain a contractor's cash flow and its domino effects on the contractual supply chain. For instance, Wu et al. (2008) argued that persistent cash flow problems have potential to force company into insolvency.Ramachandra and Rotimi (2015) asserted that immediate effects on cash flow drives contractors (including subcontractors and suppliers) to seek for additional funding by means of overdrafts, loans, trade credits or other means to sustain them. Similarly, Ang (2006) and Singh and Lakanathan (1992) claimed that poor

rates of business continuity or the survival rate in the construction industry is a result of late payments. On the contrary, Love and Holt (2000) affirmed that within mainstream construction management literature, business performance and continuity receives scant attention due to the complexity of its objectives and the role it offers in improving business.

Ward et al. (1991), Love and Holt (2000) and Kagioglou et al. (2001) argued that traditionally, a contractor's performance in construction predominantly focused on project specifics in terms of time, cost and quality. However, due to changes in the economic situation and a shift towards business development strategies, business performance and continuity are critical issues for clients, contractors and stakeholders. For example, the collapse of Carillion in 2018 raised many issues concerning construction business performance and continuity as it related to the never-ending late payment quandary. Carillion as one of the UK's largest Tier 2 clients had used late payments as its business model to hide its true financial performance and continuity.

The adverse influence of late and incomplete payments on a contractor's business and performance has been highlighted; for example, contractor non-payment is a cause of escalating disputes (Carmichael 2002), and "the risk of late payments [y] is very common in the 30 ECAM 20 industry and has driven many consulting firms to the edge of bankruptcy" (Kometa et al. 1995a).

Yet, there is little or no research that directly measures the relationship between unfair payments, business performance and continuity in the construction industry. Franco et al. (2007) asserted that the definition of the term or meaning of business performance lacks cohesiveness due diverse views and its contribution towards the body of knowledge. Moreover, the performance of a business's success or failure in achieving its goals can be illustrated in a number of ways.

Arguably, business performance in construction is considered to be synonymous with assessing financial performance and chances of business continuity using

financial ratios. Ellis (2007) argued that financial ratios such as profitability, return on assets, return on capital employed, fixed assets, return on equity, debt ratio and liquidity ratio could be used to measure business performance over time and assess its chances of continuity and also compare it with other competitors. As for this study, the researcher has used four financial ratios, namely operation profit margin, return of capital employed, liquidity ratio and debt ratio to assess a contractor's business performance in relation to unfair payment practices in the UK construction industry.

Usually in calculating any financial ratio for a business, income statements and balance sheets (financial statements) published in acceptable accounting standards are evaluated to determine the business performance of a construction firm. On the contrary, small and medium contractors are often reluctant to publicly reveal their financial statements whilst bigger companies are compelled by regulations to publicly announce and even publish their own. However, it is pertinent to note that the indicators used by accountants and economists to assess the financial performance of entities are not an accurate assessment of business performance.

For example, in 2018/19, KPMG are currently under investigation by the UK government and Financial Reporting Council (FRC) for allegedly signing off Carillion's inaccurate figures that publicly portrayed the company in a good light. Nonetheless it could be argued that most of the UK's large construction firms tend to impose aggressive account strategies such as cash boosting their working capital and maintaining profits by extending their payment terms or even paying suppliers and contractors later than agreed. McCall (2016) argued that for any construction contractor, working capital is a key financial performance of a contractor's ability to cover its short-term obligations whilst funding the ongoing project.

Thus, it is unsurprising that most small and medium-sized construction companies are more constrained by working capital pressures and later payments that lead to higher risks to business continuity and even insolvency. Pheng et al. (2010) affirmed business continuity as identification and readiness of construction businesses in maintaining adequate levels and continuous well-being of a firm in times of disruption such as inadequate capital, cash flow pressures, price fluctuations, late payments and so forth. He further argued that most construction companies appear to lack a properly integrated knowledge and approach towards its business continuity. Hughes et al. (1999) stated that most construction companies are developed without formal qualification and low capital such that it is common for a company to collapse whilst leaving others at risk of late payments.

2.16 Summary

This chapter examined a host of issues surrounding unfair payment practices, payment forms and mechanisms, the magnitude of unfair payments and the relationship between construction business performance and unfair payments.

Furthermore, the choice of literature adopted for this study does not negate any other contributing factors and effects of unfair payments associated with clients and construction contractors, despite the existence of specific payment legislation and initiatives.

CHAPTER 3

Contractual, Legislative and Regulative Measures to Payment Practices

3.1 Introduction

This chapter examines existing literatures concerning contract legislation and other *inter alia* measures regarding payment practices in the construction industry. Although it is essential to have a thorough understanding of existing measures, it is also useful to review the efficiency and credibility of payment practices such as the right to receive prompt payments, the provision of interim payments, the right to a pay lessnotice or deductions, interest on late payments, withholding of retentions and other payment related provisions.

The chapter also reviews contract theories, contractual clauses, procurements, legislative provisions, payment charters, codes, administrative procedures and other measures introduced to mitigate the problem.

3.2 Overview of Contract theories

Many contract theories exist. For example, there are traditional contract doctrines, modern contract theory, classical and neoclassical, transactional cost economic theory, principal-agent and relational contract theory. Hart and Holmstrom (1986) claimed that "the collection of contract theories is partly a reaction to our rather thorough understanding of the standard theory of perfect competition under complete markets; perhaps because of the realisation that these paradigms are insufficient to accommodate some important economic phenomena".

For instance, both neoclassical and transactional cost economic theories assert the opportunistic attributes or self-vested interests of parties bound by costs in carrying out an exchange. Diamond (1984) argued that often, contract theories are drawn from economic bargaining power and imperfection as a natural way to enrich and amend an idealised competitive model. Fundamentally, contracts establish contractual relationships to parties and economic analyses that facilitates *quid pro quo* i.e. a favour in return for an action or deed. This study seeks to review principal-agent theory, modern contract theory and relational contract theory, the reasons being that the study's variables relate to situations where clients use their vested power (commercial bargaining power) to withhold payments from their sub-agents (mostly contractors, subcontractors and suppliers).

Principal-Agent theory seeks to maximise one's utility, an ideology where the principal uses his or her commercial bargaining power to utilise value over subagents, in this case of contractors. This theory is vital to this study because most literature about unfair payment practices reveals that most clients use their commercial bargaining power and vested interests to withhold money from subcontractors, because the clients are principal agents in contractual arrangements. Arguably, the sub-agents (contractors) rarely seeks readdress (legal means) from the principal agent because of their relationship and business continuity.

Relational Contract theory was put forward by Macaulay (1963), emphasising the continuity of contractual relationships. The theory states that contractual parties do not use the complete contract when they feel their counterpart is a necessary part of their business. The theory emphasises that contractual parties prefer to negotiate and re-arrange a contract in order to keep its continuity rather than terminating it. Perhaps this theory perfectly explains why most contractors in the construction industry involved in unfair payment practices will not tolerate the option of termination (although this is an option in a contract) because of business continuity.

Arguably, modern contract theory was put forward in early 19th century England during revolutionary economic and political changes (Gluck 1979). Possibly, modern contract theory would not be complete without reference to 19th century classical Liberal thinkers such as Adam Smith, John Stuart Mill, James Mill and Jeremy Bentham. The most influential of these Liberal thinkers was Adam Smith who outlined the doctrine of economic *laissez-faire* (that is, allowing things to take their natural course without interference).

This doctrine became the cornerstone of 18th century economic thinking. The theory proposes that "if individuals were allowed to pursue their self-interest free from government interference, they will inevitably maximize their own profits and, therefore the wealth of society as a whole". In addition, Adam Smith insisted that government should only have a negative role in economic life (national defence and internal security) because competition among individual self-interests would result in social harmony through the agency of the "invisible hand".

Adam Smith's supposition is not significantly different from principal-agent theory, possibly because the two theories emphasise domineering power and profit maximisation by one agent over another. However, Adam Smith's theory is relevant to this study because of his assertion that government should only have a negative role in economic life. In contrast, the government's role should not only be negative but rather a positive, regulatory role. For example, the enactments of the Housing Government Construction Regulation Act, Local Democracy Economic Construction Act, Construction Contract Act, etc. are positive roles placed by the government to regulate unfair payment practices.

3.2.1 Historical Perspectives on Standard forms of Contracts

Most economists and lawyers believe that the Adam Smith doctrine gave birth to modern contract theory *vis-a-vis* standard forms of contracts (SFoCs). Kessler (1943) argued that the principles of contracts enunciated in the 19th century worked well, in as much as freedom of entrance (i.e. offer, acceptance, consideration, fraud, duress, misrepresentation and *non est factum*) are acceptable defence in contracts. This rule was imperative at the time because contracts were negotiated on an individual basis. Gluck (1979) asserted that while the above contract principles existed; the process of contract formation had

changed considerably by the turn of the 19th century, due to direct responses to the economic realities of the times.

Production was seriously driven by manufacturing in large quantities and standardisation became an integral part of the economic order. Profit margin was linked to efficiency and it quickly became obvious that manufacturers did not have time and money to enter into individually negotiated contracts. The preferred solution at the time was the introduction of printed, mass-produced contracts that could be used repeatedly. It could be argued that SFoCs were a complementary and natural development from mass-produced contracts (Kessler 1943).These forms of contract were subsequently developed by various interest groups into longer and much more complex forms used in virtually every aspect of commercial life today.

From the standpoint of risk allocation, the literature has arguably been scarce about the determinants of contract agreement's usefulness. This is imperative because valuable insights have emerged that contracts are generally imperfect, and that they do not fulfil *Pareto* optimality theory. *Pareto* efficiency or optimality theory explains a state of allocation of resources from the point of view that it is impossible to reallocate preference criterion that makes one party better off, without making the other party worse off. In reality, the nature of construction contracts usually favours clients and their representatives and risks are often transferred to contractors. For example, contractors normally bear the payment risks allocation in a typical construction contract.

Thus, it could be argued that *Pareto* optimality is at variance with the very nature of construction contracts. Sharkey et al. (2014) posited that standard forms of contracts in construction are frequently amended to move risk to parties that have weaker bargaining power; thus alluding to contracts of adhesion (take it or leave it); which are common phenomena between main contractors (acting as clients) and subcontractors. Theoretically, the fact that SFoCs are often drawn in advance makes them fit into "Enforcement of the contract theory or asymmetric information

contract models". These models explain that the enforcement of a contract based on lack of information may subsequently warrant breach of a contract. For example, project's parties will agree in advance to have a firm and binding contract, without the flexibility to change or alter the clauses once signed.

It is often the case that employers will allow for changes in a contract; therefore, a typical construction contract is prone to late payments (Zaghloul and Hartman 2003). Yet, the literature's references to standard forms of contract portray a perfect contract position. In practice and from a legal perspective however, there are rarely standardised forms of contract due to variations and amendments that often characterise construction contracts. Nevertheless, it is pertinent to notethat there is barely any explicit contract theory that expounds on performance indicators of SFoCs.

Figure 3.2.1 illustrates SFoCs that are mostly used in the UK. Whilst JCT and NEC forms remain the most popular and commonly used in construction, bespoke forms are still predominant although their use has fallen over the past few years. Chappel et al. (2008) argued that regardless of their nature and extent, all SFoCs consist of payment provisions. These provisions are usually guided by the Housing Grants, Construction and Regeneration Act (1996) that emphasises adequate mechanisms or procedures to secure payment for contractors and other members within the supply chain. SFoCs such as JCT(SBC/Q2016), NEC4 and FIDIC used in the construction industry will be discussed in subsequent sections.



Figure 3.2.1 Most utilised SFoCs in the UK construction Industry (adapted from National Construction Contracts and Law Report 2018)

3.2.2 Payment provisions in JCT Standard Building Contracts

In contrast to its subsequent forms, the JCT (2016 SBC/Q) introduced significant changes in payment provision to all main contracts, subcontracts and sub-subcontracts forms. The changes were deemed to supersede government payment regulations, charters and codes by synchronising payment processes and provisions throughout all tiers of the construction parties. The JCT 2016 aimed to deliver a fair payment system that would secure cashflow for the main contractor down to subcontractors and sub-subcontractors within 30 days in industry terms. These changes to JCTs constitute the introduction of an interim valuation date (IVD), the reduction of the final date for payment to 14 days (as opposed to 21 days from the due date) and the valuation date remaining consistent for contractors in the project.

As for IVDs that modify and restructure JCTs, they establish a common valuation date for all parties to the contract and all tiers involved in the building project. Clause 4.7.3 of JCT requires a contractor to submit an interim payment application before the IVD and if the application is late, the due date for the interim payment would be 7 days after the relevant IVD. As a quantity surveyor (QS) validates the interim application, the employer is required to issue payment notice not later than 5 days (Clause 4.7.2) after the due date. Moreover, if the employer wants to issue a payless notice on a contractor's application as illustrated in figure 3.2.2, a notification must be made within 5 days (Clause 4.9.5) before the final date for payment. Again, the JCT stipulates both the final date for an interim payment and the final payment, which must be issued not later than14 days from the due date (Clause 4.9.1).

The main purpose of an IVD is to streamline payments for all the tiers by setting a due date that follows the same valuation date. For example, JCT subcontracts set a due date of 12 days after the IVD, that will require the main contractor to pay a subbie for outstanding invoices; and the same for sub-sub-contracts, setting the due date for 17 days after the IVD.

Arguably, the general issue remains that over 60% of SFoCs including JCTs are often amended by the employer and main contractor to reflect their self-interests, which in turn affect the payment arrangements of the supply chain. This is evident with most of the work procured by Tier 2 clients (for instance the liquidated construction giant Carillion); where bespoke forms are mostly drafted to protect their cash flows by inputting longer payment terms and unfavourable practices or measures on subbies and suppliers.



Figure 3.2.2 Payment Timeline – JCT Design and Build Contract 2016

3.2.3 Payment Provision in NEC4

In NEC 4, payment provision and procedures are synchronised in Clause 50 and secondary optional clauses for six of its main options (i.e. a priced contract with an activity schedule, a priced contract with bills of quantities, a target contract with an activity schedule, a target contract with bills of quantities, a cost reimbursable contract and a management contract). For instance, clause 50.2 of NEC 4 stipulates its contractor's obligation to submit an interim application to the project manager before the assessment date. Failure to do so would penalise the contractor in accordance with the contract (clause 50.4).

The project manager is responsible of assessing and certifying the amount due to be submitted at each assessment date not later than one week from its assessment (clause 51.1). It must be noted under NEC4, assessment dates are made at intervals of no more than 5 weeks (clause Y2.2) and determined by the project manager himself or herself.

Again, the final date for payment must be made within 3 weeks of the assessment date (clause 51.2). In the event of failure, interest would be charged on late payments (clause 51.3). Overall, all payment procedures and provisions relating to NEC4 are dependent on the project manager's due diligence to adhere to timescales and conditions (Mcleron and McClements 2015), which in a practical sense is burdensome and risky, given the nature of the construction industry.



Figure 3.2.3 Payment Timeline – NEC 4 Contracts

3.2.4 Payment provision in FIDIC

Clause 14 of FIDIC forms provides common and fundamental payment provisions for managing costs. This includes assessing amounts at 8 week intervals (Clause 14.7), the certification of payments within 4 weeks of the due date (Clause 14.6) and issuance of a taking-over certificate within 12 weeks of the contractor's completion. The clause also provides provisions for advance payments (Clause 14.2) and remedies the non-issuance of interim payments such as the suspension of works.

The project engineer, as the main certifier in FIDIC, is responsible for assessing and validating a contractor's interim application and advising the client on the amount to be included in the final payment. The contractor is obliged to submit an interim application to the engineer by the end of month specifying the amount of completed work and the details behind its assessment. Preferably, the contractor is advised to submit supporting documents (such as invoices, photos and receipts) that will substantiate his or her claim for monies. The engineer has 4 weeks to issue an interim payment certificate as stated in the contract. In addition, the client is subjected to pay the contractor within 56 days from the submission of the interim application to the engineer (see figure 3.2.4).



Figure 3.2.4 Payment Timeline – FIDIC Contracts

Overall, the reviewed standard payment terms for JCT, NEC and FIDIC contracts stand between 21 and 28 days, indicating an approximate 30-day contract period, which may seem reasonable. FIDIC meanwhile has an average payment period of 56 days from the day of certifying an interim valuation. However, there is a need to bear in mind that FIDIC has provision for advance payments at 42 or 21 days respectively.

Conversely, it could be argued that payment provisions stipulated in SFoCs rely on a contractor's or client's due diligence to adhere to timescales, conditions and responsibilities that are uncertain. Normally, a main contractor would be handling a multitude of projects and dealing with an array of subcontractors that would place a considerable amount of risk and uncertainty as to whether provisions would be adhered to or not. Moreover, it becomes an overarching objective when terms are amended or tailored in the contracts. For instance, in Henia Investments Inc. v. Beck Interiors Ltd. (2015), the court pointed out the failure of both parties to understand the SFoCs wording of "payment in accordance with timescale in the contract" that escalated to a dispute concerning the validity of an interim application payment submitted by the contractor. On the contrary, there is no literature that suggest that SFoCs have better payment performance; but the nomenclatures and fundamentals of NEC standard form of contract has potential to encourage better payment practices.

3.2.5 Retention and Unfair payment practices in Construction

Most standard forms of contracts (SFoCs) in construction incorporate provisions that govern the use of retentions. Retentions are usually a percentage sum of a construction contract deducted and retained by the client; with half released during the practical completion and other after the expiry of the defects liability period. Retentions, as industry practices, are designed to provide insurance against contractor failure or insolvency risks to a project. Hughes et al. (1998) argued that typically, retention is set at 3-5% in most SFoCs; whereby a client withholds part of the payments due to the main contractor who then deducts the same amount of monies due to their subcontractors.

This normal practice is deemed to encourage project efficiency, yet evidence suggest that withholding practices are proven to intensify traits of unfair payments in construction. For instance, a recent report by Pye Tait (2017) asserted that over the past five years, £7 billion in unremitted retentions had been withheld from UK contractors. Significantly, 65% of surveyed contractors, mostly tier 3, had experienced longer delays and even non-payment of retentions monies due to

upstream insolvencies. Perhaps it could be argued that retentions provisions stipulated in SFoCs have detrimental effects on a subcontractor's cash flow because of uncertainties that surround client's finances in the event of main contractor insolvency. This is evident with the recent collapse of Carillion Plc and Dawnus construction, whereby a total of £700 million and £50 million in cash retentions was written off by their supply chain.

3.2.6 Procurement and Payment Practices

The fundamental principle of most procurement strategies in construction is the ability to apportion design, time, variation and cost risks to the contractual parties. The issue about who takes the risk and the certainty of payments differs from the procurement form selected by the client. Laryea and Hughes (2006) and Sherif and Kaka(2003) argued that the connection between procurement strategies, the suitability of the selected payment mechanism and uncertainties associated with cashflow would affect multiple layers of contractors and subcontractors involved in the project.

The industry's fundamental problems are its dependence on a contractor's cashflow and the risks associated in the financing of projects. Ardti and Yasamis (1998), Sherif and Kaka (2003), Hughes et al. (2006) and Masterman (2002) were of the view that regardless of the procurement strategies, contractors are expected to use their resources to execute the job and wait for payment later. Moreover, the use of management contracting procurement in the UK is seen to encourage unfair payment practices as the main contractors retain the upper hand in commercial power, with a domineering position in the supply chain to dictate and/or control payments to subcontractors/suppliers.

3.3 Legislative Provisions

Following the recommendations made by Latham (1994) and Egan (1998), the UK enacted different payment legislations and acts to address unfair payment practices in the industry. For instance, the promulgation of the Housing Grants, Construction and Regeneration Act (HGCRA 1996) was deemed as an initial step

to enhance prompt and timely payments within the supply chain. Griffiths et al. (2017) argued that although legislations have been modelled to varying degrees, UK is renowned for spearheading the development of most payment legislations. Table 3.3 presents various acts and legislations enacted by different countries to improve payment practices, with a brief discussion of UK payment legislations which will be discussed subsequently.

Table 3.3: Different payment legislations and acts enacted in different countries

No.	Country	Legislations and Acts	Function	Payment duration/
	, ,			response
1	NORWAY	Norwegian building and civil engineering contract NS 8405:2008 and general conditions of contracts for design and build contracts NS 8407 (2011)	Entitlement to progress payment	Payment duration and response must be made within 30 business days (BD) or must be agreed by the parties to the contracts
2	NETHERLAN DS	Uniform Administrative Conditions for the Executions of Works and Technical	Improve payment practices	Payment claims must be made within 30 days after progress claim is submitted and notice of
3	NEW ZEALAND	Construction Contracts Acts (2015)	Facilitate prompt payments, speedy dispute resolution and remedies for non- payments and retentions	Payment becomes due within 20 working days after receipt of invoice
4	SINGAPORE	Building and Construction Industry Payment Act	Improve cash flow	Payment schedule and response must be made within 14 days of the assessment date
5	AUSTRALIA	BuildingandConstructionIndustry(Security of Payment)Act (BCIPA)2009ConstructionContractsAct (2004)	Facilitate regular and timely payments	Payments become due within 40 days from receipt of invoice and a notice to "reject / dispute part of claim" must be given prior to the final date
6	USA	Miller Act and Prompt Payment Act	Enhance progress payments within the supply chain	Payment is made 30 BDS and 15 BDS from the time the claim is made
8	CANADA	Bill 69 Prompt Payment Act 2014	Entitlement to progress payment	Payments to contractors and subcontractors become due within 20 days from when payment claim is submitted
9	MALAYSIA	ConstructionIndustryPaymentandAdjudicationAct (2012)	Restrict conditional payment practices ('pay when paid' and 'pay if paid')	Same as UAV 2012

3.3.1 Payments under the Housing Grants, Construction and Regeneration Act (1996)

The development of the Housing Grants, Construction and Regeneration Act (HGCRA1996) was influenced by negative effects of the industry's traditional payment and withholding system. Latham's report (1994) highlighted the industry's cascade payments and withholding practices that had exposed contractors, subcontractors and so forth down the supply chain to late payments. As orated in his seminal report titled 'Constructing Team'; "... the cascade system of payment in the industry ...makes exposure of other parts of the process to insolvency and later payments of participants". Therefore, in recognising this detrimental effect, the government enacted HGCRA to expedite timely and regular payments as well as encourage speedy dispute resolutions for parties.

Under section 109 of HGCRA (1996), a party to a construction contract with an excess of 45 days in duration is entitled to interim, stage or periodic payments for any works. Periodic or interim payment is essential for contractors (mostly subcontractors) due to regular injections of cash into the project. As stated by Lord Denning in Dawnays Ltd v FG Minter Ltd, regular payments (cashflow) is "the lifeblood of construction".

The HGCRA, also known as the 'Construction contract', stipulates a requirement for contracts to provide an adequate mechanism for determining what payment become due and when. Section 110 (1) (a) of the act states that parties must agree on the payment process and intervals at which they become due and the final date for payment claims to be made. In addition, the act also ousted conditional payment clauses (such as 'pay when paid' and 'pay if paid') and withholding notices by payers. Griffiths et al. (2017) affirmed the positive effects of the legislation in improving payment culture and practices in the UK, yet late and delayed payments continue to impact small and medium-sized contactors.

3.3.2 Payments under the Local Democracy, Economic Development and Construction Act (LDEDC)

The Local Democracy, Economic Development and Construction Act (LDEDC) 2009 came into effect in both England and Wales in 2011.TheLDEDC Act constitutes changes within the operation of construction contracts that focus on amending payment provisions set out in section 110 and 111 of the Housing Grants, Construction and Regulation Act (1996); suspension of works upon non-payments; adjudication and the inclusion of unwritten contracts.

As for the revised sections, the LDEDC made payment provisions clearer and more straightforward together with the eradicating of conditional payments that had such a huge impact on subbies. This consisted of payment notices to be issued within five days prior to the due date together with specifying the notified sum. The payee is entitled to serve a payment notice in any given circumstances, the payment notice must contain the basis upon which the notified sum is calculated and the payer must give notice if they intend to pay less than the notified sum and state the withholding amount. Besides, the act does not alter the provision of an adequate mechanism in determining payments and intervals together with the entitlement of the party to receive regular payments.

Overall, the LDEDC act calls for a new payment notice regime that involves the issuing of payments and specified lesser amounts within five days prior to the due date, the introduction of pay less notices and the bolstering of payee rights to the suspension of late payments (RICS 2015).

In essence, both the HGCRA and LDEDC were formulated with the intention of facilitating timely payments; the right to receive progress payments; the prohibiting of conditional payments; and solving payment disputes by adjudication together with inducing payment rules and notices within the standard forms of construction contracts.

3.3.3 Payments under the Scheme of Construction Contracts (England and Wales) Regulations 2011

In contrast to previous legislations and acts, the Scheme of Construction regulation provides fall-back provision for a contract that does not constitute an adequate payment mechanism. Ancillary to HGCRA (1996), the scheme sets out provisions for parties that have a written contract but with void payment rules and non-compliant payment provisions. Pettigrew (2005) argued that although the scheme appears to be complex and confusing at times, it provides a proper mechanism for the valuation of completed works and payments.

The provisions stipulated under the scheme include the right to interim or stage payments, the notification of payment notices, the suspension of performance due to late payments, the omission of withholding notices and conditional payment clauses. In essence, the scheme focus on notice requirements within construction contracts, whereas the payer must give a payment notice not later than 5 days after the payment due date and pay less notice before the final date for payment. This distinction was outlined from the 1996 act, whereby the withholding notices had a consequential effect on payee invoices. The scheme replaces the regime with a pay less notice to encourage positive and good payment practices by the payer.

Undoubtedly, despite these legislative interventions, anecdotal evidences suggest that late payment problems continue to persist in the UK, particularly within the construction industry, whereby 60% of small suppliers witnessed payment terms increasing over 60 days and beyond. Wood (2016) argued that most acts are enclosed in complex processes that tend to be difficult to apply and have benefited only those who fully understood them. This is evident for most lower-tier parties (sub-contractors and suppliers) within the construction industry that lack of knowledge and understanding of the regulations and acts, such that they merely use the stated provisions.

Moreover, most subcontractors and suppliers are reluctant to employ payment provisions such as interest on late payments or suspension due to non-payment because they fear being removed from the Tier 2 supply chain.

3.3.4 Public Contracts Payment Regulation 2015

The public contracts regulation 2015 stipulates payment provisions and procedures for public bodies when dealing with and paying suppliers that commit to fair and good practices in their supply chain. The legislation was specifically enacted by the government to promote fair and consistent payment periods and ethical commercial behaviours within public supply chain contracts.

Regulation 113 of the contract imposes a duty on public bodies to ensure that payments to their Tier 1 contractors are made within 30 days andthat same effect is contained in each tier of the supply chain. The mandate of the 30-day payment rule is deemed to be applicable to all public construction contracts and payments made later than the 30 days term will be subject to statutory interest.

3.4 Construction Payment Charters and Codes

In recognition of this late payment culture, the UK government has enacted different codes and charters to minimise the occurrence of such financial detriments within construction. These codes and charters are designed to create transparency and fair payments for parties working on construction projects. Specific to supply chain payments and finance, the charters and codes are focused on enhancing prompt and timely payments, 30 days of industry's terms, zero retentions and good practices within construction contracts. The following codes and charters were introduced by the government in line with the Construction 2025 strategy that aims to "... create conditions for construction supply chains to thrive by addressing access to finance and payment practices...".

3.4.1 The Prompt Payment Code

Formulated by the Chartered Institute of Credit Management (CICM) the Prompt Payment Code ("the code") calls for changes in payment culture across all businesses, particularly construction, where payments are often delayed. The code, as a representation of the UK government's image, sets out principles for the construction industry to follow when dealing with and paying their suppliers within an agreed time frame.

The code requirements stipulate three specific areas that signatories must adhere to and commit fair and transparent practices within their supply chain. These requirements include paying suppliers on time, giving clearguidance to suppliers and encouraging good practices within their supply chain. Moreover, the code calls for signatories to the three requirements to undertakereductions in payment timelines from a maximum of 60 days towards 30 days as an industry norm and to avoid any unfair practices that could adversely affect the supply chain.

The FMB (2017) argued that although the code had gained the majority of Tier 2 signatories including the Top 100 construction companies; it had failed to readdress the late payment problems facing many suppliers. This is highlighted by the recent demise of Carillion which enforced payment terms of 120 days and beyond, despite being a signatory to the code.

3.4.2 The Fair Payment Charter

The fair payment charter sets out values and arrangements relating to fair and transparent payment practices for construction contracts. The charter builds on existing legislations and policies that aim to create a collaborative culture and sustainable supply chain in projects. Administered by the Construction Leadership Council (CLC), the charter sets out 11 commitments to be adhered to by clients and contractors working to achieve the best and fair practices within their supply chains.

The 11 commitments stipulated by the Fair Payment Charter include the following: the right to receive full payment, fair payment, transparency, open book accounting, equitable distribution of cash flow, 30-day payment rule, issuance of pay less notice and prohibiting the withholding payment rule, etc. (Office of Government Commerce 2007). Again, the charter entitles the use of the document as not being legally binding for its members and that it should not

to be used in constructing any contractual commitment. In other words, it implies that the charter is voluntary for its members to commit to and must not be intended as a legal document.

Cartlidge (2013) argued that although nine companies had committed to adopting the charter, there are concerns regarding how the charter would be policed and enforced, since there is no legislation or sanction governing it.

3.4.3 The Construction Supply Chain Payment Charter

Corresponding to the Prompt Payment Code, the Construction Supply Chain Payment Charter (CSCPC) seeks to establish fair and transparent payment practices for construction projects. The CSCPC stipulates 11 fair payments commitments that aimed to reduce supply chain terms to a 30-day payment rule by 2018 and zero retentions by 2025 (Build UK 2017). The CSCPC again called for payments not be made after more than 60 days, for all construction contracts for works carried out or supplied from 2015.

Moreover, other commitments stipulated by the CSCPC include the abolition of the withholding payment rule, the use of project bank accounts and electronic payments, zero retentions and the adoption of a supply chain finance scheme for all parties. Mason (2016) argued that contractors and organisations that had signed up to the charter should be monitored against a set of key performance indicators (KPIs) for the purposes of compliance and the performance of their supply chain when awarding contracts.

Overall, it could be argued that the voluntary approach imposed in these aforementioned charters has proved to be a drawback in readdressing payment problems that sit at the heart of contractors' businesses and the industry's economic structure. The Department for Business, Energy and Industrial Strategy (2017) report asserted that it was clear that both codes and charters were having no desired effects or sanctions on companies that had signed up to or adhered to its stipulated principles, requirements or commitments.

3.5 Alternative Measures of Enhancing Fair Payments in Construction

The use of administrative measures in construction is deemed to be an industry attempt to deal with payment problems that affect most contractors. The measures are normally included in SFoCs and construction contracts to minimise and provide remedies related to late payments. For instance, JCT 2016 Design and Build affirms the use of escrow (or trust) accounts to hold money, particularly retentions for completed works on site.

An escrow (or trust account) is a ring-fenced account set up by client where a sum of money is deposited for the project. It is normally managedand delivered by a third party (escrow agent) in a condition precedent for a contractor's performance in the contract. Latham (1994) argued on the use of escrow or trust accounts as form of protection against a client's insolvency, whereas if the client failed to pay the amount due to the contractor, the escrowfund would release payment according to the progress of the works.

Furthermore, Hughes et al. (1998) affirmed the withholding of money in a trust account to be a powerful means of returning contractors to a site and rectifying defective works. On the contrary, it could be argued that in construction, holding money in a single account creates cash flow pressures on a contractor's, particularly Tier 2, which results in unfair payments. Moreover, Davis(1991), Hughes et al (1998), Supardi et al (2011), and Ramachandra(2013) argued that the use of bonds, guarantees, insurances, charging orders, notices and direct payments as security for subcontractors and suppliersagainst insolvency nevertheless risks unfair payment practices by clients or contractors in a project.

Arguably, a significant feature in all administrative measures has been the use of project bank accounts (PBA) in readdressing the late payment problem. Kilgallon (2013) affirmed a PBA as a ring-fenced bank account from which payments would be made to ensure that the contractor and its supply chain received timely payment for the amount due. The PBA model, as compared to an escrow account, is either held by the contractor himself or herself (the single authorisation model) or both client and contractor (the dual authorisation model) for a particular project (Griffiths et al. 2017).

Under a PBA, the subcontractor submits an interim application to the main contractor showing a breakdown of payments claimed by each supplier with the purpose of counterchecking if necessary, should any adjustment be required by the contractor. Once approved, the client will pay the total amount of monies into the PBA, through which payments are directly made to each member of the supply chain. If the client wants to reduce or adjust payments due to a contractor's defaults, the contractor will be required to make a top-up payment to the account so that the subcontractors and suppliers can be paid on time. Moreover, to avoid a deficit of payments, a client must ensure that the balance remains at an agreed minimum level under a PBA (Kilgallon 2013).

Arguably, the novelty behind the aforementioned PBA model has been the government's recommendation to achieve public spending savings and readdress chronic unfair payment practices in the construction supply chain. The National Audit Office report (2005) posited that "... there is need to provide specialist small and medium sized contractors or suppliers with greater certainty that they will be paid on time ... if this does not exist the supply chain will have little incentive to innovate". However, a serious number of concerns and dissatisfactions have been raised by practitioners concerning the administrative burdens and the costs of setting up PBAs. In particular, strong opposition has been queried by Tier 2 clients (main contractors) as it undermines their commercial advantage to withhold payments.

3.6 Potentiality of Emerging Technologies in Minimising Late Payments

Peter and Arewa (2017) stressed that there was no gainsaying that emerging technologies (such as 5D-BIM, Automated Payment Systems (APS), Smart contracts, Agresso Unit 4 and Artificial Intelligence) had the potential to minimise late payment problems in construction. For instance, the use of the 5D-BIM model enables users to extract comprehensive and accurate cost information by creating a relationship between the elements, specifications and

properties of individual elements and objects within the project. Boon and Prigg (2012) defined 5D-BIM as "data—rich" objects that support the function of cost modelling; assemblies of cost components added either by incorporating cost data in the model itself or "live-linking" it to estimating software tools. The 5D-BIM model itself links the cost data to the information model (normally 3D) that forms the foundation of a living cost plan (Mitchell 2012). This allows simultaneous changes to finance, funding, budget variances, cost forecasting, investment decisions as well as negotiations with contractors. Forgues et al. (2012) argued that such features assisted practitioners (especially quantity surveyors) to achieve cost certainty, effectivedesign, time savings in take-off of quantities, estimation; transparency and increased control and predictability of project stakeholders.

Moreover, other contemporary technologies have been proven to have great leverage over construction payments. Barber (2012) posited that digital devices such as Automated Payment Systems (APS), Smart Contracts and Agresso Unit 4 provided opportunities for instant and timely payment practices that were accurate, efficient and easily interrogated within project base organisations. These devices have dynamic links to modelled information that allow for easy recompilation of quantities that enhance the progress of payment calculations visible to the supply chain.

Peter and Arewa (2017) were of the view that these technologies had potentials to minimise administrative inefficiencies, inaccurate valuations, bureaucratic tendencies, multi-tiered hierarchical structures, delinquencies and the industry culture of 'pay-when-paid' that influence late payments to construction contractors.

3.7 Contractual Relationships and Unfair Payment Practices

In most construction practices, there exist three types of contractual relationships: client to main contractor (C-MC), main contractor to subcontractor (MC-SC), subcontractor-sub-subcontractor (SC-SSC) and sub subcontractor-suppliers/vendors (SSC-S/V). Although each of these working

relationships are interdependent, most are characterised by a high degree of fragmentation such as less cooperation, limited mutual trust, self-vested interests, ineffective communication and hostile-dependent relationships, etc. (Chan et al. 2004; Clough et al. 2015; Moore et al. 1992). For instance, within both main and subcontractor boundaries, the subcontractors are expected to work for the main contractor and carry out a specialised area of construction works on their behalf, whereby the former is responsible for supervising and ensuring that the construction works are carried in accordance with the client's requirements.

Again, at the sub-subcontractor and sub-supplier contract's interface, the subcontractor will engage other sub-subcontractors to carry out the whole or a portion of the construction works while the latter will coordinate a group of suppliers or vendors to obtain construction works or materials. To be precise, all parties from the client, the main contractor and the subcontractors are integral to the process and are interdependent. Arguably, within the construction industry, most of these aforementioned relationships are characterised by either long-term or short-term business relationships or hostile-dependent relationships (Tan and Xue 2017).

To begin with, long-term business relationships are deemed to exist between C-MC or MC-C with the purpose of achieving specific business objectives and comparative advantages in their resources. Long-term business relationships are deemed to be used interchangeably with the term 'partnering' in construction as this emphasises greater performance and improvement through a collaborative or long-term business relationship. In other words, partnering is benchmarked by parties working together in an open, trusting and long-term relationship based on mutual objectives, agreed methods to problem resolution and continuous improvements (Construction Excellence 2004).

For instance, the ability of the main contractor to form long-term relationships with subcontractors stems from a good performance record and cooperation over a number of years that takes the form of relational contracting. Such relationships are deemed to be fair, adequate and well established by both parties, with the purpose of treating each other as long-term partners or teammates (Eom et al. 2015). Similarly, a long-term business relationship occurs between the client and the main contractor if the project goals (cost, time and schedule) are constantly satisfied by the contractor or a similar partner that undertakes construction activities for clients.

As for short term or one-off business relationships, these are formed mostly between MC-SC or SC-SCC as a result of one party or member possessing specific abilities or competencies in carrying out specialised construction work, then disbanding once completed (Chiang 2009; Dainty et al. 2001; Moore et al. 1992). Although most of these relationships are one-off or specific to a project, most are deemed to be somewhat adequate and cooperative between the main contractor and the subcontractor.

A hostile or unsatisfactory relationship exists between SSC-supplier; as it focuses on individual interests and commercial gain as opposed to the former (Chalker et al. 2016). Relationships between SSC-supplier are thus conflicted. Figure 3.7 depicts the nature of client-contractor relationships between C-MC, MC-SC, SC-SSC and SC-SCC. It could be argued that long-term businesses have contributed to an adequate relationship with the upper tiers in construction while the lower tiers are seemingly susceptible to short term and hostile-dependent relationships.

For instance, the Group Scape (2019) survey asserted that 93% of subcontractors and suppliers prioritised and protected the long-term relationships between themselves and Carillion, thus enabling the contracting company to hide its true financial difficulties. Arguably, most Tier 1 and Tier 2 client relationships are generally deemed to be cordial in terms of payments and their relationship can also be adjusted to be fair. However, the literature considers relationships between Tier 2 and Tier 3 or Tier 3 and Tier 4 to be chequered when it comes to unfair payments.



Figure 3.7 Nature of the Client-Contractor Relationship

3.8 Summary

Overall, this chapter has reviewed existing solutions and measures for late payment dilemmas in construction. The solutions and remedies were reviewed within contractual, legal and administrative as well as technological contexts. In terms of contractual provision, the chapter has focused on payment provisions in SFoC's that are used in the UK construction industry. Furthermore, within a legislative and administrative context, the chapter has reviewed various charters, legislations, codes and measures that attempt to minimise unfair payment practices. The chapter has highlighted that the efficacy of contractual and legislative provisions is to reinforce certainty of payment and protection for the supply chain in case of a client's or contractor's insolvency. Furthermore, there are proven potentialities (i.e. qualities that can likely be developed) within emerging digital technologies for minimising unfair payment practices in construction. Henceforth, the next chapter will explain the research methodology used in investigating the study problem identified in the reviewed literature.

CHAPTER 4

Research Methodology

4.1 Introduction

This chapter reviews the research methodology and methods used in this study. The elements described in this chapter are research philosophies, paradigms, approaches, methods, strategies, time horizons and data collection techniques and procedures. In addition, the chapter explains the validity and reliability of the research findings and ethical principles adhered to by the study.

Key elements of research methodology and methods are described in Saunders' et al. (2016) 'research onion' concept in Figure 4.1. The figure explains the connection between theories and arguments that inform the research and methodological choices for data collection. Churchill (1979) and Crotty (1998) ascertained it is essential for research to provide a direction (or means) for collecting and analysing data at a certain period of the study.



Figure 4.1 Research Onion Concept adapted from Saunders et al. 2016

4.2 Research Philosophy

A research philosophy is "a basic set of beliefs and assumptions" that guide the actions or worldviews of the researcher (Guba 1990). Burrell and Morgan (1979) claimed that whether consciously aware or not, a significant number of assumptions are made or encountered at each stage of a research that influences the understanding of the research questions. These philosophical assumptions include the following:

- Ontology: refers to assumptions about the form and nature of realities. In the broader sense, ontology is concerned with questions of existence about what truly exists, what does it look like, what units make a research problem and how does it interact? (Blaike 2000) For instance, do unfair payments truly exist? Which tier of construction clients are prone to unfair practices? Why does the current contractual or legislative measures fail to address the problem?
- Epistemology: refers to assumptions about knowledge, what is constituted as acceptable, valid and legitimate knowledge and how knowledge is acquired (Burell and Morgan 2012). As for this study, the research problem spans from systemic inquiry to a constructive view of knowledge. For instance, both descriptive and explanatory knowledge (statistical data, textual and visual, facts from interpretations and narratives) are all considered acceptable in illustrating the nature of the problem.
- Axiology: refers to values and ethics that influence a research process (Saunders et al. 2019). Does the researcher input some of his or her own presumptions to the studied problem? Does the researcher espouse his or her own values on the data collection and analysis of the results?

On the other hand, many researchers like Bogdan and Bilken (1998), Patton (2002), Neuman (2009), Lincoln et al. (2011), Mertens (2010) and Creswell and Creswell (2018) have argued that sometimes the term 'worldview' is referred to as paradigm.

4.3 Research Paradigms

A paradigm is a loose collection of logically related assumptions, concepts or propositions that guide and inform a researcher's thinking (Bogdan and Biklen 1998). MacNaughton et al. (2001) defined paradigms as beliefs about the nature of knowledge, methodologies and the validity of undertaking a research study.

According to Mackenzie and Knipe (2006), there are many research paradigms discussed in literatures, including positivist (and postpositivist), constructivist, interpretivist, transformative, emancipatory, critical, pragmatic and deconstructive. Definitions of some of the most common paradigms and the choice of the chosen paradigm will be discussed as follows.

4.3.1 Post-positivist Paradigm

Known as the scientific method or empirical science in research, post-positivism is a deterministic philosophy in which causes are used to determine effects or outcomes (Creswell and Creswell 2018). Post-positivism reduces ideas (or causes) into small discrete sets that can be experimented on or measured using instances of research hypotheses, questions, or theories.

In social science, O'Leary (2004) argued that post-positivism sees the world as ambiguous, variable and multiple in its realties. He described this notion based on human behaviour that can be vague and inconsistent, stating that "what might be true for one person or cultural group may not be truth for another". For instance, what is perceived to be fair to the client or main contractors may perhaps not be fair to subcontractors or suppliers, i.e. interest or discounts for prompt payments. He further suggested that this paradigm or worldview is intuitive, comprehensive, inductive and investigative in findings that are quantitative in nature.

However, Mertens (2005) contended that positivism gives emphasis to the assumption that the social world can be studied in the same way as the natural world. It is a method of studying the social world that is value free and thus explanations of a causal nature can be provided.

4.3.2 Interpretivist Paradigm

The interpretivist (or constructivist) paradigm is based on an interpretive understanding called hermeneutics (Mertens 2005). The interpretivist paradigm seeks to have an understanding of the world from given subjective experiences or backgrounds. Creswell and Creswell (2018) argued that interpretivists "focus on the specific contexts in which people live and work in order to understand the historical and cultural settings of the participants".

Hence, interpretivism requires researchers to provide open-ended questions and even observations that will obtain information and study the meanings which individuals have formed. Creswell (2007) argued that rather than starting a new theory as in positivism, interpretivists generate or inductively develop theory from their data collection methods. Mackenzie and Knipe (2006) ascertained that interpretivist research is hinged on qualitative data collection methods and analyses or a combination of both qualitative and quantitative methods (mixed methods).

4.3.3 Pragmatic Paradigm

The pragmatist paradigm emanates from actions, situations and consequences opposed by antecedent conditions as post-positivism. Patton (1990) and Creswell (2003) argues that pragmatism focuses on 'what works' and 'solutions to problems' instead of engaging in philosophical arguments about laws of reality (ontology), theories of knowledge (epistemology) or values (axiology). Thus, the pragmatic paradigm emphasises the use of all research approaches and methods to understand the research problem and questions (Creswell 2003).

Similarly, Creswell (2018) asserted that pragmatism holds the door to multiple methods, different worldviews and different assumptions, that form and permit the use of different data collection and analyses. Therefore, the pragmatic paradigm is seen as providing the underlying philosophical framework of mixed-methods research (Tashakkori and Teddlie 2003; Somekh and Lewin 2005).

4.3.4 The Transformative Paradigm

According to Mertens (2010), the transformative paradigm upholds that the research inquiry needs to be intertwined with politics and a political change agenda that confronts social oppression at each stage at which it occurs. In essence, research needs to contain an action agenda that will advocate and reform the lives of participants, the institutions where they work or live and the researcher's life as well (Creswell 2014). For instance, there are key issues that need to be addressed that impact construction contractors in the UK, such as unfair payment practices, the use of adhesion contracts, imbalance in the supply chain, cowboy tactics and asymmetric business relationships. Mackenzie and Knipe (2006) claim the transformative paradigm utilises both qualitative and quantitative (mixed) methods in same way as interpretivists/constructivists. Again, Somekh and Lewin (2005) affirmed that integrating a mixed methods approach in a transformative paradigm provides a complete understanding of "our social world through the use of multiple perspectives and lenses".

Based on the predefined objectives and questions of this research study, the pragmatic paradigm was chosen. This is because the nature of the research study itself, i.e. 'unfair payment practices', is investigative, practice oriented and a real industry problem in construction as illustrated in table 4.3.4a. Again, the researcher acknowledges 'unfair payments practices' are pluralistic in nature as there is more than one contributing factor and seeks to understand the 'what' and how' (research questions) using different methods, as seen in table 4.3.4b.

Positivist/ Postpositivist	Interpretivist/ Constructivist	Transformative	Pragmatic
Experimental Quasi-experimental Correlational Reductionism Theory verification Causal comparative Determination Normative	Naturalistic Phenomenological Hermeneutic Interpretivist Ethnographic Multiple participant meanings Social and historical construction Theory generation Symbolic interaction	Critical theory Neo-Marxist Feminist Critical Race Participatory Emancipatory Advocacy Grand Narrative Empowerment issue oriented Change-oriented Interventionist Queer theory Race specific Political	Consequences of actions Problem-centred Pluralistic Real-world practice oriented Mixed models

Table 4.3.4a Paradigm language commonly used in research (adapted from Mertens (2005) andCreswell (2003) cited in Mackenzie and Knipe (2006)

Table 4.3.4b Research paradigms, methods and tools (adapted from Mertens (2005) and Creswell (2003) cited in Mackenzie and Knipe 2006)

Paradigm	Predominant research methods	Example of data collection tools
Positivist/ Postpositivist	Quantitative	Experiments Quasi-experiments Tests Scales
Interpretivist/ Constructivist	Qualitative methods	Interviews Observations Document Reviews Visual Data Analysis
Transformative	Qualitative methods with quantitative and mixed methods.	Diverse range of tools – that will particularly avoid discrimination, e.g. sexism, racism and homophobia.
Pragmatic	Qualitative and/or quantitative methods. Methods are matched to the research questions and aim.	May include tools from both positivist and interpretivist paradigms, e.g. interviews, observations, testing and experiments.

4.4 Research Approach

Saunders et al. (2016) stated three principal approaches to research: deductive, inductive and abductive. The deductive approach is the collection of data with a view to testing or verifying one's theory. The deductive approach derives conclusions via a set of premises or analyses that contain some form of truthfulness or knowledge (Ketokivi and Mantere 2010). The deductive approach is also known as 'top-down', allowing the researcher to begin with a broad area and then move to specific suppositions that can be tested (i.e. collecting data from literature, then developing theories or ideas to test it).

On the other hand, the inductive is the process by which theory is developed after observations are made. Saunders et al. (2016) asserted that the inductive approach explores phenomena and identifies themes and patterns that will develop the research hypotheses. Therefore, the inductive is the converse of deductive, as it generates theories after observations as opposed to verifying existing theories (i.e. observations, patterns, tentative hypotheses, theories).

Miller and Brewer (2003) affirmed that the abductive approach uses both empirical observations and theoretical suppositions to create new ideas. The abductive

approach is a mixture of deductive and inductive approaches and its main characteristic is the continuous movement between the empirical world and the modelled world (Dubois and Gade 2002), with a continuous interplay of both concept and data (Van Maneen et al. 2007). In simple terms, the abductive approach seeks to discover new ideas and variables, rather than conform to existing theories and theory generation.

This study adopted the abductive approach because it matches the nature of the research objectives and the extent of the subject matter. The approach underpins the fact that it takes account of a sizeable amount of theories, literatures and knowledge concerning unfair payments. Moreover, the researcher uses different approaches to extract and reveal data from inductive and open-ended research settings whilst using deductive attempts to verify hypotheses (Yin 2003). Moreover, the abductive approach supports the nature of the research study as 'pragmatic' and attempts to integrate systemic empirical observations and theoretical models to understand the study problem.

4.5 Research Strategy and Time Horizons

Denscombe (2017) described a research strategy as a distinct logic, actions and goals driven to achieve a specific objective. It entails the process and the direction in which the research is to be conducted. Saunders et al. (2009) ascertained that the selection of an appropriate research strategy is guided by the research questions, the objectives, the amount of existing knowledge, time, the resources available and the underpinnings of the researcher's own philosophical assumptions.

Based on this criterion, different researchers have proposed various types of strategies. For instance, Trochim and Donnelly (2008) have classified research strategies into the following: experimental and non-experimental (surveys). Hair et al. (2007) described strategies as exploratory, causal or descriptive. Other strategies proposed by Saunders et al. (2009) and Crotty (2004) include case studies, action research, grounded theory, ethnography and archival etc. Table 4.5 provides a list of different research strategies corresponding to their purposes and methods used in research.
This study will use concurrent research strategy to provide a comprehensive understanding of the research problem. The concurrent strategy uses or merges both quantitative and qualitative methods to collect and analyse data in the study (Creswell 2003). Todd (1979) ascertained that the concurrent strategy allows researchers to have more confidence in their results, to stimulate the creation of inventive methods and find new ways of capturing a problem and understanding the phenomenon beyond the study.

Strategy	Purpose of Research	Method
Surveys and Experimental	 Measures trends, attitudes or opinions of social phenomena Gathers facts to test a theory Identifies the cause of something 	Quantitative
Case studies	 Understands the complex relationship between factors, for instance if they operate within a particular social setting 	
Ethnography	 Describes cultural practices and traditions Interprets social interactions within culture 	Qualitative
Grounded theory	 Analyses concepts or produces new theories 	
Action (or Narrative) research	 Solves practical problems Produces guidelines for best practice 	
Phenomenology	 Describes the essence of specific types of personal experience Understands things through the eyes of someone else 	
Concurrent Explanatory or Exploratory	 Evaluates anew Combines strategies or aspects Compares alternative perspectives of a phenomenon 	Mixed method

Table 4.5 Research strategies and their corresponding purposes and methods (adapted fromCreswell 2018)

Conversely, Saunders et al. (2009) ascertained that it is imperative to consider the time frame for the investigation of a research problem. According to Bryman and Bell (2015), there are two types of time horizon: longitudinal and cross sectional. A longitudinal study is a process that covers a period of time, often years or decades. In other words, longitudinal research is a design that collects data from a sample (or cohort) at a repeated number of times or observations.

Longitudinal study has the ability to answer questions about causes and consequences by comparing results obtained in different ways. Cross-sectional research is defined as a design conducted at a particular point in time, known as a "snapshot". A cross-sectional study collects data to make inferences abouta population of interest (universe) at one point in time. Punch (2013) however noted that with consistent change in a population's characteristics, the actual situation may not be much reflected by a cross-sectional study.

Arguably due to the time constraints of this research, a cross-sectional study was deemed to be appropriate. Saunders et al. (2009) ascertained that often, a cross-sectional study 'snapshot' uses descriptive surveys and employs the mixed method strategy as well.

4.6 Research Method

A research method refers to the procedures or techniques used to collect and analyse data. Myers (2013) defined a research method as a strategy of inquiry that moves from underlying philosophical assumptions to the research design and data collection. Research methods are broadly categorised into three methods: qualitative, quantitative or mixed methods (Saunders et al. 2019). The following subheadings will explain these methods briefly, followed with an explanation of the adopted research method.

4.6.1 Quantitative and Qualitative Methods

According to Mack et al. (2005), quantitative methods focus on data collection techniques (questionnaires) or analysis procedures (statistics) that generate numerical data. Quantitative methods seek to explain phenomena and analyses using mathematical methods. Easterby-Smith (2002) and Amaratunga et al. (2002) argued that quantitative methods are advantageous in the following: the search for causal explanations and fundamental laws, the formulation of hypotheses for subsequent verification, the simplification of a problem to facilitate analysis and the determination of reliability.

Conversely, Bryman and Bell (2007) argued that qualitative research methods were originally developed in social science to study phenomena that were not explained through numbers and indices, but through world views via human perceptions. Qualitative methods constitute part of interpretive paradigms that seek to "describe, decode, translate and otherwise come to terms with the meaning, not the frequency, of certain more or less naturally occurring phenomena in the social world" (Van Maanen 1983).

Yin (2003) ascertained that the strength of qualitative research lies in its explanation of complex issues within the natural setting of a research phenomenon, Skyes (1990) opined that qualitative methods offer greater flexibility and responsive interactions between researcher and participants.

4.6.2 Mixed Method

The mixed research method denotes the integration of both quantitative and qualitative methods in a single study. A mixed method research considers multiple viewpoints, standpoints, data collection and analysis techniques, using both quantitative and qualitative research (Johnson et al. 2007). The purpose is to draw the strengths and minimise the weaknesses of both, rather than replacing or using either method in a single study (Johnson and Onwuegbuzie 2004). According to Creswell (2003), using mixed methods in research helps to provide comprehensive evidence for studying a research problem and answering questions rather than using quantitative or qualitative methodologies alone.

Creswell and Clark (2018) argued that in overlapping mixed methods, research designs would vary depending on whether qualitative and quantitative data would be collected either sequentially or concurrently. Figure 4.6.2 illustrates the mixed method research and its core design (note \rightarrow stands for sequential and + for concurrent).

The sequential design refers to the collection and analysis of one form of data prior to a second form of data, for instance the QUAN \rightarrow QUAL concept, meaning the quantitative method is the lead data collection instrument. Quantitative data is first

collected and analysed, then qualitative data is collected, analysed and used to validate the quantative findings. The concurrent design however requires the collection and analyses of both quantitative and qualitative data in a single phase, as the researcher had to implement and prioritise both qualitative and quantitative methods to collect and analyse the data, then merge both sets of results for the overall interpretations.





4.7 Justification for the Chosen Method

The study adopted the concurrent mixed research method to enable the researcher to capture the study problem using multi-methods that help to achieve both the reliability and validity of the study's outcome. This is further deemed to be in line with the researcher's philosophical position and the strategies adopted to support the use of the mixed method.

The concurrent mixed method used in this study follows the precedent of the researcher collecting both qualitative and quantitative data, analysing them separately and then merging the results to obtain a more comprehensive understanding of the subject matter (Creswell and Creswell 2018). The key deduction of this method is that both qualitative and quantitative data often provide different types of information. For instance, detailed experiences of the participants can be captured qualitatively and the instrument scores from the quantitative results can be merged to interpretate the

findings (Creswell and Creswell 2018). Johnson and Onwuegbuzie (2004) identified that using the concurrent mixed method could help to strengthen both confirmation and validation in a single study.

Thus, the study's quantitative data were collected from both archival documents and questionnaire surveys. The study's archival data relating to payment performance and practices was obtained from reputable sources such as Build UK, individual companies' published data and the Companies House Website. In addition, a semi-structured interview was carried out to gain an in-depth standpoint on the prevalence of unfair payments in UK construction. Table 4.7 represents a flow chart of the research method that denotes the use of the concurrent mixed method in the study.





4.7.1 Quantitative Method of the Main Study

To meet the aim of study, both a closed and open-ended questionnaire was developed and used to gather data from participants regarding their perceptions and opinions about unfair payment practices in the construction industry. Saunders et al. (2009) and Bryman and Bell (2007) argued that questionnaires are considered to be one of the most commonly used methods or tools for eliciting data from a sizeable population in research.

The authors further state that questionnaires allow the researcher to extend the knowledge and findings concerning the research problem prior to understanding the subject. However, both the nature of this study and the objectives and population sample influenced the researcher's decision to use a questionnaire survey as one option for the data collection. Thus, the following sub-section will provide an explanation of the questionnaire design, the selection of the participants, the piloting of the questionnaire, the data collection procedures, the measure scale and design of the questionnaires and the strategies used to improve them.

4.7.1.1 Questionnaire Design

The questionnaires for this study were designed to answer a predefined aim and the research questions (Burgess 2001). A semi-structured questionnaire constituting both closed and open-ended questions was used to capture all the necessary information needed for the study. Open-ended questions were used to give the participants a degree of freedom in answering and in gathering their insights concerning unfair payment practices in the UK construction industry. The closed-ended section in the questionnaire meanwhile was designed to elicit specific response from them (Fellows and Liu 2008). The closed-ended questions were left to the participants to input or give their own wording of their answers or experiences to the study problem and the questions asked.

The researcher developed two set of questionnaires for the study, whereby questionnaire survey 1 captured the general understanding of the research problem and questionnaire survey 2 measured clients' perceptions of it. The two sets of

questionnaires were distributed at different time intervals to enhance the study's reliability. Questionnaire survey 1 consisted of two sections. Section 1 consisted of 18 questions with corresponding likert scales aimed at obtaining participants' perceptions or opinions of unfair payment practices in the construction industry. This section included both multiple and open-ended questions that yielded in-depth information and insights from the participants. The second section consisted of demographic information such as professional designation, years of experience, form of contract, nature of business and type of organisation. Questionnaire survey 1 consisted generic question about unfair payment practices and covered the study's objectives.

Similarly, Questionnaire 2 consisted of two sections. Section 1 consisted of several likert-scales and multiple-choice questions, while section 2 consisted of demographic data. Questionnaire 2 was more specific with its questions regarding the nature and extent of unfair payment practices, payment durations and clients' perceptions of unfair payment practices.

4.7.1.2 Piloting of the questionnaire

The designing of the questionnaires was carried out in two phases. The first phase consisted of sample questionnaires for piloting purposes. This phase included obtaining information needed for the questionnaire, the type of questionnaire and method of administration, the content of the questions (knowledge, behavioural or perceptions), the form of response, the sequence and wording, the layout and output and then piloting of the questions. Peat et al. (2002) argued that it is important for a researcher to pilot their questionnaires prior to data collection in order to appraise the reliability of the questionnaire's findings.

Critique and feedback from the pilot study were used to refine and redesign the questionnaire to enhance reliability. See appendices A and B for questionnaires A and B, designed to measure the opinions of participants regarding unfair practices in the UK construction industry.

4.7.1.3 Measurement of the questionnaire

The measurement section of the questionnaire was designed to capture the perceptions of the study participants about unfair payments in the industry. Holt et al. (2013) suggested that the design of scale must consider the type of data it will generate as well as the analysis model that will be used.

The measurement scale of the study's questionnaire was designed using a likert scale to obtain the participants' opinions, perceptions and the extent to which they agreed with the statements in the questionnaire. A five point likert-scale (0 to 5) was used, ranging from (i) '0=very strongly disagree' (ii) '1=strongly disagree' (iii) '2=disagree' (iv) '3=agree', (v) '4=strongly agree' and (vi) '5=very strongly agree'. The "unsure" option was added and marked as = -'. Effects of unfair payment practices as independent constructs were also assessed using a likert scale, though the terms used were (i)'1=very low effect (ii) '2=low effect (iii) '3=moderate effects', (iv) '4=high effect agree' and (v) '5=very high effect'.

Table 4.7.1.3 below illustrates the designed questionnaire's measurement scale section and percentage scores for the codes assigned to each measurement scale.

Research question		Very strongly agree	Strongly agree	Agree	Disagree	Strongly disagree	Very strongly disagree	Unsure
andard forms of s-contractor n chronic actices in the dustry?	Number assigned to each response to question	5	4	3	2	1	0	-
What role does st contract and dien relationship play c unfair payment pr UK construction ir	% score of each range	18X5/90=100%	18X4/90=80%	18X3/90=60%	18X2/90=40%	18X1/90=20%	18X0/90=0%	-

Table 4.7.1.3:	Questionnaire	measurement scale

Note: the calculation of percentage scores in Table 4.7.1.3 are based on assumptions for 18 questions in the designed questionnaire. A code of 5 was assigned to 'very strongly agree' down to zero (0) 'very strongly disagree' and 'unsure'. For instance, by

multiplying the total questions in the questionnaire (in this case 18) by the maximum assigned code of 5, giving a total of 90, the percentage score for 'very strongly agree' = $18 \times 5/90 = 100\%$, 'strongly agree' = $18 \times 4/90 = 80\%$, and so on. See the percentage calculation of each 'response scale' in Table 4.7.1.3 above. Note that no number was assigned for the option 'unsure' on the likert scale because it was assumed to be purely an indecisive response within the scale.

4.7.1.4 Participant selection and sample

Easterby-Smith (2002) argued that it is important for researchers to select participants who can provide insights concerning the research problem. Hence, the selected participants for this study were based on a population sample with practical experience and robust perceptions of construction industry payment practices. Dornyei (2007) argued that a group of participants taking part in an empirical investigation is a population sample for a study.

Purposive sampling was used to select participants who were proficient and wellinformed about the study problem. Denscombe (2017) argued that although random or systematic sampling portrays the ideal basis for a representative sample, purposive sampling provides the opportunity for the researcher to extract the best information from persons or instances with valuable experiences of or insights into the research topic.

Purposive sampling, however, was particularly used by the researcher based on prior knowledge of specific people or events and particular participants were deliberately selected to explore their perceptions and explanations of the study problem (Miles et al.2014).Hence, a sizeable population of professionals and other construction practitioners were selected such as quantity surveyors, project managers, architects, consultants, clients, contractor and subcontractors working the UK construction industry.

4.7.1.5 Data collection and procedures

To ensure that the data obtained from the questionnaires truly reflected the circumstances surrounding the research problem, certain procedures were adhered

during collection. These procedures constitute methods used during the distribution of the questionnaire and time constraints inputted to complete it.

To begin with, the first stage of the questionnaire distribution was carried out through site visits, whereby questionnaires were distributed to participants to complete and return. This was crucial to obtain first-hand information about the research problem and build a good rapport to facilitate the subsequent qualitative data collection (interviews).

The second stage was carried out through a web-based questionnaire that was designed and sent to the participants via emails and contact addresses obtained from online searches, telephones, conferences or workshops, networking and referrals. An online tool called Online Surveys (or BOS) was used to design and collect the questionnaires, via a Uniform Resource Locator link (URL) created and sent to ease the collection of responses from the participants and import the collected data into statistical packages for analysis.

Note that it was important for the researcher to collect both on-site, remote and selfadministrated questions to yield better responses and accurate results for the study. As to the time constraints inputted in the study, the questionnaires were planned to be collected from 6 to 12 months on the basis of follow-ups and other efforts made by the researcher. The questionnaire link was created and was sent from 14th September 2018 to the closing date of 23th November 2019. However, for the online questionnaire, participants started to respond from 1st November 2018 because of the time taken in sending individual links to reach out to participants and professionals from the industry. The questionnaire was kept open for a further for 2 weeks until 9/12/2019. The on-site and on-line questionnaires were collected simultaneously.

4.7.2 Qualitative Method for the main study

To gain in-depth understanding of the research problem, this study conducted qualitative interviews. The collection of qualitative data helped the researcher to obtain information about the experiences, perceptions and opinions of construction

practitioners with regard to unfair payment practices in the UK construction industry. This research method was also merged with the study questionnaire to buttress and complement the weaknesses of both.

According to Qu and Dumay (2011), the use of qualitative interviews provides a powerful means to discover new knowledge and capture the accounts of experts in the field in a more open, consistent and systematic manner. To obtain qualitative data, semi-structured interviews were conducted with seasoned construction professionals. The researcher prepared a list of questions that intended to address the study's aim and objectives (see appendix C). Saunders et al. (2009) claimed that semi-structured interviews give the researcher freedom to probe further about the research problem; with the flexibility to cover or change certain themes and questions that will emerge from the interviewee's answers.

4.7.2.1 Interview design and measure

In order to conduct an effective structured interview, Klenke (2016) argued that having a guide or protocol is essential to the entire process. The interview protocol or interview guide is list of questions or topics that the participants will be asked about. The interview guide is helpful to cover the key research questions and probe aspects of what the participants say.

A semi-structured interview was used, comprising a set of prepared open-ended questions that enabled the researcher to gather answers and to probe for any issues raised that were related to the subject matter. Consequently, the first phase of the face-to-face qualitative data method commenced with a few personal or factual comments about the researcher, the purpose of the study, individual consent, the duration of the study and any foreseeable risks or discomforts to the subject. This was done to build rapport between interviewer and interviewees.

A set of prepared questions was then used to cover different areas of the subject matter, while open ended questions encouraged the interviewees to make further considerations while enabling the interviewer to probe aspects of what they were saying. Probing questions were constructed from any follow-up questions, gestures or verbal excerpts from the participants during the course of the interviews to generate further interpretations from them. During the interviews, the researcher opted to be a keen or good listener rather than speaking more to extract further information or insights given by the interviewees.

4.7.2.2 Participant selection and the sample

The study participants considered for the interviews were construction clients, project managers, quantity surveyors, contract managers, commercial managers, contractors, subcontractors, representatives of construction trade bodies and construction specialist providers. The selection of participants was mainly drawn from questionnaires administered during site visits, a contact directory via https://www.linkedin.com/home, construction professional bodies, building magazines and networking from continuous professional development programmes (CPD).

As previously discussed in the quantitative method, purposive sampling was also employed by the researcher to select interviewee who were proficient and wellinformed about the study problem. The purposive sampling was comprised of seasoned industry practitioners and contractors that possessed construction experience and were recognised as members of high professional bodies. These groups were carefully considered to extract their practical and industrial experiences of unfair payment practices in the UK construction industry.

4.7.2.3 Procedure and data collection

After obtaining the participants' consent to take part in the proposed interviews from site visits and online sources, arrangements were made as to the scheduled date, time and location of the interviews. Prior to the interviews, the participants were sent a Research Participant Information Sheet (PIS) that outlined the details of the research, the consent form and the ethnical checks carried out by both the researcher and the research directors (see appendix D). Moreover, due to geographical and access constraints to reaching some of the participants, the researcher engaged the use of some digital technologies such as Skype, Microsoft Teams, Zoom and others to conduct interviews depending on the availability and preferences of the interviewees.

4.7.2.4 Equipment

The equipment used in conducting the interviews were a digital tape recorder, phone recorder and the physical jotting of key points. The digital recorder was used by the researcher in recording and preserving the participants' opinions given in the interviews, as note taking proved to be cumbersome. Moreover, before the commencement of an interview session, the participants were notified in advance by the researcher regarding the purpose of the study, the duration of the interview and their right to participate or withdraw from the process. Hence, a participant's information sheet and consent form were sent out prior to the interview.

4.7.3 Archival data method for the study

Apart from quantitative and qualitative methods used in the study, archival data from Companies House was extracted and analysed to measure the effects of unfair payments on contractors' business performance. Financial statements of some UK companies involved in the study were sought to investigate the companies' performance and payment practices (Table 6.3, 6.3 and 6.4).

A total of 40 UK construction companies' financial data were obtained from networking and from interviews conducted by the researcher: 14 were mechanical and electrical companies, 9 were concrete companies, 4 were roofing companies, 4 were interior designers and installers, 1 was a floor screeder; 3 were lift companies, 2 were demolition companies and 3 were specialist ground works contractors. Afterward, the researcher went to obtain their financial performance data constituting profit and loss accounts from 2014 to 2018. Other financial data were retrieved from Companies House via the website:

https://www.gov.uk/government/organisations/companies- house

The archival data are presented in section 6.4.1 and 6.4.6 of the analysis of business performance. The list of construction companies together with their operating profit margin, turnover, return of capital employed, current assets and current liabilities is attached in appendix E. Moreover, 15 case studies were added as part of the archival data to investigate the prevalence and effects of unfair payment practices on construction companies.

Moreover, documented payment data from a government website was obtained and analysed in order to have broader view of payment practices and the performance of different tiers of construction clients. The data was sought from the Build UK Group Limited and is available at the following URL link <u>https://builduk.org/information/</u> and presented in Tables 6.3.1, 6.3.2 and 6.3.3.

4.8 Data Analysis

4.8.1 Quantitative data analysis

For a robust quantitative analysis, all data obtained were edited and any questionnaires with invalid or inaccurate responses were eliminated. The data were then coded by assigning numbers to the questionnaire's options (i.e. 'very strongly agree'=5, 'strongly agree'=4, 'agree'=3, 'disagree'=2 and 'very strongly disagree'=1) as illustrated in Table 3.4. Full details of the analysis on an Excel spread sheet is attached in Appendix F. Using the Excel spread sheet, a cross tabulation of data was manually carried out. Data were inputted with each question succeeding another on the basis of how they were returned or received.

Each questionnaire was denoted as Participant 1, 2, 3, etc. (up to the total number received) on the vertical axis and question variables were arranged on the horizontal axis to match with each response to the question. Using the Statistical Package for the Social Sciences (SPSS.25), different statistical tests were then conducted to analyse the collected data and test the research questions. Nachmias and Nachmias (2008) argued that it is vital for researchers to carry out statistical analyses from data obtained from questionnaires and surveys.

Descriptive statistics such as mean, standard deviation, rank scores and relative important index (RII) were carried out to provide a description of the robustness of standard forms of contracts, the magnitude of unfair payment practice and the causes and effects of unfair payment practices. Moreover, internal reliability and data consistency tests such as Cronbach's Alpha test, Mann Whitney U-test and the skewness test were conducted. Inferential analysis techniques such as Pearson' coefficient correlation test were also used to investigate the relationships between the study variables.

4.8.2 Qualitative data analysis

Data obtained through the interviews were analysed using content analysis. This method of analysis helps to examine written texts, verbal or visual communication messages by making replicable and valid inferences from the data to a few content categories (Elo and Kyngas 2008). Often it is argued that classifying words into the same categories, phrases and such like, share the same meanings or interpretations (Cavanagh 1997).

Therefore, the data sourced from the interviews were recorded using a digital or phone recorder and transcribed into written forms and texts by listening to the taped conversations or discussions held between the researcher and the study participants. The transcribing into manuscript was carried out where different textual excerpts, concepts or themes had emerged from the interview data. The textual excerpts were then inputted into Nvivo 12 software, where codes were assigned to key themes to facilitate the filtering and sorting of data. Codes were used to attach meanings to disparate information and responses to the data. The themes from the study objectives and aim were used to create codes and sub-codes from the transcribed data. The coded data are analysed in Chapter 7 as 'textual excerpts or contents' from the participants' responses. Accordingly, excerpts from the interviews were obtained using content analysis; by counting the number of processes and extracting systematic and objective meanings from each content via valid inferences from both verbal and archival data.

4.9 Reliability and Validity

To uphold the credibility of research data, validity and reliability are used to measure the quality and trustworthiness of the qualitative scientific and experimental studies, often based on standardised methods according to predetermined instruments.

Winter (2000) argued that there seem to be endless theoretical arguments concerning the nature or definition of the terms 'validity' or 'reliability' in academic and social research. However, Kuzmanić (2009) stressed that there is a "pure form of truth" somewhere out there, that can be discovered by using appropriate and most importantly, valid research methods. For straightforwardness, this study infers reliability to the quantitative method and validity to the qualitative, in order to construct different interpretations (or social worlds) for the readers.

Reliability in the questionnaires was addressed by the detailed design of the questionnaire aligned to predetermined objectives using an appropriate scale of measurement. Also, other standards designed to uphold reliability included data collection procedures, a pilot study and analysis techniques. As for the semi-structured interviews carried out by the researcher, validity was upheld through the selection of experienced participants, the design of the interview questions, the instrument of collection, the transcription and processing of the data and the presentation and interpretation of the interview data. Moreover, the use of both multi-methods in investigating the research question enhanced the reliability and validity of the research findings (Saunders et al. 2009 and Denscombe 2017).

4.10 Ethical Considerations

Adherence to established and recognised ethical principles is a primary concern for any researcher. Saunders et al. (2009) defined ethics as the researcher's behaviour towards the research participants or people affected by the research. Ethics can be said to be moral principles and social norms that dictate behaviour and activity. To ensure that this study was carried out to high ethical standards, the researcher sought approval from the established research ethics committee at Coventry University. A copy of the ethics certificate and clearance is attached in Appendix G. The following subsection will briefly explain the key ethical issues that were addressed in this study.

4.10.1 Consent of the Participants

Consent of the participants is a crucial aspect of ethical considerations in any research. The study participants' consent for this study was sought using a consent form, as seen in appendix H. The consent to participate in the interviews was sought before any site visits or interviews carried out by the researcher. Prior to all site

visits, full information regarding the purpose of the interview, its duration, location, time and date were thoroughly made clear before a site visit took place

For the online survey, a participant's information sheet (PIS) was attached at the beginning of the questionnaire to give information concerning the nature of the project, the purpose of the study, any risks; and also instruction and guidance on accessing the questionnaire. In addition, the participants were made aware that their participation was voluntarily, and they had the right to withdraw at any stage of the research

4.10.2 Privacy and Confidentiality

Bell and Bryman (2007) stressed that in any form of qualitative research, issues related to confidentiality and anonymity must be taken into consideration. For instance, prior to this study's data collection, the researcher had to assure all participants that the data collected would be used for academic purposes and would not be disclosed to any third party. Thus, for the online questionnaire survey, the BOS survey tool was used by the researcher to keep the collected data private and confidential as well as protecting the participant's anonymity.

The BOS offered an URL link access to the questionnaire that was emailed by the researcher directly to the participants and their representative organisations. Once the survey was completed, all data from the responses was downloaded to the researcher's personal computer for the purposes of analyses. All interview data were tape recorded, transcribed and kept strictly confidential by the researcher, with a password used to prevent unlawful access.

4.11 Summary

This chapter outlined the research methodology and methods used for this study. The chapter took account of the 'Research Onion' concept to explain the current stance of the research in terms of its philosophical assumptions, research approaches, strategies and methods used in the data collection. The chapter also presented measures taken to ensure the validity and reliability of the study.

CHAPTER 5

Quantitative Data Analysis

5.1 Introduction

This chapter presents the quantitative analyses for the questionnaire data. The results are presented under five subsections aligned with the research objectives discussed in Chapter 1.3. Section 1 provides a brief description of both the questionnaire surveys and the methods used; while Section 2 presents the sample representativeness and demographic information of the questionnaire survey. Sections 3 and 4 provide results of Mann-Whitney U-test, Skewness test and Cronbach's Alpha for the questionnaire data collected from both hard copy (as in face to face) and online sources. These tests were essential for the questionnaire survey to ascertain the measures for the internal consistency and reliability of the collected data.

The analyses of unfair payment practices in the UK construction industry are then discussed in sub-themes such as the forms and magnitude of unfair payments, clients' perceptions of unfair payment practices, the causes of unfair payment practices, the robustness of payment procedures and their provision in standard forms of contracts. Also, the role of the client-contractor relationship in unfair payment practices and the effect on contractors' business performance were subsequently analysed. Furthermore, financial ratios were employed to measure and assess the business performance of construction companies prone to unfair payment practices, namely gross profit, operating profit margin, return of capital employed (ROCE), current ratio and quick ratio. The analyses and discussions in this section are presented in various tables and graphs whilst conclusions are presented in Chapter 8.

5.2 Questionnaire Description

Two sets of academic questionnaires were used for this study. The questionnaires were administered to different study participants at different time intervals. The design of questionnaire 1 specifically measured generic issues about unfair payment practices such as their causes, the scale of the problem and its impact on business performance. Questionnaire 2 was designed to measure the disparity in commercial

bargaining power, contractual relationships and construction clients' perceptions of unfair payment practices in the industry. The study also used two sets of questionnaires to enhance its reliability and credibility by collecting data from two distinct spectrums of the population sample.

Both questionnaire surveys were administered to seasoned stakeholders and practitioners working in the UK construction industry. Table 5.2 presents the total number of questionnaires received with regard to their validity and the methods used. A total of 201 questionnaires were collected, but some of the questionnaires were filled incorrectly. Therefore 179 were considered for analysis.

Methods employed for	Total number of	Total number of valid	Percentage o f invalid
data collection	questionnaires	questionnaires received	questionnaire response
	received		(%)
Online	132	115	17
Face to face	69	64	4
Total	201	179	32

Table 5.2 Total number of valid questionnaires received from each method

5.3 The Wilcoxon Signed Rank Test for Data Homogeneity

Table 5.3.1 illustrates the total perceptions of the study participants for questionnaire 1 of the collected data. Groups 1 and 2 below denote questionnaire data collected via online platforms and face to face, respectively.

Using SPSS version 25, a test for homogeneity was conducted to ascertain if there were significant differences between data collected via online methods (URL/emails) (coded as group1) and the data obtained face-to-face (coded as group 2), as illustrated in table 5.3.1

No of respondent	Total percentage response from online data	Total percentage response from online data	Total percentage response from online data	Total percentage response from online data	Total percentage response from online data	Total percentage response from face to face data	Total percentage response from face to face data	
	Group 1	Group 1	Group 1	Group 1	Group 1	Group 2	Group 2	
1	73	70	53	60	64	58	59	
2	66	64	62	61	43	60	39	
3	66	64	63	42	53	66	48	
4	59	54	62	51	47	63	57	
5	51	59	60	47	51	66	56	
6	63	71	43	50	59	60	60	
7	48	61	51	59	57	51	59	
8	49	67	64	57	50	66	68	
9	56	63	64	51	52	66	67	
10	72	63	57	66	50	66	51	
11	57	53	57	43	64	39	28	
12	47	59	50	68	65	56	59	
13	59	54	52	52	64	70	60	
14	62	53	43	59	63	66	67	
15	64	66	68	57	60	61	50	
16	40	66	71	51		66	64	
17	56	63	62	63		61	59	
18	49	51	69	43		73	57	
19	50	48	64	68		62	57	
20	54	60	64	61		64	48	
21	57	63	47	50		52	36	
22	50	71	52	51		60	54	
23	63	58	57	68		51	67	
24	43	69	73	67		58	65	
25	68	48	62	53		70		
26	60	56	69	46		66		
27	68 69	46	58	51		69 60		
28 20	80 60	69 53	66 66	59 56		69 65		
29	58	58	58	19		67		
30		an of norcer			-57%	07		
		an of percen	tage respon	565 (11=135) ses from on	-31 % lina LIPL /r=	64) -50%		
	Overall mean of percentage responses from online URL (n=64) =59% Note: 179 study participants overall were grouped into three categories in this particular table							

Table 5.3.1: Total perceptions of study participants to questionnaire surevy-1

Test Statistics						
	Face to face - Online					
Z	-1.967 ^b					
Asymp. Sig. (2-tailed)	0.081					
a. Wilcoxon Signed Ranks	Test					
b. Based on positive ranks.						

Table 5.3.1.1 Wilcoxon signed rank test on questionnaire data collected- survey 1

Table 5.3.1.1 illustrates findings for the Wilcoxon Signed Rank test conducted between groups 1 and 2. The statistical test revealed asymptotic significance (2-tailed) for group 1 vs. group 2 as (Z= -1.967, p=0.081). This implies that there was no significant difference between the data collected from online and face-to-face methods, as the p value exceeded 0.05. The homogeneity test confirmed that the group of data collected from these different sources were homogeneous; thus, both data could be used for further quantitative analysis.

No of respondent	Total percentage response from online data	Total percentage response from online data	Total percentage response from online data	Total percentage response from online data	Total percentage response from face to face data	Total percentage response from face to face data			
	Group 1	Group 1	Group 1	Group 1	Group 2	Group 2			
1	87	80	60	64	87	86			
2	73	63	61	43	73	73			
3	63	63	42	53	63	63			
4	63	96	51	47	63	80			
5	73	57	47	51	80	68			
6	68	73	50	59	68	73			
7	53	68	59	57	67	68			
8	67	53	57	50	67	64			
9	104	63	51	52	68	68			
10	73	57	66	50	67	64			
11	63	73	43	64	68	64			
12	63	68	68	65	73	56			
13	96	53	52	64	68	73			
14	57	57	59	62	64	68			
15	63	68	57	-	67	53			
16	57	67	51		57	56			
17	73	57	63		67	60			
18	68	87	43		57	70			
19	53	73	68		73	73			
20	57	69	61		68	68			
	20 57 69 61 68 68 Overall mean of percentage responses (n=74) =66% Overall mean of percentage responses (n=74) =66%								

Table 5.3.2: Total perception of study participants to questionnaire survey 2

Test Statistics				
	Face to face - Online			
Z	246 ^b			
Asymp. Sig. (2-tailed)	. 805			
a. Wilcoxon Signed Ranks Test				
b. Based on positive rai	nks.			

Table 5.3.2.1 Wilcoxon signed rank test on questionnaire data collected- survey 2

Similarly, table 5.3.2.2 illustrates the Wilcoxon Signed Rank test conducted between groups 1 and 2. Note that the data used for this analysis was from questionnaire 2 collected separately from questionnaire 1. The statistical test reveals asymptotic significance (2-tailed) for group 1 vs. group 2 as (Z= -.246, p=.805). The finding implies similarity to that stated in the Wilcoxon Signed Rank test reported in table 5.3.2.1.

5.4 Test for reliability (Cronbach's Alpha)

Table 5.4.1 (a and b) present alpha coefficients (α) values of 0.702 and 0.602 respectively. The findings in table 5.4.1 below show that the internal reliability of the two data sets is acceptable, while the internal reliability of Table 5.4.1b may be questionable based on the interpretation of the Cronbach's alpha test in table 5.4.2 put forward by Shavelson (2004).

Cronbach's Alpha	Cronbach's Alpha Based on Standardised Items	N of Items
0.726	0.759	24

Table 5.4.1a Cronbach's alpha test on data collected in questionnaire 1

Table 5.4.1b Cronbach's alpha test on data collected in questionnaire 2

Cronbach's Alpha	Cronbach's Alpha Based on Standardised Items	N of Items
0.602	0.6114	18

Cronbach's Alpha test was essential to determine the internal consistency of the questionnaires' data sets.

Cronbach's alpha value	Internal consistency
0.9 < a	Excellent
0.8 < a < 0.9	Good
0.7 < a < 0.8	Acceptable
0.6 < a < 0.7	Questionable
0.5 < a < 0.6	Poor
$\alpha < 0.5$	Unacceptable

Table 5.4.2: Interpretation of Cronbach's alpha reliability test values adapted from Shavelson (2004).

5.5 Descriptive statistics

Table 5.5a Descriptive Statistics for Questionnaire 1

	N	Range	Minimum	Maximum	Mean	Std. Deviation	Variance	Skewness	
	Statistic	Statistic	Statistic	Statistic	Statistic	Statistic	Statistic	Statistic	Std. Error
VAR00001	179	5	0	5	3.60	1.032	1.07	-0.259	0.182
VAR00002	179	5	0	5	2.78	1.015	1.03	0.78	0.182
VAR00003	179	5	0	5	3.28	1.121	1.26	-0.19	0.186
VAR00004	179	5	0	5	3.54	1.175	1.38	-0.41	0.183
VAR00005	179	3	1	4	2.42	0.923	0.85	0.01	0.182
VAR00006	179	4	1	5	2.18	0.858	0.74	0.55	0.182
VAR00007	179	5	0	5	2.20	0.939	0.88	-0.28	0.182
VAR00008	179	4	1	5	3.16	1.026	1.05	0.21	0.182
VAR00009	179	4	1	5	4.11	0.818	0.67	-0.45	0.182
VAR00010	179	4	1	5	3.92	0.804	0.65	-0.24	0.183
VAR00011	179	4	1	5	2.72	0.787	0.62	0.26	0.184
VAR00012	179	4	1	5	3.55	1.081	1.17	-0.49	0.183
VAR00013	179	5	0	5	2.77	0.931	0.87	-0.42	0.207
VAR00014	179	4	1	5	3.72	0.811	0.66	-0.08	0.183
VAR00015	179	4	1	5	3.39	0.856	0.73	0.32	0.184
VAR00016	179	3	2	5	3.75	0.811	0.66	-0.04	0.184
VAR00017	179	3	0	3	1.47	0.785	0.62	-0.53	0.183
VAR00018	179	4	0	4	1.81	1.195	1.43	-0.28	0.214
Valid N (listwise)	98		•	•	:		:	•	•

Descriptive Statistics										
	N	Range	Minimum	Maximum	Mean		Std. Deviation	Variance	Skew	ness
	Statistic	Statistic	Statistic	Statistic	Statistic	Std. Error	Statistic	Statistic	Statistic	Std. Error
VAR00001	94	3	2	5	3.617	0.0727	0.70492	0.497	0.137	0.249
VAR00002	94	2	1	3	1.68	0.0645	0.62584	0.392	0.355	0.249
VAR00003	94	2	3	5	3.595	0.0647	0.62766	0.394	0.562	0.249
VAR00004	80	4	1	5	3.5	0.1052	0.94132	0.886	-1.074	0.269
VAR00005	94	2	3	5	3.51	0.06	0.58189	0.339	0.627	0.249
VAR00006	94	2	3	5	3.659	0.0633	0.6144	0.377	0.358	0.249
Valid N (listwise)	80		 	 			 			r

Table 5.5b: Descriptive Statistics for Questionnaire 2

Table 5.5 (a and b) presents the descriptive statistics of the questionnaire data. The descriptive statistics describes the data behaviour regarding the study participants' perceptions of the questions asked. The data range shows a minimum value of 0 and a maximum value of 5; meaning that the data dispersion level is normal. The data is not scattered, nor does it have minimal variability. The mean value for the data set shows the average responses to each question. This denotes that most study participants agreed on the questions asked; except for questions 17 and 18 with mean values of 1.47 and 1.81 in table 5.5a and question 2 in table 5.5b.

A critical review of these questions with a low mean will be discussed subsequently. The standard deviation (SD) of the data set in table 5.5b reveals a small value compared to the data set in table 5.5a. The SD of 5.5b reveals closeness of responses provided by the study participants to the mean score, meaning that the SD in table 5.5b is even or is close to the mean value. However, the SD data in table 5.5 seems to show the contrary. Moreover, the variance values for both tables show a low value ranging from 0.34 to 1.43; meaning the rate at which the study participants changed their perceptions was not very wide. For a thorough understanding of the data set's behaviour, a skewness statistics test was conducted to ascertain the even and uneven distribution normality of the questionnaire data.

5.6 The Skewness Test

Figure 5.6 (a, b and c) illustrates the skewness diagrams and values for variables 1, 2, 3, 4, 5, 7, 8 and 9, that represent questions 1, 2, 3, 4, 5, 7, 8 and 9 in questionnaire survey 1; and the variables 2, 3, 5 and 6, that represent questions 2, 3, 5 and 6 in questionnaire survey 2. The graph for figure 5.6a shows a distribute skewed to the right-hand side, denoting that responses to the questions received a positive response, while graph 5.6 b shows a normally distributed response of an average of 3. However, figure 5.6 c presents the skewness diagram and values for variables 6,10 and 11, that represent questions 6,10 and 11 in questionnaire survey 1. The graph shows a distribute skewed to the left-hand side, denoting that responses to the questions for the questions received a negative response at an average of less than 2.



Figure 5.6a Pictorial view of skewness for Questionnaire 1



Figure 5.6b Pictorial view of skewness for Questionnaire 2



Figure 5.6c Pictorial view of skewness for Questionnaire 1

5.7 Profile of the study participants

Majority of participants that took part in the study indicated their professional backgrounds as follows: 31% contractors, 27% subcontractors, 21% quantity surveyors; 11% clients, 6% project managers and 4% architects. Table 7.6 provides a summary of the study participant sample.

The distribution of the study participants' years of work experience in the construction industry shows the follow demographics: 33% of study participants had 11-15 years of work experience; 27%-6-10; 18%-16-20; 10%-21-25 and 7% less than 5 years, respectively. Furthermore, the data also reveals the participants' affiliation to professional bodies. The findings show that 33% were affiliated to the Chartered Institute of Building (CIOB), 7% to the Royal Institution of Chartered Surveyors (RICS), 1% to the Royal Institute of British Architects (RIBA) and 59% to other professional bodies. In terms of the type of contract frequently used in construction, the data shows that 30% used JCT forms, 12% NEC, 9% other standard forms, 5% FIDIC, and 44% used bespoke forms of contracts.

Overall, the data reveals that the study participants' information, work experience and their professional affiliation demonstrated their understanding of the research problem. Hence, credibility of response was obtained.

	Table 5.7	Profile	of the	study	partici	pants
--	-----------	---------	--------	-------	---------	-------

General information		Total	Percentage				
General information Quantity Surveyor Project Manager Architect Contractor Subcontractors Clients 1-5 years 6-10 years		Total 37 11 7 56 48 20 12 59 48	Percentage				
experience	16-20 years 21-25 years More than 25 years	32 18 10	6- 10 years 33 1-5 years 7 0 20 40				
Nature of business	Residential building Commercial building Property Infrastructure Heavy construction	72 64 18 10 15	Heavy Construction Infrastructure Property Development Commercial Building Residential Building 0 20 40 60				
Frequently used standard forms of contract (SFoCs)	JCT FIDIC NEC Bespoke SFoCs Other SFoCs	53 6 24 79 14	Bespoke forms Other SFocS NEC FIDIC JCT 0 10 20 30 40 50				
Professional affiliation bodies	ACIOB MCIOB MRCIS Others MRIBA	3 56 13 105 2	Other MRIBA MRICS MCIOB ACIOB 1 7 59 0 20 31 0 20 40 60 80				

5.8 Quantitative Analysis of the study's objectives

5.8.1 Objective 1- To evaluate the robustness of payment procedures in various standard forms of contract used in the construction industry

For emphasis, objective 1 is 'To evaluate the robustness of payment procedures in various standard forms of contract used in the UK construction industry'. Questions 5, 6, 7 and 8 (from survey 1) sought the study participants' opinions regarding objective 1. Table 5.8.1 presents mean scores of 2.46, 2.18, 2.20 and 4.11 for questions 5, 6, 7 and 8, respectively. These values denote that majority of participants disagreed that payment procedures clauses in SFoCs help to alleviate unfair payment practices. Note that the mean score of 4.11 in question 8 indicates that the study participants strongly agreed with the question asked.

It can be argued that participants' response of "Strongly Agree" to question 8 is a true reflection of industry payment practices; because in reality, contractors are often reluctant to use legal means to recover money owed to them. In many cases, SFoCs are normally amended to suit the client's interests; perhaps creating a tendency to limit the fairness and promptness of paying monies owed to contractors. Findings in table 5.8.1 will be discussed further in Chapter 8 (the discussion chapter); where the quantitative, qualitative and archival data will be critically reviewed.

Question No. in the Questionnaire	Percentage agreement to which participants agree or disagree with the following statements:	Mean Score	S.D	Mean score in (%)		
5	Payment provisions in standard forms of contracts (SFoCs) helps to minimise unfair payment occurrence	2.46	0.92	49.20		
6	SFoCs (JCT/NEC/FIDIC) have sufficient mechanisms to protect contractors against unfair payment practices	2.18	0.86	43.60		
7	28 days payment period stipulated by various SFoCS helps to minimise unfair payments	2.20	0.94	44.00		
8	Contractors are reluctant to use penalties stipulated in regulations, SFoC and charters concerning unfair payment because of fear that it will damage their business relationship	4.11	0.82	82.20		
Total percentage agreement to SFoC's unfair payment practices in the construction industry2.740.8854.75						
Note: Mean 5.00 = Very Strongly agree, 4= Strongly agree, 3= Agree, 2= Disagree, 1= Strongly disagree, 0 = Very Strongly disagree and - was assigned unsure						

Table 5.8.1 Percentage agreement to payment procedures and provisions in standard forms of contracts

Figure 5.8.1 presents a summary chart of the study participants' views in response to study objective 1. The chart largely shows that the majority of participants strongly

disagreed that SFoCs payment mechanisms protected contractors against unfair payment practices; as "very strongly disagree" had a response score of 43%, "strongly disagree" 39% and "disagree" 13%, respectively.



Figure 5.8.1: Perceptions of study participants on SFoCs mechanisms in protecting contractors against unfair payment practices

5.8.2 Objective 2- To determine the magnitude (degree) of unfair payment practices among tiers of construction clients

Questions 11, 12 and 13 (from survey 1) were used to measure the magnitude of unfair payment practices between different tiers of construction clients in the UK. Similarly, questions were asked in the second set of questionnaires. In questionnaire 2, participants were asked to indicate the nature of unfair payment practices within construction in terms of types and commonalities. Note that as explained in Chapter 1 section 1.3, the study has classified tiers of construction clients as Tier 1 clients (major construction clients); Tier 2 clients (main contractors acting as clients to subcontractors) and Tier 3 clients (subcontractors acting as clients to other subcontractors).

Figure 5.8.2.1 illustrates that 45% of study participants agreed that unfair payment practices are frequent among Tier 2 clients. 37% of unfair payment practices are attributed to tier 1 clients and 18% to tier 3 clients.

This indicates that unfair payment practices cut across all tiers of clients in the construction industry regardless of their size, nature and contracting experience, etc. On the other hand, 31% of study participants disagreed that tier 3 clients would withhold or induce unfair payment practices to their contractors and suppliers in the construction industry.



Figure 5.8.2.1 Magnitude of unfair payment practices among tiers of construction clients

Furthermore, questions 6, 7, and 8 in survey 2 asked study participants to indicate types and commonalities of unfair payment practices synonymous with different clients. Findings from figure 5.8.2.2 indicate that over 35% of study participants scored late payments as the most common unfair payment practice for all tiers of construction clients.

Statistically, a breakdown of the figures shows that 38% of late payments cases are attributed to Tier 1 clients; 20% unpaid retentions; 15% pay if notified; 11% imposition of discounted rates for early payments; 9% imposition of rates (bill of quantities); 6% disparities in rates for items and 1% exclusion of provisional remedy.

The findings also reveal that Tier 2 clients are responsible for 37% late payments, 30% unpaid retentions, 20% pay when paid; 5% imposition of discounted rates for early payments, 4% exclusion of provisional remedy, 2% disparities in rates for items and 1% imposition of rates (bill of quantities). Tier 3 clients meanwhile were attributed to

35% of late payments, 29% pay when paid, 26% unpaid retentions, 6% disparities in rates for items, 2% exclusion of provisional remedy, 1% imposition of discounted rates for early payments and 1% imposition of rates (bill of quantities). The prevalence of these unfair payment practices is illustrated in figure 5.8.2.2 below.



Figure 5.8.2.2 Percentage commonality of unfair payment practices among construction clients

In addition, when participants were asked to indicate the duration of receiving payments from clients; their responses are presented thus: 43% indicated payment terms of 31 to 50 days; 30% between 51 to 90 days; 14% claimed to receive payment between 91 to 120 days and 9% had longer payments terms of 120 days and beyond.

Meanwhile, 4% of study participants claimed to receive regular payments based on agreed contractual terms within 28 days as illustrated in figure 5.8.2.3. The existing literature shows that longer payment terms beyond contractually agreed terms are generally considered as a construction industry norm; with an average time of 46 days for tier 2 clients' (main contractors) to pay invoices to subcontractors and suppliers.



Figure 5.8.2.3 Average duration in days to receive payment from clients

5.8.3 Objective 3 -To investigate the contemporary causes of unfair payment practices in the UK construction industry.

Table 5.8.3.1 presents the rankings for causes of unfair payment practices. The causes were derived from both study's literatures and initial interviews conducted by the researcher. A total of 23 companies' business strategies and causes were identified and subsequently categorised under five main groups, namely (i) cultural; (ii) industrial, (iii) technical, (iv) regulatory and (v) others. The study participants were asked to select from an array of 23 factors presented in the questionnaire using a likert scale ranging from 5= 'very strongly' agree to 0= 'very strongly disagree'.

The mean causes were then calculated based on the participants' responses and groupings, as illustrated in table 5.8.3.1. Note that table 5.8.3.1 presents the study findings grouped into three broad categories, namely contractors (comprising the main contractor, subcontractors and sub-subcontractors), consultants (comprising quantity surveyors, architects and project managers) and clients (private or public).

Group	No.	Causas	Contractor		Consultants		Client	
		Causes	Mean	Rank	Mean	Rank	Mean	Rank
	1	Client-contractor relationship	3.64	1	3.48	2	3.49	3
	2	Pay when paid	3.60	2	3.39	3	3.53	2
	3	'Would not pay' attitude of clients	3.01	5	3.09	5	3.39	5
Cultural	4	'Cannot pay' attitude of clients	3.47	4	3.15	4	3.40	4
causes	5	Prevailing culture	3.50	3	3.57	1	3.59	1
	6	Cowboy bullying	3.21	2	3.38	2	3.29	3
	7	Corruption	2.54	5	2.99	5	3.00	5
	8	Supply chain finance	3.04	4	3.32	3	3.37	2
	9	Low barriers to entry and exit	3.10	3	3.29	4	3.18	4
Industrial causes	10	Industry norm 'work first get paid later'	3.35	1	3.41	1	3.40	1
	11	Errors in submitting claims and information carrying out valuation evaluation	2.64	3	3.08	2	3.17	2
	12	Defective works	2.72	4	3.14	1	3.20	1
Technical	13	Delay in approval of works	3.07	2	2.86	4	2.88	4
causes	14	Insufficient documentation	3.15	1	2.89	3	3.10	3
	15	Ambiguities within standard forms of contracts	2.69	3	3.17	3	3.17	2
	16	Multi-tiered hierarchical structure	3.43	2	3.30	1	3.26	1
	17	Complex payment legislation	2.50	4	3.09	4	3.00	4
Regulatory causes	18	Widespread use of adhesion contracts ' take it or leave it'	3.65	1	3.26	2	3.14	3
	19	Lack of transparency	2.46	5	2.89	5	2.94	5
	20	Exploiting fierce competition among subcontractors	3.02	3	3.38	2	3.34	2
	21	Undue commercial interest	3.04	2	3.31	3	3.29	3
Other	22	Administration inefficiencies	2.76	4	3.29	4	3.22	4
causes	23	Cashflow strategy	3.76	1	3.49	1	3.52	1

Table 5.8.3.1 Group rankings and mean of causes of unfair payment practices

Findings from table 5.8.3.1 relating to contemporary causes of unfair payment practices common to contractors illustrate that "cash flow strategy" had the highest average of 3.76; meaning that some construction clients use unfair payment practices as a strategy to enhance their cash flow. This was followed by "widespread use of adhesion contracts" with a mean score mean of 3.64, followed by "pay when paid" with a mean of 3.60 and "prevailing culture" with a mean score of 3.50. These findings reveal that main contractors acting as clients to subcontractors use late payments as a business strategy to boost their cash flow and at the same time, this is a causal factor.

Similarly, causes of unfair payment practices synonymous with construction project consultants revealed that "prevailing culture" in the industry had a mean score of 3.57; followed by use of "cash flow as a strategy" with a mean score of 3.49, "client-contractor relationship" with a mean score of 3.48 and "pay when paid" with a mean score of 3.39. Meanwhile, causes of unfair payment practices linked to client groups revealed that "prevailing culture in the industry" had a mean score of 3.59; followed by "pay when paid" with a mean score of 3.53, cashflow strategy with mean of 3.52 and "client-contractor relationship" with average score of 3.49.

Overall, contemporary causes of unfair payment practices based on clients, consultants and contractors' analyses show that the use of "cash flow strategy" to boost a client's finances is the most prevalent, followed by "prevailing culture in the industry", consideration for the "client-contractor relationship" and the "pay when paid" culture, as summarised in table 5.8.3.2. Tables 5.8.3.1 and 5.8.3.2 infer that contemporary causes of unfair payment practices in the construction industry are wide-ranging and require extensive understanding to curb.

Causes	RII	Rank
Cashflow strategy	3.59	1
Prevailing culture	3.55	2
Client-contractor relationship	3.54	3
Pay when paid	3.51	4
Industry adage 'work first get paid later'	3.39	5
Widespread use of adhesion contracts 'take it or leave it'	3.35	6
'Cannot pay' attitude of client	3.34	7
Multi-tiered hierarchical structure	3.33	8
Cowboy bullying	3.29	9
Exploiting fierce competition among subcontractors	3.25	10
Supply chain finance	3.24	11
Undue commercial interest	3.21	12
Low barriers to entry and exit	3.19	13
'Would not pay' attitude	3.16	14
Administration inefficiencies	3.09	15
Errors in submitting claims and valuation work	3.05	16
Defective works	3.02	17
Ambiguities within standard forms of contracts	3.01	18
Insufficient documentation and information in carrying out valuation		
evaluation	2.96	19
Delay in approval of works	2.94	20
Complex payment legislation	2.86	21
Corruption	2.84	22
Lack of transparency	2.76	23

5.8.3.2 Relative Importance Index of contemporary causes of Unfair payment practices

Therefore, the null hypothesis (H1) should be rejected and the alternative hypothesis (H1) accepted; because the RII analyses in tables 5.8.3.1 and 5.8.3.2 show that unfair

payment practices are deliberate acts used by construction companies as a business strategy to enhance their cash flow.

5.8.4 Objective 4- To ascertain whether unfair payment practices influence contractors' business performance.

Objective 4 was set to ascertain whether unfair payment practices influence a contractor's business performance. Questions 14, 15 and 16 sought study participants' perceptions regarding the impact of unfair payments on contractors' business performance.

Table 5.8.4 presents the summary of mean scores, standard deviation and percentage mean scores for questions 14, 15 and 16 in survey 1. Findings in table 5.7.4 show that majority of study participants agreed that unfair payment practices influenced contractors' business performance, as illustrated in figure 5.8.4 below.

Table 5.8.4 Percentage agreement as to whether unfair payment practices influence contractors'business performance

QN	Percentage agreement to which participants' agree or disagree with the following statements:	Mean Score	S.D	Mean score in (%)
14	Unfair payment influences overall contractors' financial performance.	3.72	0.81	74.35
15	Unfair payments significantly influence contractors' profitability	3.39	0.86	67.77
16	Unfair payments are a threat to contractors' business survival.	3.75	0.81	67.77
	Total percentage agreement to SFoC's unfair payment practices in construction industry	3.62	0.83	69.96
	Note: Mean 5.00 = Very Strongly agree,4= Strongly agree,3= Agree, 2= Disagree,1= Strongly disagree, 0 =Very Strongly disagree and – was assigned unsure			
			1	



Figure 5.8.4 presents a pictorial chart of the study participants' views in response to objective 4. The chart illustrates that the majority of participants - 95% - agreed that unfair payment practices influenced contractor business performance, whereas 5% of study participants disagreed. There is a need to further corroborate the relationship between unfair payment practices and construction business performance using archival data. Thus, a Pearson's coefficient corelation analysis was used to determine the statistical relationship between these variables, performed in section 5.7.4.1 and chapter 6.

5.8.5 Relationship between Unfair Payment practices and Construction Business Performance

Table 5.8.5 presents a summary of the study participants' agreements and their own construction business performance that completed the questionnaire. Note that out of the 179 study participants that completed survey one; the researcher managed to obtain a complete five years' worth of financial statements for 34 of the study participants' companies.

Key assumption: the study assumed that the study participants completed the questionnaire based on their judgement regarding companies' financial performance; thus connoting that as they completed the questionnaire, they were indirectly scoring their own business performance. Column B to F in table 5.7.4.1 presents data obtained from Companies' House using the Freedom of Information Act. Column B denotes thecontractor's average turnover over 5 years; Column C- contractor's average operatingmargin over 5 years; Column D; - contractor's average ROCE over 5 years; Column E-contractor's average liquidity ratio over 5 years; and Column F- contractor's average debt ratio over 5 years.

The study considered five years' worth of financial statements because most financial statements are based on five years of financial averages. Lucey (2009) argued that five years of average costs and commercial data are significant to make informed financial decisions. Therefore, to determine the statistical relationship, the Pearson correlation analysis (using SPSS 25) was used to conduct the correlation coefficient (r²) value. A set of r values were obtained for columns A and B; columns A and C; columns A and D; columns A and E; and columns A and F, respectively.
Table 5.8.5 Summary of participants' agreements on unfair payment practices and contractor's business performance over a five-year period

S/No.	Percentage agreement on unfair payment practices by study participants'	Contractor's average turnover over 5 years	Contractor's average operating margin over 5 years	Contractor's average ROCE over 5 years	Contractor's average liquidity ratio over 5 years	Contractor' s average debt ratio over 5 years
	Α	В	С	D	E	F
Company 1	73.33	11152199000	1.27	30.61	0.70	1.30
Company 2	65.88	34692000000	1.35	22.47	0.93	1.27
Company 3	65.56	10741	5.43	28.72	1.39	1.34
Company 4	51.11	14806	3.32	27.63	1.06	1.20
Company 5	63.33	86309222	3.10	10.97	0.86	10.60
Company 6	68.89	87596	8.75	34.36	0.94	0.98
Company 7	55.56	847554156	1.07	4.59	0.84	0.97
Company 8	72.22	1039446234	1.72	10.32	1.03	0.89
Company 9	56.67	2175796639	2.77	2.77	1.29	1.46
Company 10	46.67	346674103	2.88	14.61	0.49	1.17
Company 11	58.89	12053.7	5.11	39.97	1.03	1.34
Company 12	62.22	7386306	6.48	29.19	1.11	1.08
Company 13	68.24	270424700	13.13	29.10	1.44	0.90
Company 14	51.43	6063016	4.36	24.44	0.69	0.89
Company 15	55.56	8032652	5.98	6.44	1.20	1.46
Company 16	57.65	7056800	2.50	5.46	0.94	1.31
Company 17	60.00	25257200	3.73	12.63	0.86	1.13
Company 18	52.94	4430006000	3.37	8.43	1.34	0.96
Company 19	52.00	4377033844	5.10	21.94	1.12	5.10
Company 20	71.76	5114614180	3.58	24.50	0.96	1.16
Company 21	67.78	3901900000	1.48	12.67	1.11	1.13
Company 22	60.00	2706275000	5.95	8.56	0.67	0.52
Company 23	51.11	3902411900	8.07	28.68	0.62	1.19
Company 24	64.44	2220986558	5.05	19.93	0.91	1.28
Company 25	64.44	3612872000	3.20	26.33	1.26	1.46
Company 26	56.67	2201518000	11.53	9.62	0.59	2.25
Company 27	56.67	2285283000	6.08	11.36	1.02	1.21
Company 28	50.00	2757403000	3.78	4.57	1.13	1.36
Company 29	68.89	2042081384	2.31	17.02	1.14	1.20
Company 30	57.78	2377218070	3.21	22.44	1.16	1.55
Company 31	66.67	4330267000	2.59	25.78	1.02	1.35
Company 32	65.56	2757403000	3.78	25.38	1.09	1.36
Company 33	73.33	2727433000	3.60	15.27	0.86	1.24
Company 34	62.22	1435730000	15.20	49.33	0.86	2.60
Mean	61	3054257681	4.73	19.59	0.99	1.65

Figure 5.8.5(a) presents a scatter diagram showing the statistical relationship of study participants; percentage agreements concerning unfair payment practices and a contractor's average operating profit margin over a five year period. The R^2 value displayed via the scatter diagram produced a (p) = 0.01, n= 34, p<0.05 which shows a weak monotonic relationship between the variables. Moreover, the Pearson correlation coefficient produces a (u = 0.11) denoting a very weak relationship. This implies there is no significant relationship between the percentage agreements of the study participants regarding unfair payments practices and contractor's operating profit margin. Please refer to Table 6.4.5 for an interpretation of the correlation coefficients.



Figure 5.8.5.a: Scatter diagram comparing values of percentage agreements of study participants on unfair payment practices to a company's average profit margin (Column C v. Column B)

Figure 5.8.5 (b) illustrates a scatter diagram showing the statistical relationship of the percentage agreements of study participants concerning unfair payment practices and contractor's average return on capital employed over a five year period. The R^2 value displayed via the scatter diagram produced a (p) = 0.05, n= 34, p<0.05 showing a linear relationship between agreement on unfair payment practices and contractor's capital employed.

Moreover, the Pearson correlation coefficient produces a (u = 0.25) denoting a somewhat weak relationship. This implies there is a somewhat slight relationship between the percentage agreements of study participants regarding unfair payments practices and contractor's return on capital employed.



Figure 5.8.5 b: Scatter diagram comparing values of the percentage agreements of study participants on unfair payment practices to contractor's average ROCE (Column D v. Column B)

Figure 5.8.5c illustrates a scatter diagram showing a statistical relationship of study participants' percentage agreements concerning unfair payment practices and a contractor's average liquidity ratio over a five year period. The R² value displayed via the scatter diagram produced a (p) = 0.02, n= 34, p<0.05 showing a weak monotonic relationship between agreement on unfair payment practices and contractor's liquidity. Moreover, the Pearson correlation coefficient produces a (u = -0.010) denoting a very weak negative relationship. This implies there is no significant relationship between the percentage agreements of the study participants regarding unfair payments practices and contractor's liquidity ratio.



Figure 5.8.5c: Scatter diagram comparing values of the percentage agreements of study participants on unfair payment practices to a contractor's average liquidity ratio (Column E v. Column B)

Furthermore, figure 5.8.5 d illustrates a scatter diagram showing a statistical relationship of study participants' percentage agreements concerning unfair payment practices and a contractor's average debt ratio over a five-year period. The R² value displayed via the scatter diagram produced a (p) = 0.06, n= 34, p<0.05 denoting a weak relationship between agreement on unfair payment practices and a contractor's debt ratio. Moreover, the Pearson correlation coefficient produces a (u = -0.08) denoting a very weak negative relationship. This implies there is no significant relationship between the percentage agreements of study participants regarding unfair payments practices and contractor's debt ratios.

These findings are similar to Baum's et al. (2007) discovery about American industrial companies; that there was no significant relationship between companies' debt margin and business financial performance.



Figure 5.8.5d: Scatter diagram comparing values of the percentage agreements of study participants on unfair payment practices to a contractor's average debt ratio (Column F v. Column B)

The overall findings from the four correlation analyses reveal there is no significant relationship between the percentage agreements of the study participants and contractor's business performance. Therefore, the null hypothesis (H_0) should be rejected, and the alterative hypothesis (H_1) should be accepted.

5.8.6 Objective 5- To evaluate the role of the client-contractor relationship regarding unfair payment practices

Question 10 in survey 2 was asked to enable the researcher to measure the role of the client-contractor relationship regarding unfair payment practices in the UK construction industry. Figure 5.8.6 presents findings regarding this objective. 8% 'very strongly agreed', 53% 'strongly agreed', 22% 'agreed', 6% 'disagreed', 9% 'strongly disagreed', and 2% 'very strongly disagreed'.

The findings in figure 5.8.6(a) reveal that contracting parties' consideration to protect client-contractor relationships contributes to unfair payment practices in the construction industry. For instance, Coopla (2018) argued that before Carillion Plc collapsed in 2018, the company used its client's position to influence and induce longer

payment terms of 120 days on its subcontractors and suppliers. Unfortunately, majority of subbies and supplies affected by longer and unfair payment terms rarely complained because of their desire to protect the client-contractor relationship.



Figure 5.8.6a Role of the client-contractor relationship to unfair payment practices

Furthermore, question 11 in survey 2 asked the study participants to express their perceptions of construction clients' attitudes to unfair payment issues in the construction industry. For easy identification, question 11 in the questionnaire was benchmarked against five of the most common factors acknowledged in the study's literature, including:

- 1. perceived as normal industry practice
- 2. contractors agreeing to longer payments terms
- 3. indifference to unfair payment practices
- 4. very concerned about contractors' payments
- 5. perceived as a form of finance

Figure 5.8.6b illustrates that 44% of study participants believe that clients see unfair payment practices as normal industry practice; 23% believe that the problem is not client-related but rather contractors agreeing to longer payment terms; 19% agree that clients use unfair payment practices as a form of finance, 11% believe that clients are simply indifferent to unfair payment practices; while 3 % believe that clients are very concerned about unfair payment practices. Other contributory factors identified by the study participants regarding clients' perceptions of unfair payment practices include the following: "undercapitalised" nature of the construction industry; none of their concern; industry culture; and not considered as project objectives in terms of time, cost and quality.



Figure 5.8.6b Clients' perceptions of unfair payment practices

5.8.7 Measurement of Supplementary factors about unfair payment practices

5.8.7.1 Impact of unfair payment practices

Table 5.8.7.1 presents the measured additional impact of unfair payment practices. The table illustrates the mean scores and relative important index (RII) for five predetermined rank factors identified from the study's literature. Question 22 in survey 1 measured participant agreement regarding predetermined effects listed in table 5.8.7.1 using a five-point Likert scale of 5= 'very high effect' to 1= 'very low effect'.

The findings in table 5.8.7.1 show that unfair payment practices create cash flow difficulties for contractors with the highest mean score of 3.81, followed by impact on administration and company management with an average score of 3.68, diminished

organisational reputation with a mean score of 3.43, bullying of construction firms with an average score of 3.33 and increment of credit crunches with a mean score of 3.19.

Usually, when payments are delayed to contractors their cash flows are negatively disrupted by both incoming payments (receipts) and outgoing payments. This is because cash is a strong pedigree (bloodline) that drives most businesses. Any interference to cashflow often results in disastrous outcome for business, particularly for SMEs contractors who largely dependent cash to run their everyday operations and pay suppliers or subcontractors working on a project.

These findings align with Richard and Mori's (2018) study which inferred that unfair payments affected the administration and management of companies due to high costs incurred in chasing payments. Usually, costs incurred by contractors normally relate to financing, additional time and morale hazard problems. For example, BACS payment schemes (2019) confirmed that on average, UK construction businesses spent 130 hours per year chasing late payments, with an approximate cost of £1,500 per business.

	Effects	Mean	RII	Rank
1	Creates cash flow difficulties	3.81	0.76	1
[Affects the administration and	[
2	management of a company	3.68	0.74	2
[Diminishes organisational	[
3	reputation	3.43	0.69	3
	Results in bullying of construction			
4	firms	3.33	0.67	4
5	Increases credit crunches	3.19	0.64	5

Table 5.8.7.1 Effect of unfair payment practices

5.8.7.2 Disparity in commercial bargaining power due to unfair payment practices

Question 4 in survey 2 was asked to enable the researcher to measure the role of disparities in commercial bargaining power between parties in unfair payment practices. Findings from the study reveal that parties with lower bargaining power, especially subcontractors and suppliers, find it difficult to challenge or influence the

payment terms exerted on them by clients due to the imbalance in commercial bargaining power between the parties.

The study participants identified that organisation hierarchy structure in construction industry basically subjected subcontractors and suppliers to a weaker bargaining position in the supply chain. Hence, clients used their commercial bargaining power to input unfavourable payment terms and practices. Figure 5.7.6.2 shows that majority of study participants (42%) 'strongly agreed', 16% 'very strongly agreed' and 13% 'agreed' that clients used their commercial bargaining powerto input unfavourable payment terms to contractors.



Figure 5.8.7.2 Disparity in commercial bargaining power between parties

5.8.7.3 Potential of emerging technologies in minimising unfair payment practices

Figure 5.8.7.3 presents the study participants' perceptions regarding the use of emerging technologies in minimising unfair payment practices in the UK construction industry. Participants were asked to evaluate the potentiality of emerging technologies to alleviate unfair practices.

The findings reveal that majority of participants were of view that escrow/ project bank accounts, automated payment systems and 5D-BIM have potential to

minimise unfair payments in the construction industry. 10% of study participants ranked escrow /project bank accounts with 'moderate potentiality' followed by automated payment systems with 7 % and 5D-BIM 6 %. On the other hand, the study participants ranked other emerging technologies with 'very low potentiality' to minimise late payments. 83% of study participants scored Smart contracts with 'very low potentiality' followed by Agresso 4 Unit with 79% and Block chain with 75%.



Figure 5.8.7.3 Potentiality of emerging technologies in minimising unfair payment practices

5.9 Results and findings

Findings from this chapter provide meaningful insights into unfair payment practices in the UK construction industry. The profile of the study participants reveals that a significant proportion had more than 5 years' work experience in the UK construction industry, though some participants were not affiliated with professional bodies. The summary of quantitative findings reveals that bespoke contracts (i.e. modified standard forms of contracts) are the most commonly used followed by JCT suite of contracts. Arguably, the former reaffirms the inappropriateness and amendment of clauses to standard forms of contract. Perhaps, amendments to SFoCs contribute to growing cases of unfair payment practices in the UK construction industry.

In terms of the study's objectives, the findings reveal that three-quarters of the study participants strongly disagreed or disagreed that payment provisions in standard forms of contracts (SFoCs) helped to minimise unfair payment occurrences, as illustrated in figure 5.8.1.

The summary of the findings in this chapter also reveals that unfair payment practices are common features in the construction industry. The study specifically identifies tier 2 clients (principal contractors) as the most notorious late payers with a score of 47%, followed by tier 1 clients (main construction clients) with 35% and tier 3 clients (subcontractors acting as clients to other subcontractors) with 18%. The study identified that over 35% of unfair payment cases are linked to late payments (34%) followed by unleased retentions to contractors, as illustrated in figure 5.8.2.1

The study confirms that study participants considered longer payment terms with a payment duration ranging from "31 to 50" days and "51 to 90 days" to be standard in the construction industry. The implications of this findings denote that majority of construction contractors' invoices are paid after the agreed 30 days, as stipulated in most contract documents.

Contemporary causes of unfair payment practices identified by the study were categorised into five main groups: cultural, industrial, technical, regulatory and others causes. The findings indicate that the most common contemporary causes of unfair payment practices include the use of late payments to contractors as a cash flow strategy to boost a client's finances, the prevailing culture, the client-contractor relationship, "pay when paid" and "work first get paid later" to be common causes of unfair payment practices. Cash flow strategy was ranked first with the highest score of 3.59 by the study participants.

In addition, a correlation coefficient analysis was conducted to ascertain whether unfair payment practices influenced contractors' business performance. The findings in figure 5.8.5 (a), (b), (c) and (d) affirm a weak monotonic relationship between the

percentage agreements of the study participants and company's average operating profit margin, return of capital employed, liquidity and debt ratios (p = 0.01, p = 0.06, p = 0.02 and p = 0.06). However, the use of financial ratios to measure business performance is considered to be subjective in understanding the true performance of a business.

5.10 Summary

This chapter presented the study's quantitative analyses. The chapter began with a profile of the study participants, Wilcoxon signed rank, descriptive statistics and host of statisticalanalyses regarding the two sets of questionnaire surveys were conducted by the study. Cronbach's alpha test was conducted to measure the internal reliability of the quantitative data. Subsequently, a wide range of quantitative analyses were conducted and the findings well documented for a proper understanding of the study. The next chapter presents the study's archival analyses.

CHAPTER 6

Archival Data Analysis

6.1 Introduction

This chapter presents the analysis of the study's archival data gathered from different documentary sources. As previously discussed in Chapter four, archival data was considered to be vital to this study because it helps to understand in detail, the true nature of unfair payment issues in the UK construction industry. Moreover, this chapter analyse real time payment performance data in the construction industry and its effect on contractor business performance. The archival analyses are presented in three sections: **Section 1** provides a brief description of the procedure used to collect the archive data; **Section 2** presents the payment performance of different tiers of construction clients in the UK together with their business performance analyses; **Section 3** reviews fifteen case studies of small and medium-sized companies that went into administration or insolvency processes due to unfair payments practices. The companies selected faced financial pressures and cashflow problems that led to their collapse. The analyses and discussions in this chapter are presented in various tables and graphs whilst conclusions are presented in chapter 8.

6.2 Archival data collection for construction clients' payment performance and case studies

Section 4.7.3 in the study's methodology chapter provided a detailed explanation of how archival data was obtained by the researcher. It must be noted that majority of data used for this section were drawn from online reports and surveys published by government sources. The published payment data 2019/2020 from BuildUK (here) was extracted for a better understanding of construction clients' payment performance, while financial statements from Companies House (here) were obtained to measure contractors' business performance. Furthermore, case studies concerning small and medium-sized construction companies were obtained from administrators' reports filed in the Company Check website (here)

Overall, majority of data in this section was drawn from credible UK government sources and can be confirmed and verified.

6.3 Payment performance of different tiers of construction clients and contractors

Table 6.3.1 illustrates the payment performance of Tier 1 clients that pay for high value construction deliverables. Findings from the data indicate that majority of private Tier 1 clients seem to be prompt payers within agreed contractual terms, except for Rolls Royce (71 days), Peabody (53 days), Lidli (48 days), the National Grid and Aldi(40 days).

Moreover, looking at the average percentage of invoices that are not paid within the agreed terms, Peabody had the highest percentage of invoices at 48%, followed by SHE Transmission at 37%, Great Portland Estates at 35%, St Modwen at 34%; The National Grid at 17%, British Land at 14% and Centrica at 1%. As to the percentage of invoices paid within 60 days, 89% of Tier 1 clients seemed to be settling their invoiceswithin 60 days of the date of invoice receipt to the date the contractor received payment. This clearly illustrates a pattern of poor behaviour, as none of the companieshad exceptions to 30 days' standard payment terms.

Table 6.3.1 Payment Performance for Tier	1 Main (private)	construction	clients to c	ontractors
(adapted from Build UK 2019/2020)				

No.	Date when data was collected	Company Name	Percentage (%) of invoices not paid within agreed terms	Numbers of days taken to pay invoices	Percentage (%) of invoices paid within 60 days
1	25/10/2019	Great Portland Estates	35	31	91
2	01/04/2020	Landsec	5	17	96
3	29/07/2019	Heathrow	7	22	99
4	30/04/2020	British Land	14	17	97
5	30/04/2020	Berkeley Homes	10	25	94
6	01/03/2019	Centrica	1	17	92
7	30/04/2020	Euro Garages	1	24	100
8	30/04/2020	Places For People	6	24	95
9	30/04/2020	Lidli	6	48	84
10	30/04/2020	Aldi	6	40	97
11	30/04/2020	Rolls-Royce	8	71	24
12	30/04/2020	Valero Energy	9	22	99
13	30/04/2020	IM Properties	11	18	97
14	30/04/2020	National Grid	17	40	97
15	30/04/2020	St Modwen	34	32	95
16	30/04/2020	SHE Transmission	37	29	95
17	30/04/2020	Peabody	48	53	72
		Average	16	21	90

Furthermore, Table 6.3.2 illustrates payment performance of public Tier 1 clients that engage and procure high value works using main contractors in the UK. Findings from Table 6.3.2 reveal that majority of public Tier 1 clients pay their invoices within 30 days, with an average of 96% of invoices settled in 30 days and 85% of invoices issued within 5 days from the date of receipt of invoice. This implies that most government clients and sectors are good in making prompt payments, as agreed in their contracts.

However, it could be argued there have been many headlines and misconceptions about public clients paying beyond the agreed terms and challenging their supply chain payment terms. For instance, the National Federation of Builders (NFB) have consistently reported that public clients, including central government, do not pay on time and that many of its members do not challenge their terms

No.	Date when data was collected	Client Name	Percentage (%) of Invoices Paid within 30 Days	Percentage (%) of Invoices Paid within 5 Days
1	31/03/2020	Cabinet Office	97	88
2	31/03/2020	Department for Business, Energy & Industrial Strategy (BEIS)	99	94
3	31/03/2020	Department for Digital, Culture, Media & Sport (DCMS)	98	86
4	31/03/2019	Department for Education (DfE)**	91	66
5	31/03/2020	Department for Environment Food & Rural Affairs (DEFRA)	93	85
6	31/03/2020	Department for International Development (DfID)	98	89
7	31/03/2020	Department for Transport (DfT)	99	95
8	31/12/2019	Department of Health & Social Care (DHSC)	94	49
9	30/09/2019	Department of Work and Pensions (DWP)	96	97
10	30/09/2019	Foreign & Commonwealth Office (FCO)	100	98
11	31/03/2020	HM Revenue and Customs (HMRC)	99	94
12	31/03/2020	HM Treasury (HMT)	96	80
13	31/03/2020	Home Office	98	93
14	31/03/2020	Ministry for Homes Communities and Local Government (MHCLG)*	99	87

Ministry of Defence (MoD)

Ministry of Justice (MoJ)

Birmingham City Council

Highway England

Sheffield Council

Average

100

97

92

88

91

96

92

87

83

65

92

85

15

16

17

18 19 31/03/2020

31/03/2020

05/01/2019

25/02/2019

15/01/2019

Table 6.3.2 Payment performance: Tier 1 public clients acting as Main Construction Clients to
contractors (adapted from Build UK 2019/2020)

On the other hand, payment performance data between Tier 2 clients (main client) to their subcontractors and suppliers extracted from Build UK show that most Tier 2 clients do not pay their invoices within the contractual agreed date. Moreover, 90% of Tier 2 clients settle their invoices within 60 days. The payment performance data of 26 of the biggest construction contractors reveals that Multiplex had the longest payment dates of 55 days on average, followed by Seddon (51days), Engine (50 days); Taylor Wimpey (48 days), BAM Nutall (47 days) and Visitry Partnerships (44 days). Furthermore, the same companies accounted for the highest proportion of invoices not paid within the agreed terms, ranging from 62% to 17%.

Conversely the best payment performance contractors were IIKE Homes with 20 days of payment terms on average, followed by Vinci at 26 days, Morgan Sindall and Redrow Homes at 27 days and Mace at 28 days. Further findings from table 6.3.3 reveal the average proportion of invoices paid within 60 days and the average proportion of invoices paid within 60 days and the average proportion of invoices paid after 30 days' terms. Overall, these statistics show that Tier 2 clients have longer payment terms of 46 days on average.

No.	Date when data was collected	, Company Name	Numbers of days taken to pay invoices	Percentage (%) of invoices not paid within agree terms	Percentage (%) of invoice paid within 60 days
1	30/11/2019	Interserve	32	38	94
2	24/02/2020	Murphy Group	37	23	86
3	27/11/2019	John Sisk & Son	32	17	94
4	10/02/2019	Wates	32	17	97
5	10/02/2020	Balfour Beatty	41	31	89
6	26/07/2010	BAM Construct	38	37	00
7	17/02/2020	ISC	3/	23	92 84
<u> </u>	30/07/2010	Kior	34	23	80
0	26/10/2010	Multiplox	55	47	75
10	31/11/2019	Amov	- 33	51	75 95
11	26/07/2019	Mace	23	26	95 97
12	15/02/2020	Osborne	20	20	94
13	25/01/2020	Vinci	29	13	95 97
1/	26/02/2020	Bouvques	/1	27	02 02
15	27/11/2019	Sir Robert McAlpine	31	21	92
16	26/11/2019	Galliford Try	42	10	89
17	27/07/2019	Morgan Sindall	27	12	98
18	26/07/2019	Vokler Wessels	21	24	94
19	26/07/2019	like Homes	20	55	98
20	19/03/2020	Skanska		8	92
21	24/11/2019	Canary Wharf Contractors	37	20	89
22	20/01/2020	Willmott Dixon	31	12	99
23	10/02/2020	Seddon	51	18	90
24	25/10/2019	Laing O'Rourke	31	22	90
25	25/10/2020	Melaren Construction	42	62	80
26	04/01/2020	Engie	50	1	03
27	04/01/2020	Morrison Utility Services	33		92 94
28	10/02/2020	Winvic Construction	37	15	97
29	10/02/2020	Wilson James	33	16	96
30	10/02/2020	Lendlease	33	16	96
31	10/02/2020	Bowmer and Kirkland	37	27	95
32	10/02/2020	Graham Construction	42	31	83
33	10/02/2020	BAM Nutall	47	32	88
34	10/02/2020	Costain	38	32	90
35	10/02/2020	Roberston Construction	34	33	93
36	10/02/2020	Bloor Homes	37	3	91
37	10/02/2020	Cala	37	10	88
38	10/02/2020	Redrow Homes	27	12	93
39	10/02/2020	Vistry Partnerships	44	17	87
40	10/02/2020	Barrat Developments	26	19	98
41	10/02/2020	Persimmon Homes	31	19	93
42	10/02/2020	Crest Nicholoson	37	24	90
43	10/02/2020	Miller Homes	43	30	85
44	10/02/2020	Taylor Wimpey	48	32	86
45	10/02/2020	Bellway Homes	34	48	92
		Average	46	25	90
	1		10		

Table 6.3.3 Payment Performance: Tier 2 Clients (Main Contractors) to subcontractors(adapted from Build UK 2019/2020)

Furthermore, table 6.3.4 illustrates the payment performance of Tier 3 clients, as subcontractors who often sublet part of their building works to other subsubcontractors and suppliers in the construction industry.

Findings from Table 6.3.4 reveal that Tier 3 clients were the worst payers among the three tiers of construction clients, with Cape Industrial Services taking the lead with 101 days to pay on average, followed by Brown and Mason (84 days), Spie (75 days), T Clarke (70 days); Wood Industrial Services (69 days), Brand Energy &

Infrastructure (68 days) and T.E Scudder (54 days). The quickest payer was SSE Contracting with the shortest payment terms of 16 days on average, followed by Overbury (21 days), Stanmore (23 days) and Portview (29 days). However, comparing the findings of Table 6.3.4 to 6.3.3 and 6.3.1, Tier 3 clients had the lowest proportion (74%) of invoices paid within 60 days and 26% were not paid within the agreed terms.

No.	Date when data was	Company Name	Numbers of days taken to	Percentage (%) of invoices not paid within	Percentage (%) of invoice paid within
	collected		pav invoices	agree terms	60 davs
1	10/04/2020	Expanded Structure	37	17	77
2	10/04/2020	Carey Group	37	50	74
3	10/04/2020	Tamdown	59	17	57
4	10/04/2020	AJ Morrisoe and Sons	33	5	95
5	10/04/2020	Byrne Brothers	46	4	87
6	10/04/2020	Getjar	45	6	90
7	10/04/2020	Brown and Mason	84	47	32
8	29/06/2019	McGee Group	46	12	76
9	30/11/2019	Erith	48	11	79
10	30/11/2019	Keltbray	55	15	57
11	28/01/2019	Keller	46	57	85
12	30/11/2019	Cementation Skanska	52	10	90
13	10/04/2020	William Hare	36	16	79
14	30/10/2019	Severfield	42	16	95
15	10/04/2020	Billington	50	11	75
16	28/10/2019	T.E. Scrudder	62	65	59
17	10/04/2020	CJ O'Shea	44	8	84
18	10/04/2020	Van Elle	72	48	25
19	10/04/2020	Roger Bullivant	54	17	60
20	10/04/2020	Imtech	60	27	40
21	10/04/2020	NG Bailey	50	60	71
22	10/04/2020	SSE Contracting	16	48	100
23	29/07/2019	Spie	75	49	74
24	10/04/2020	J. Reddington	27	6	84
25	10/04/2020	BW Interiors	29	40	60
26	10/04/2020	Bachy Soletanche	45	49	83
27	10/04/2020	Parkeray Interiors	33	0	100
28	10/04/2020	Overbury	21	10	96
29	10/04/2020	Portview	29	10	97
30	10/04/2020	Stanmore	23	10	100
31	10/04/2020	Willmott Dixon Interiors	30	17	97
32	10/04/2020	ISG Fit Out	32	20	94
33	10/04/2020	Skanska Rashleigh Weatherfoil	55	11	88
34	10/04/2020	Crown House Technologies	33	21	84
35	10/04/2020	T Clarke	70	30	34
36	10/04/2020	SES	37	68	86
37	10/04/2020	Prater	54	8	62
38	10/04/2020	Novus Property Solutions	33	9	94
39	10/04/2020	Cape Industrial Services	101	38	46
40	10/04/2020	Enigma Industrial Services	58	42	46
41	10/04/2020	Ltd Wood Industrial Services	90 PA		31
	10/04/2020	Brand Energy &			
42	10/04/2020	Infrastructure	68	42	80
		Average	50	26	74

Table 6.3.4 Payment Performance: Tier 3 Clients (Subcontractors) to sub-subcontractors and suppliers (adapted from Build UK 2019/2020)

6.4 Contractors' business performance using Archival data

Data obtained in Table 6.3.3 and 6.3.4 were used to conduct further analyses regarding companies' business performance. Four financial ratios, namely operating profit margin (OPM), return of capital employed (ROCE), current ratio (CR) and debt ratio (DR) were considered for measuring contractor's business performance.

Operating profit margin is the percentage of operating profit to the cost of sales or revenue (Atrill and McLaney 2008); and is usually expressed as follows:

Operating profit margin = Operating profit x 100 Cost of sales/revenue

OPM measures the ability of a business to derive profits from its cost of sales or revenues, for instance overheads, office rents, salaries, bonuses, production costs and others. The rule of thumb is that if business 'cost of sales' increases, the OPM decreases as the business is spending too much on its operations. Return of capital employed (ROCE) measures a firm's performance by dividing operating profit by capital employed. Atrill and McLaney (2008) argued that ROCE is the most fundamental measure of business performance as it describes a business's ability to generate returns from its capital. ROCE is usually expressed as follows:

```
Return on capital employed (ROCE) =
```

Operating Profit x 100 Capital employed

Current ratio analysis measures a business's liquidity as the ability to pay its shortterm obligations. The higher the CR (over 1.0), the more liquid the business is at paying its debts on time. The current ratio is calculated as follows:

> Current ratio = Current assets Current liabilities

The debt ratio indicates a business's long-term debt paying ability (Quesada 2019). A lower ratio not exceeding 0.5 indicates that it is less likely that business debts are high compared to assets; and greater than 0.5 shows that most of the business assets are financed by debts. In other words, the business is at risk of going into insolvency or bankruptcy. Debt ratio is calculated as follows:

> Debt ratio = Total liabilities Total assets

6.4.1 Operating Profit Margin of a Contractor's Business (Tier 3 And 4 Clients')

Table 6.4.1 presents a summary of four financial ratios used to conduct analyses for contractors' business performance over a five year period. The companies from the listed table 6.3.4 were merged with a total of 40 small and medium-sized construction companies. Then financial statements were extracted by the researcher from Companies House, whereby a five years' average of financial statements were considered for measuring business performance. Lucey (2009) argued that five years' average costs and commercial data was significant to make informed financial decisions.

Columns C to F in Table 6.4.1 present data obtained from Companies' House website, whereby Colum C denotes contractor's average operating margin over 5 years; Column D, contractor's average ROCE over 5 years; Column E, contractor's average liquidity ratio over 5 years and Column F, contractor's average debt ratio over 5 years. Colum A presents the average time frame for honouring invoices by companies extracted from Table 6.3.4. and Column B shows the difference between average time frame for honouring invoices and standard payment term of 30 days. A full analysis of each business's turnover, operating profit, shareholder funds (equity), current liabilities current assets, total assets and total liabilities is attached in appendix I.

Overall, the mean score in column C of table 6.4.1 seems to reflect precarious high profit margins for smaller and medium sized businesses. The FTI consulting report (2018) argued that 'though margins are significantly low in UK construction; the average turnover profitability for a smaller business is higher compared to bigger companies as it staggers from 1.5% to 8%". A possible explanation for this higher margin is perhaps due to the size of these companies, their lower risk profiles, diversification of clients, higher valued assets and lower operating and administrative costs that are common for smaller businesses.

Table 6.4.1: Summary of time frame for honouring invoices and average construction contractors' business performance over a five-year period

S/No.	Average time frame for honouring invoices	Difference between average time frame for honouring invoices and industry standard payment terms (30 days)	Average 5 years' operating profit margin ratio	Average 5 years' return of capital employed (ROCE)	Average 5 years' Liquidity ratio	Average value 5 years' Debt ratio
I	Α	В	С	i D	E	F
Company 1	50	20	2.96	18.54	0.78	0.64
Company 2	66	<u>i 36</u>	5.31	28.08	0.87	0.60
Company 3	52	22	5.85	6.44	0.99	0.71
Company 4	44	14	2.85	10.40	1.08	0.78
Company 5	54	24	7.06	28.04	0.86	0.75
Company 6	45	15	3.73	18.95	0.81	0.78
Company 7	42	i 12	3.62	16.30	0.92	0.68
Company 8	42	12	9.18	8.95	1.12	0.54
Company 9	43	13	1.59	6.37	0.78	0.57
Company 10	39	i 9	5.69	9.57	0.97	1.46
Company 11	45	15	4.36	33.00	0.98	0.64
Company 12	47	17	4.23	29.28	0.81	1.32
Company 13	52	22	4.35	10.36	0.93	1.20
Company 14	40	10	2.40	8.22	0.79	0.59
Company 15	35	5	3.59	16.15	0.81	0.51
Company 16	47	17	3.33	2.88	0.91	0.81
Company 17	44	14	2.96	11.22	0.97	1.13
Company 18	35	5	16.52	16.16	1.10	1.36
Company 19	32	2	2.66	12.26	0.53	0.66
Company 20	41	11	8.10	14.81	0.85	0.67
Company 21	34	4	4.23	23.46	0.03	0.33
Company 22	33	3	8.27	33.44	0.58	0.49
Company 23	44	14	1.11	15.47	0.58	0.67
Company 24	59	29	2.45	35.82	0.43	0.84
Company 25	61	31	1.92	22.46	0.99	0.87
Company 26	36	6	7.64	32.64	0.43	0.66
Company 27	62	32	5.77	25.25	0.91	0.34
Company 28	36	<u>6</u>	20.86	29.18	0.92	0.91
Company 29	55	25	7.97	16.41	0.91	0.91
Company 30	45	15	6.17	41.11	0.78	0.35
Company 31	86	<u>56</u>	2.66	<u>5.41</u>	0.53	0.66
Company 32	47	17	2.26	18.06	0.89	0.79
Company 33	39	9	6.64	14.59	0.97	0.35
Company 34	43	13	4.24	13.98	0.77	0.61
Company 35	33	3	4.12	5.79	0.53	0.20
Company 36	43	13	4.86	4.50	0.98	0.55
Company 37	51	21	10.62	<u>19.17</u>	0.18	0.50
Company 38	46	16	7.19	18.71	0.16	0.90
Company 39	101	71	1.24	36.03	0.31	0.31
Company 40	50	20	4.84	9.76	0.70	0.66
Mean	47.18	17.18	5.71	18.18	0.68	0.71

6.4.2 Return on capital employed for a contractor's business (Tier 3 and 4 clients)

Return on capital employed was used in this study as a fundamental measure of business performance. ROCE measures a business's overall efficiency and profitability generated from capital employed. Singh and Yadava (2013) claimed that the use of ROCE is a stringent tool and measure for shareholders and investors to assess whether a business generates sufficient revenues and profits using its capital investments.

Column D in Table 6.4.1 shows a summary of average ROCE extracted from appendix I; whereby operating profit margin was compared to shareholder funds (equity) over a five year period. Overall, values from top to bottom (in column D) depict an uneven distribution of ROCE in small and medium-sized businesses. Moreover, a mean ratio of 18.18 indicates a high ROCE for businesses that seemed to be good or feasible in utilising shareholder funds and investments to generate profits.

However, it is pertinent to note that due to the undercapitalised nature of construction, ROCE is mostly reliant on large businesses to sustain their capital investments through heavy borrowings, use of trade credits and longer payment terms imposed on their contactors.

6.4.3 Liquidity ratio of a contractor's business (Tier 3 and 4 clients)

Column E in Table 6.4.1 shows a summary of a contractor's liquidity ratio measured against business's short-term liabilities to its assets. The liquidity or current ratio is a sole criterion for a business's survival and its ability to sell or exchange cash at short notice (Brealey et al. 2010). In other words, the liquidity ratio is used to measure a business's ability to meet its short-term financial obligations.

The mean score of 0.68 in column E of table 6.4.1 reflects the average liquidity ratio for construction contractors that is significantly lower when compared to other industries. Moreover, in the construction industry, a liquidity ratio less than 0.9 signifies that business is not financially buoyant. Possible explanations for such lower liquidity include insufficient working capital, fluctuating demand, poor credit, cashflow issues and late payments that impair contractor's financial performance.

6.4.4 Debt ratio for a contractor's business (Tier 3 and 4 clients)

Column F in Table 6.4.1 presents a summary of contractor's debt or asset ratio over five year period. The debt ratio measures the amount of debt compared to a business's total assets acquired. A mean score of 0.71 in column F of table 6.4.1 indicates that the average debt ratio for a smaller business is high and thus it is riskier for them to survive, as 71% of companies' assets are financed by debts. A debt ratio greater than 0.50 indicates a higher risk of insolvency, because the total amount of liabilities exceeds a company's' assets. Arguably, a higher debt ratio is synonymous with small and medium-sized contractors, as the majority have less fixed assets and volatile cashflows that result in a higher insolvency risk.

A key deduction from these ratio analyses is that small and medium-sized construction businesses appear to have little or no cash inflow perhaps when required to carry out their business performance, as most are constrained by larger contractors and by longer and more unfair payment terms. Small and medium-sized construction businesses are deemed to be undercapitalised and their reliance on payment for work done puts this category of business at a greater risk of debts and insolvency.

It must be noted that the financial business performance of companies used in these analyses were only for the small and medium-sized companies obtained by the researcher. Data for sole traders or 'man and van' businesses could not be obtained from Companies House due to insufficient transactions and turnover.

Column A of Table 6.4.1 illustrates the average time frame for honouring invoices for contractors' businesses, while column B represents the difference between the average time for honouring invoices and the industry's standard terms of 30 days. The mean score of 46 days illustrates a poor payment performance for contractors in settling their invoices beyond contractually agreed terms. Conversely, considering the construction industry norm, this may be seen as a reasonable time for contractors and suppliers to receive payments. The data reveals that some companies take much longer than 100 days to honour invoices, as illustrated in column A. The average

the majority of contractors take longer than two weeks to settle invoices from the agreed date, which is common practice in the industry.

6.4.5 Statistical analysis for the time frame in honouring invoices and construction business performance

The data from Table 6.4.1 was subjected to SPSS and used scatter diagrams (see figure 6.4.5a, b, c and d) to determine the linearity and distribution of the paired data. Moreover, a set of r values were obtained for column A and B; column A and C; column A and D; column A and E; and column A and F, respectively.



Figure 6.4.5a: Scatter diagram showing the relationship between delayed time frame in honoring contractors' invoices Vs. a business's average operating profit margin (Column B v. Column C)

Figure 6.4.5a illustrates a scatter diagram showing the statistical relationship between a delayed time frame for honoring contractors' invoices and average business operating profit margin over a five-year period. The R^2 value displayed via the scatter diagram produced a (p) = 0.04, n= 40, p<0.05 which shows a weak monotonic relationship between the variables.

Moreover, the Pearson correlation coefficient produces a (u = 0.21) denoting a somewhat weak relationship. This implies there is a slight and somewhat weak relationship between a delayed time frame for honoring invoices and the contractor's operating profit margin. Please refer to Table 6.4.5 for an interpretation of the correlation coefficients.



Figure6.4.5b: Scatter diagram showing the relationship between delayed time frame in honoring invoices and a business's average capital employed (Column B v. Column D)

Figure 6.4.5(b) illustrates a scatter diagram showing the statistical relationship between delayed time frame for honoring invoices and contractor's average return on capital employed over a five year period. The R^2 value displayed via the scatter diagram produced a (p) = 0.09, n= 40, p<0.05 showing a linear relationship between delayed time frame for honoring invoices and contractor's capital employed. Moreover, the Pearson correlation coefficient produces a (u = 0.16) denoting a very weak relationship. This implies there is no significant relation between additional days for receiving payments and contractor's return on capital employed.

Similarly, Figure 6.45c illustrates a scatter diagram showing the statistical relationship between additional days for contractors honoring invoices and a contractor's average liquidity ratio over a five year period. The R^2 value displayed via the scatter diagram produced a (p) = 0.09, n= 40, p<0.05 showing a weak monotonic relationship between agreement on unfair payment practices and a contractor's liquidity. Moreover, the Pearson correlation coefficient produces (u = -0.03) denoting a weak negative relationship. This implies there is no significant relationship between delayed payments and contractor's liquidity ratios.



Figure 6.4.5c: Scatter diagram showing relationship between a delayed time frame in honoring invoices and a business's average liqudity ratio (Column B v. Column E)



Figure 6.4.5d: Scatter diagram showing relationship between a delayed time frame in honoring invoices and a business's average debt ratio (Column B v. Column F)

Figure 6.4.5 d illustrates a scatter diagram showing the statistical relationship between additional days for honoring invoices and a contractor's average debt ratio over a five year period. The R^2 value displayed via the scatter diagram produced a (p) = 0.01, n= 40, p<0.05 denoting a weak relationship between delayed days and a contractor's debt ratio. Moreover, the Pearson correlation coefficient produces a (u = 0.04) denoting a very weak negative relationship. This implies there is no significant relationship between additional days for honoring invoices and a contractor's debt ratio.

Overall, findings from the four correlation analyses calculated in figure 6.4.5 a, b, c and d reveal there is no significant relationship between the average time frame for honoring contractors' invoices and a contractor's business performance for small and medium-sized companies. Therefore, the null hypothesis (H₀) should be rejected and the alterative hypothesis (H₁) should be accepted.

Note that the interpretation of Pearson's correlation coefficients and r-squared for the above values was based on Higgins' (2003) correlation coefficient table, as illustrated in table 6.4.5.

Coefficient	Relationship
±0.01 and ±0.19	Very weak relationship
±0.20 and ±0.39	Weak relationship
±0.4 and ±0.59	Moderate relationship
±0.6 and ±0.79	Strong relationship
±0.8 and ±1.00	Very strong relationship

Table 6.4.5 Interpretation of correlation coefficient (Higgins 2003)

6.4.6 Relationship between time frame in honouring invoices and construction business performance (Tier 2 clients)

Table 6.4.6 shows a summary of four financial ratios used in the analyses of contractor business performance, specifically for main contractors in the UK. This study has put together a total of 22 construction companies from table 6.3.3 whereas financial statements for the aforementioned were extracted from the Companies House website. A full analysis of each company's turnover, operating profit margin, return of capital employed (ROCE), current assets, current liabilities, total assets and total liabilities over five years is attached in appendix I.

Column C in Table 6.4.6 illustrates the average operating profit margin for a main contractor with reference to its annual turnover. Subsequently, column D presents a contractor's average return of capital employed; column E, a contractor's average liquidity ratio; and column F, a contractor's average debt ratio over a 5-year period in all analyses.

Overall, the mean scores of both the operating profit margin 4.34 and ROCE 25.73 indicate a significant deterioration of contractors' business performance as a result of increasing trade credits, for instance 60 to 90 days payments that constrain contractors' working capital. In recent times, between 2017-2020, difficulties in the economic climate caused by unprecedented events such as Brexit and COVID-19 have compelled many businesses to indulge in unfair payment practices to boost their working capital and profits. This is evident in the current ratio calculated in Column D in Table 6.4, whereby most large construction businesses have shown less capability in meeting their short-terms liabilities. The mean score of 0.71 in column E reflects the average liquidity ratio, while the mean score of 0.79 in column F represents the debt ratio for main contractors' business performance.

Similarly, a Pearson's correlation analysis was conducted with a squared correlation coefficient (r²) value to determine the statistical relationship between a delayed time frame in honouring invoices and a contractor's business performance. Using SPSS 25, the following data were subjected to SPSS scatter diagrams: Table 6.4.6 column B vs. C; column B vs. D; column B vs. E and column B vs. F as follows:

S/No.	Average time frame for honouring invoices A	Difference between average time frame for honouring invoices and industry standard payment terms (30) B	Average 5 years profitability of main contractors C	Average value for ROCE D	Average value for Liquidity Ratio E	Average value for Debt Ratio F
Company 1	120	90	3.89	13.88	0.81	0.97
Company 2	50	20	5.03	15.51	0.76	0.79
Company 3	44	14	5.43	21.53	0.79	0.75
Company 4	41	11	3.32	45.96	0.69	0.84
Company 5	57	27	3.10	43.41	0.86	0.85
Company 6	43	13	7.43	17.65	0.53	1.00
Company 7	46	16	3.01	43.54	0.85	1.00
Company 8	42	12	7.16	25.19	0.03	0.72
Company 9	43	13	1.38	24.68	0.58	0.64
Company 10	38	8	2.03	16.92	0.58	0.80
Company 11	45	15	5.11	43.09	0.87	0.75
Company 12	47	17	6.48	30.58	0.58	0.93
Company 13	52	22	5.63	16.30	0.57	0.62
Company 14	40	10	4.99	19.60	0.98	1.15
Company 15	35	5	5.98	11.71	0.75	0.68
Company 16	47	17	2.50	39.85	0.78	0.79
Company 17	44	14	3.62	13.13	0.67	0.88
Company 18	50	20	3.37	12.91	0.87	0.86
Company 19	47	17	5.10	22.80	0.58	0.55
Company 20	41	11	3.58	55.09	0.67	0.54
Company 21	41	11	1.48	22.93	0.89	0.68
Company 22	39	9	5.95	9.89	0.99	0.54
Mean	47.82	17.82	4.34	25.73	0.71	0.79

Table	6.4.6	Summary	of	time	frame	for	honouring	contractors'	invoices	and
construction contractors' average business performance over a five-year period										

Figure 6.4.6a presents a scatter diagram showing the statistical relationship between time frame for honoring contractors' invoices and business's average operating profit margin over a five year period. The R² value displayed via the scatter diagram produced a (p) =0.05, n= 22, p<0.05 which shows a weak monotonic relationship between the variables.

Moreover, the Pearson correlation coefficient produces (u = -0.07) denoting a very weak negative relationship. This implies there is no significant relationship between delayed payments and contractor's operating profit margin. Please refer to Table 6.4.5 for an interpretation of the correlation coefficients.



Figure 6.4.6a Scatter diagram showing relationship between delayed time frame in honoring contractors' invoices Vs. a business's average operating profit margin (Column B v. Column C)

Figure 6.4.6b below illustrates a scatter diagram showing the statistical relationship between delayed time frame for honoring invoices and contractor's average return on capital employed over a five year period. The R² value displayed via the scatter diagram produced a (p) = 0.02, n= 22, p<0.05 showing a linear relationship between delayed time frame for honoring invoices and a contractor's capital employed. Moreover, the Pearson correlation coefficient produced a (u=-0.13) denoting a very weak negative relationship. This implies there is no significant relationship between additional days for receiving payments and a contractor's return on capital employed. A possible explanation for such a result is because the size of the data obtained is smaller and poses a greater likelihood of obtaining a spurious figure.

Similarly, figure 6.4.6c illustrates a scatter diagram showing a statistical relationship between additional days for honoring invoices and contractor's average liquidity ratio over a five year period. The R² value displayed via the scatter diagram produced a (p) = 0.03, n= 22, p<0.05 showing a weak monotonic relationship between a delayed time frame for honoring invoices and contractor's liquidity. Moreover, the Pearson correlation coefficient produced (u =0.19) denoting a very weak negative relationship. This implies there is no significant relationship between delayed payments and contractor's liquidity ratios.



Figure 6.4.6b: Scatter diagram showing the relationship between delayed time frame in honoring invoices and a business's average capital employed (Column B v/ Column D)



Figure 6.4.6c: Scatter diagram showing the relationship between delayed time frame in honoring invoices and a business's average liquidity ratio (Column B v/ Column E)

Figure 6.4.6d below presents a scatter diagram showing a statistical relationship between additional days for honoring invoices and contractor's average debt ratio over a five year period. The R² value displayed via the scatter diagram produced a (p) = 0.07, n= 22, p<0.05 denoting a weak relationship between delayed days and contractor's debt ratio. Moreover, the Pearson correlation coefficient produced a (u = 0.48) denoting a somewhat weak relationship. This implies that for Tier 1 main clients, there is a slightly weak relationship between delayed time frame in honoring invoices and contractor's debt ratio. A possible explanation for this is perhaps due to the late receipt of payments affecting a contractor's cashflows and streaming the use of debt-constraint expropriation to secure extra cash.



Figure 6.4.6d: Scatter diagram showing the relationship between a delayed time frame in honoring invoices and a business's average debt ratio (Column B v. Column F)

Overall findings from the four correlation analyses carried out in figures 6.4.6 (a, b, c and d) and 6.4.6 (a, b and c) reveal there is a weak relationship between a delayed time frame for honoring invoices and a contractor's business performance.

Therefore, the null hypothesis (H_0) should be rejected and the alterative hypothesis (H_1) should be accepted.

6.5 Case Studies

This section reviews 15 case studies of small and medium-sized companies that fell into administration or insolvency as a result of unfair payments. The companies selected had several financial difficulties, cashflow problems and unfair payments that resulted in business failure. The companies were sourced via Companies House administrator reports (here) and Construction News (here) based on insolvency due to unfair payment practices and poor cashflow. Table 6.5 presents summary of small and medium-sized construction companies in terms of the following: age, nature of clients, turnover over a five-year period, outstanding payments, business performance, average payment days and reasons for failing.

The analysis from Table 6.5 reveals that most companies had a long history of trading and carrying out construction works in the UK and abroad and were established for more than 10 years. The nature of these companies' clients reveals that all tiers of clients are involved in unfair payment practices.

Data from table 6.5 shows an average turnover of £166m with 2.32% profits over a five years period. A full analysis of each company's turnover, operating profit margin, return of capital employed (ROCE), current assets, current liabilities, total assets and total liabilities over five years is attached in appendix J.

Findings from these analyses reveal a significant difference between turnover and contractor's operating profit margins, as turnover seemed to be higher compared to profits. From the business phrase, "turnover is vanity and profit is sanity but cash is king", it could be argued that insufficient working capital coupled with low priced tenders are the main reasons for contractors failing. For instance, it was reported earlier in 2018 that the Dawnus group had a high turnover of £170m while Shaylor Group had £152m, though in the following year both contractors called for technical liquidation and ceased doing business, thereby putting other companies into administration processes.

Perhaps it could be argued that the notion of profit margins and turnovers does not represent the true financial performance of a business. In other words, most contractors usually forecast their turnover and profits from tenders or works they have just secured or bid for, which is simply forecasting their margins and working capital before being paid.

In terms of outstanding payments and the average amount of days to pay, the data reveal interesting findings on contractors. The total amount of outstanding payments owed to all contractors that failed was £378m in the year 2019, which may be deemed to be double their turnover. The average amount of days taken by contractors to pay suppliers or subcontractors is revealed by the data to be the same average payment period published by Build UK (2019). It takes approximately 55 days for a contractor to pay a subcontractor from the day of receiving invoices from their clients.

In terms of liquidity and debt ratio analysis, the findings reveal that contractors had lower liquidity with a mean score of 0.78 and a high debt score of 2.0. This illustrates that the majority of contractors had more liabilities than assets. Practically, this is the most common cause of financial difficulties and inabilities to sustain cash pressures for most construction companies.

Moreover, as each company affirmed different reasons for going into administration such as project delays, shortages in bank borrowings, onerous contracts, underbidding of tenders and diversification of business; the most commonly cited cause for all of them was lack of cash. The reason was that most contractors had to quickly fund their project costs i.e. site mobilisation, paying labourers and suppliers, purchasing material etc. and using their turnover for potentially 35 to 75 days. Therefore if payments were not received or promptly paid by clients, contractors would experience cashflow problems leading to insolvency risks. For instance, when the ground engineering specialist contractor Aspin fell into administration in July 2019, the £40 m turnover firm blamed cashflows problems, delayed payments of £800,000 from Carillion's collapse and shortages in bank lending as reasons for its collapse (Price 2019). Overall, the real industry data analysed in this chapter reveals that late payments to contractors significantly causes cashflow problems for many businesses in the sector.

Table 6.5: Summary of Construction Clients: nature, turnover, outstanding payments, business performance, average number of days for payment and causes for going out of business (sourced from Construction News 2020)

No	Company	Age of Compan y	Nature of Clients	Turnover over 5years	Operating profit over 5 years	Current ratio over 5 years	Outstanding payments (in millions)	Debt ratio	Average number of days to pay	Reasons for going out of Business
1	Vaughan Engineering Limited	46	Tier 2&3 clients	211630849	2.05	0.88	800,000	1.06	65	Stagnation in cashflow, delayed payments, Carillion effect
2	Pochin	50	Tier 3 clients	706856285	2.87	0.82	70,000,000	1.21	59	Legacy issues of contracts, Underbidding of tenders, cashflow pressures
3	Aspin	30	Tier 3&4 clients	104011020	2.16	0.42	17,000,000	1.33	60	Problematic contracts, Carillion effect, cashflow problems
4	Lincoln-Simons Group	75	Tier 3&4 clients	603403	1.69	0.76	78,000,000	1.12	50	Economic uncertainties (Brexit), contract delays, unsustainable cashflow
5	Hawk Plant	40	Tier 3&4 clients	166782111	2.26	0.83	28,000,000	1.18	60	Difficulties in trading conditions, loss of major clients, problematic contracts, delayed projects, cashflow pressures
6	Paragon Interiors	34	Tier 3 clients	19927629	4.43	0.59	17,000,000	1.3	55	Cashflow problems, business diversification
7	Lakesmere Group	29	Tier 3 clients	70804877	1.97	0.88	26,000,000	0.93	46	Business diversification, stagnation of cashflow, difficulties in trading conditions
9	Marcus Worthington and Company	43	Tier 3 clients	31474212	0.84	0.86	23000000	2.59	55	Economic uncertainty, Fall in bank borrowings, Cashflow pressures
10	Shaylor Group	10	Tier 2&3 clients	87931	4.10	0.69	34030000	1.18	90	Cashflow problems late payments, bad debts, tightening of credit terms
11	Bardsley Construction	55	Tier 3 clients	316604817	2.44	0.99	45000000	1.72	39	Delays in payments, Contractual disputes, Economic uncertainties, Political environment, time overruns, Resources shortage, under- pricing of projects, Cash issues
12	Sanderson Contracts	20	Tier 3&4 clients	217031270	1.18	0.86	36700000	0.66	50	Cashflow pressures, bad debts
13	UKD Groundworks & Civil Engineering	20	Tier 3 clients	155741172	1.84	0.81	3000000	0.93	48	Cashflow pressures, bad debts, unpaid retentions, difficulties in trading conditions, loss of major clients
14	Clugston	90	Tier 4 clients	2638657	0.46	0.44	2638407	8.82	45	Delayed payments, contractual disputes, under-pricing of projects, cash flow problems
15	Blackbourne Limited	49	Tier 4 clients	20527128	-1.20	1.76	3873444	2.79	40	Cashflow problems, late payments, bad debts, tightening of credit terms
[Total			166796298	2.32	0.78	378,530,000	2	54.4	; ;

6.4 Results and Findings

The findings from this chapter provide meaningful, real insights concerning unfair payment practices in terms of payment durations, the nature of clients and their impact on business failure. The payment performance data from tables 6.4.1, 6.4.2, 6.4.3 and 6.4.4 show that tier 1 clients pay their contractors promptly, with an average of 90% of invoices settled within 30 days and 84% of invoices issued within 5 days from the date of receipt of invoice. Tier 2 and 3 clients were discovered to be notorious in paying their supply chains late, although the findings illustrated that Tier 3 had been the worst payers among the three levels of construction clients.

In terms of business performance, the statistical analyses reveal a somewhat weak relationship between unfair payments and contractor performance. The results in figure 6.4.5 (a), (b), (c) and (d) affirm a weak monotonic relationship between a delayed time frame for honoring invoices and a contractors' average operating profit margin, return of capital employed, liquidity and debt ratios (p = 0.04, p = 0.09, p = 0.09 and p = 0.01). The implication of these findings is that there is a relationship between delays in honoring invoices and a construction contractor's financial performance, although the relationship is weak.

6.5 Summary

In summary this chapter has looked at archival data and case studies of various construction companies and their financial performance gathered from different sources and surveys. The data presents the payment performance of three tiers of construction clients in the UK; specifically the average time for honouring invoices, the percentage of payments made within 60 days and the percentage of invoices not paid within the agreed terms (late payments). In addition, business performance for the tiers of construction clients were obtained to ascertain whether unfair payment practices affected business performance in the construction industry. Case studies were also discussed about companies that had faced financial distress as a result of unfair payments.
CHAPTER 7

Qualitative Data Analysis - Semi-Structured Interviews

7.1 Introduction

This chapter presents the results of the qualitative data inquiry for an in-depth understanding of the research problem. Firstly, the chapter examines the profile of the study participants and the sample representativeness of the interviewees. A brief description of the interview data and the structure of the interviews are subsequently analysed. After the interview data had been transcribed, coded and analysed, the views of the study participants were then presented under five subheadings in line with the research objectives. For a better understanding of the research problem, interviewees' perceptions were sought regarding an overview of unfair payment practices and their consequences for the construction industry: clients' interests/attitudes to unfair payment issues; and an overview of measures to enhance payment practices in the construction industry.

7.2 Overview of the participants' profile and the Population Sample

The study participants were selected using stratified sampling methods. The choice of study participants was influenced by their background, years of experience, professional qualifications, scope of works and their understanding of the research problem. Table 7.2 illustrates the profile and background of the study participants. The study's population sample consisted of seasoned quantity surveyors, managing directors of subcontracting companies, senior project managers, commercial lawyers, project/cost consultants, construction clients, procurement officers in government departments and commercial and contract managers. A total of 19 participants took part in the study interviews. A breakdown of the study shows that 45% were quantity surveyors, 27% subcontractors, 15% construction project managers, 8% main construction clients and 5 % government representative bodies. In addition, 70% of the study participants were qualified professionals with over 15 years' experience in various disciplines.

Table 7.2:	Profile and	number	of participants	interviewed

S / No.	Profession	Nature of work	Years of experience	Professional qualification
1	Project manager- Main contractor	Building works- small to large	12	MCIOB
2	Managing director- Subcontractor	Mechanical and electrical	15	MCIOB
3	Chairman government representative body	Industry representative engaged with government	20	MCIOB
4	Managing director – Subcontractor	Flooring and Screeding	8	None
5	Project manager – Private client	Building and Property-small to large	5	MCIOB
6	Managing director - Subcontractor	Mechanical and Electrical	40	None
7	Quantity surveyor– Main contractor	Building and property	8	MRICS
8	Quantity surveyor – Subcontractor	Building and civil engineering	10	None
9	Managing director – Subcontractor	Building and property works-small to medium	9	None
10	Quantity surveyor – Main contractor	Building and property	14	MRICS
11	Quantity surveyor – Main contractor	Building and civil engineering	18	MRCIS, MCIOB
12	Quantity surveyor - Client	Building and property	6	MRCIS, MCIOB
13	Project manager - Main contractor	Building and civil engineering	38	MRICS
14	Quantity surveyor – Subcontractor	Building and civil engineering	10	None
15	Quantity surveyor – Main contractor	Building and property	15	MRCIS, MCIOB
16	Managing director – Contract manager and specialist subcontractor	Construction and engineering	28	CIARB
17	Quantity surveyor – Main contractor	Building and property	20	MRICS
18	Contract Manager - Subcontractor	Partitions and Fittings	33	MCIOB
19	Private client	Building and civil engineering	10	Other

7.3 Brief description of the interview data and analysis

Sections 4.7.2.2 and 4.7.2.4 explain how the study's interview data was obtained and analysed.

7.3.1 Validity of the findings

To uphold the validity of the study's qualitative findings, five areas were specifically addressed: the selection of the participants' profile for the study; the preparation of the interview questions; the processing of the interview data; the presentation of the interview data; and the interpretation of the findings.

7.4 Qualitative findings

This section presents the study participants' views in subheadings identified in line with the set objectives. The headings range from general perceptions of unfair payment issues in the construction industry, the robustness of standards forms of contracts and other regulative measures designed to curb unfair payment practices, the scale of late payments between tiers of clients/contractors and the impacts of late payment practices on business performance. Below are some key excerpts trimmed from the interview data.

7.4.1 General views of study participants concerning unfair payment practice

The study participants were asked to express their views about unfair payment practices in the construction industry. The responses provided led to a wide range or array of perceptions concerning unfair payment practices in the industry. For a better understanding of the research questions, some excerpts are highlighted as follows:

"...construction industry is doomed with unfair payment practices especially late payment and unreleased retentions ... such that for the past 38 years I haven't seen any company, regardless of its profile, size or value that gets paid on time nor receives their full retentions because of fat cats (paymasters) always wanting to withhold money in their accounts till the last minute..." (Senior Project Manager- Main contractor)

"... the industry is known for a corporate culture that allows main contractors (like what happened with Carillion Plc's directors), to cash cows (milk or squeeze) their subcontractors or small suppliers' payments so as to boost their working capital and portray a good picture of their company.." (Chairman of government representative body BEIS)

"...Systemic abuse of cash retentions has left SMEs in debt and going out of business... each week we hear both large and small construction companies going bust with absurd reasons, in reality a lack of regular payments is the problem..." (Project Manager-Contractor)

The study participants were probed further regarding the scale and state of unfair payment practices within the construction sector. Some specific questions asked include the following: "you said unfair payment practices is a chronic problem in the construction sectors; can you help me with specific examples in the supply chain where unfair payment practices are endemic? Besides, who bears the brunt of unfair payment practices the most within the construction supply chain?" The study

participants provided wide ranging responses to the questions asked. Some of quotes are highlighted as follows for better understanding:

"... I think unfair practices is a huge issue for small companies or subcontractors because they rely heavily on regular cash flow to survive compared to large construction companies that have huge working capital, better assets and other business..." (Project Manager – Main contractor)

"...overall, late payment problems is the most common issue in the construction sector ... the scale of the problem often depends on your position in the supply chain... from my view, subcontractors and suppliers are more vulnerable to late payment problems normally because of unfair contracts and clients' commercial domineering power..." (Project Manager – Private client)

However, there were dissenting voices among some study participants regarding the scale of unfair payment practices in the construction industry. Some study participants were of the view that unfair payments were not a problem in construction, perhaps because it is the nature of the industry itself. Below are some excerpts regarding their views to the problem:

"...unfair payment practices such as late payment is not that common in construction ... personally I work with clients and they pay us on time ...though we may probably be late with a couple of days paying subcontractors..." (Senior Quantity Surveyor – Main contractor)

"...late payment certainly exists but it is not a problem to us because we have some kind of mechanisms in place that makes us work closely with the entire supply chain to mitigate potential risks like payments and others..." (Project Manager – Main contractor)

Generically, the majority of study participants were of the view that unfair payment practices, particularly late payments, were common in the construction industry. Some study participants perceived unfair payment practices as a genuine and deep-rooted problem characterised by industry culture and unethical practices in the construction sector.

7.4.2 Perceptions of study participants to unfair payment issues in the construction industry

The views of the study participants were sought concerning client perceptions of unfair payment practices. Interviewees were asked to express their views regarding the context and nature as to why clients seemed to be apathetic about unfair payment issues. Other questions asked included: "Do you think clients are sympathetic to unfair payments issues in the construction industry?" The participants' views of the question varied significantly with some descriptions relating to types of clients, intuitions, mechanisms, self-interests and so forth. Some of the excerpts are quoted verbatim for a better understanding of the responses from the study participants:

"...Clients have always been sympathetic, and they have taken action against it... but from my experience private clients are more proactive in paying as they are not constrained compared to public clients where money is heavily scrutinised and bureaucracy is longer..." (Quantity Surveyor – Main contractor)

"...From an operation standpoint, clients have never been sympathetic to late payments ... because they do not see companies struggling ... they just want their projects to be completed. Private clients are more sympathetic because they're investing their money, while public clients see it more as a job..." (Quantity Surveyor – Subcontractor)

"...Six years ago, I worked with a main contractor who are not proactive or interested at paying their supply chain on time. but they're best at banking money into their accounts and not paying other suppliers until some are sort of screaming for it..."(Quantity Surveyor – Subcontractor)

Obviously, the above views suggest that client's interests in unfair payment issues are polarised due to the fact that clients can be apathetic about paying on time; the type of client, the process in place, blame culture, a lack of trust and the fragmented nature of the construction industry with its potential to deflect clients' interests.

In addition, most Tier 2 construction clients (main contractors) were accused of using their bargaining powers as main industry players to push their use of some payment mechanisms such as early payment schemes and supply chain finance in order to boost their cashflow and profit margins. Thus, it could be argued that clients' interests concerning unfair payment issues are polarised around bargaining power, cashflow issues and a lack of business goodwill.

7.4.3 Views of study participants on clients' indifference to unfair payment practices

Furthermore, opinions were sought from participants if they believed that clients were indifferent to unfair payment practices in the construction industry. Some textual excerpts mentioned by participants are highlighted as follows:

"I would like to say private clients are often deliberate in withholding payments due to the profit incentivised nature of business... if you have more money in the bank you get more interest compared to paying it out while public clients have more processes that tend to be difficult in paying on time..." (Project Manager – Main contractor)

"...In most projects I have worked, private clients offer short term payment while public offer longer terms and most are not flexible..."(Quantity Surveyor - Main contractor)

"...Private clients are quick and committed (more) in paying till the end of a project while public are less driven because it's not their project...I think client indifference is influenced by their nature..." (Managing Director – Small flooring and screeding company)

Generically, the views of the study participants portray that clients are indifferent to payment practices in the construction industry. Public clients are considered to be lax towards timely payments due to bureaucracy and the accountability of public funds, whilest private clients are seen as proactive and deliberate regarding payments for reasons of financial risk aversion. Moreover, the construction market has recently been dominated by private sector construction activities that account for high levels of unfair payment practices in the industry.

7.4.4 Views of participants on the robustness of payment practices within various standard forms of contracts (SFoCs) and other legislative measures

The views of participants concerning the robustness of various standard forms of contracts and legislative measures were sought out in order to have a better understanding of research objective one. A substantial number of participants were unanimous that the use of payment terms and practices stipulated in SFoCs and other legislative measures to curb unfair payments were ineffectual in the industry. Some of the textual excerpts by study participants are highlighted as follows:

"... In fact, there's hesitance among subcontractors and even main ones in maintaining amicably working relationships than using the inputted 'aggressive measure' towards the problem. Overall, I would think it's only good as a piece of paper in reality..." (Quantity Surveyor – Subcontractor)

"...To say, most of the contracts and legislation consist of loopholes that give flexibility to payments such that clients and main contractors use or manipulate payment clauses as long as they want..." (Project Manager-Main contractor company)

"...Contractors like Carillion and many others have abused and manipulated clauses and regulations designed to curb unfair payment practice...thereby forcing subbies to accept longer terms and even charging interest to get their payment earlier..." (Chairman of Government Representative body BEIS)

"...most of the contracts are bespoke (**back to back**), I have never signed a pure standard form of construction contract . . .it will be their price and terms of payments and I have no choice but to abide since I want the work as well.." (Managing Director -Subcontractor)

7.4.5 View of participants on the scale of unfair payment practices between different tiers of construction clients

In addition, the study participants were asked to describe the scale of unfair payment practices between different tiers of contractors. Some of the excerpts proffered by the participants are highlighted as follows:

"...Definitely agree that late payments vary between different tiers of contractors but from my experience they're all affected by the problems..." (Quantity Surveyor – Main contractor)

"...Based on the scale of preference, contractors and suppliers in the lower positions of the supply chain are mostly affected by unfair payment practices..." (Quantity Surveyor - Public client)

"...the scale of the problem is enormous in construction... as principal contractors are notorious in withholding payments ... in fact they always pay below the invoices and keep retentions up to 5 years..." (Quantity Surveyor – Subcontractor)

"...In reality, many subcontractors are on longer payment terms with their main contractors or clients and shorter with suppliers or vendors... so most are prone to all kinds of unfair practices ..." (Quantity Surveyor -Maincontractor) The quotes above clearly show that unfair payment practices are prevalent among Tier 2 clients and reasonable to Tier 3 clients and other members down the supply chain. Perhaps it is partly driven by industry notions and a business culture that allows unfair payment practices to be insignificant for upper tier construction clients but a major problem for lower tiers.

7.4.6 Participants views on unfair payment practices effects on a contractor's business performance

The study participants' views were sought on whether unfair payment practices influenced a contractor's business performance. Significant responses and insights were received into various aspects or approaches used to measure business performance. The majority of study participants seemed to indicate that short-term financial indicators such as profits, turnover, operating margins, share prices and return of capital employed were significantly affected due to late or non-payments by the client. On the other hand, some participants had different notions about other non-financial measures that were of significance to businesses, for instance loss of competitive advantage, productivity, morale, credit ratings, administrative burdens, failure to pay salaries and annual bonuses, loss of staff retention, health and safety issues, poor quality of work and loss of future investment, etc. Some of the textual excerpts uttered by participants are highlighted as follows:

"...Mostly, unfair payment affects contractor cashflow and profits since payments are retained by the QS or client...so that has a direct impact on those two and not so much longer, a business will collapse..." (Project Manager- Main contractor)

"...Profit foremost, though indirectly it has other consequences, loss in competitive advantage, increase in administrative costs as you need to allocate people and resources to chase unpaid invoices, losing the best deals, morale, productivity, progress of work on site, materials or quality issues..."(Managing Director - Subcontractor)

"...From my experience, unfair payments have a negative effect on human costs. I know of a company director that had to use all his savings to pay staff and suppliers that resulted in a heavy burden, loss of investments and stress from chasing payments owed to a client..." (Project Manager – Contractor)

"Apart from financial consequences on business, unfair payments have a profound effect on one's life...I faced a lot of stress, panic and sleepless nights because payments were not coming from main contractors while I had to pay workers and suppliers on site...." (Managing Director - Subcontractor)

7.4.7 Participants' views regarding the role of the client-contractor relationship on unfair payment practices in the UK construction industry

In addition, the study participants were asked to describe the role of the clientcontractor relationship in unfair payment practices. Some excerpts proffered by participants are highlighted as follows:

"...Most clients, especially public, pay us late. But even when they do so we don't send some sort of letters or try to pull workers or plants off site since we still want to maintain a good working relationship..." (Quantity Surveyor -Main contractor)

"...Definitely most subbies wouldn't like to speak ill of contractors or clients, since we still want to keep the business running and have a good working relationship... we sort of help each other as much as you could ..." (Managing Director -Subcontractor)

".... I would say if a client is late to pay, I wouldn't pick a call or send any sort of letter, for it's not conducive for working relationships. I have to look after them more than they look after me. They've got the work and they know you need it. You need to give and take or even accept some of their terms as your main source of income..." (Subcontractor)

Overall, from the above views by study participants, the role of the client-contractor relationship plays a significant factor in unfair payment practices as it is desirous for most parties to prioritise their relationship at all cost.

The reason is that the industry's "give and take" philosophy towards client-contractor relationships plays a huge role; as a significant proportion of contractors find it difficult to query the issue and be firm with their clients, because in doing so, they could jeopardise both their relationships and future work as well. As mentioned by one interviewee, they would never use any sort of measures or actions against their clients as they still want to maintain a good relationship in the long run. Again, this "give and take" attitude may go somewhere towards explaining a client's indifference to unfair payment practices because most clients, including main contractors, have a strong supposition that subcontractors and other lower tiers would not complain because of their commercial bargaining power in securing and tendering for major contracts in the industry.

7.5 Results and findings

Overall, the findings from qualitative data have highlighted interestingly mixed views concerning unfair payment issues, clients' indifference to unfair payments, the robustness of SFoCs and other legislative measures, the scale of unfair payments on different tiers of construction clients, the role of the client-contractor relationship and the effects of unfair payments on a contractor's business performance.

Most findings have confirmed that unfair payments, particularly late payments, are part of corporate culture and a trademark of the construction industry. For example, most interviewees have affirmed late payment as a prevailing culture that it is acceptable for clients and main contractors because of the industry's hierarchical structure and their bargaining power.

Moreover, when the participants were asked about client perceptions and attitudes to unfair practices, their views seemed to be mixed and polarised, as it may not be in the client's best interests to deal with such issues. As for the robustness of SFoCs and other legislative measures in readdressing unfair payment practices, the findings reveal that most measures are ineffective and less stringent for parties to adhere to. This is because of the modification of contract clauses and forms that often disadvantage the lower tiers (i.e. the use of adhesion contracts and unfair terms) and the ability to utilise them is impaired by the client-contractor relationship. In terms of client indifference to unfair payment practices, public clients seemed to be more lax due to the bureaucratic accountability of public funds, while private clients were more proactive towards timely payments. These evidences however are contrary to the majority of unfair payments practices wherein the problem is centred on Tier 2 and 3 clients, most of whom are private.

On the other hand, the qualitative findings gave an interesting perspective on the effects of unfair payment practices on a contractor's business performance that would not have been captured by the quantitative data, as seen in 5.8.5. For example, the interview data highlighted other factors that directly influence a contractor's business performance such as reputation, loss of competitive advantage, productivity, credit ratings, administrative burdens, failure to pay salaries and annual bonuses, the loss of staff retention, health and safety issues, poor quality of work and loss of future investments, etc. These effects are deemed not to be quantifiable although they have

significant consequences for business performance and continuity. While the key measures of industry performance remain financial outcomes such as profits, turnover and return of capital employed, there is a growing concern about further impacts on the individuals running these businesses. Similarly, the notion can be viewed with the current situation of UK construction companies failing whilst causing ripple effects for other individuals.

Overall, one of the key deductions from the views highlighted by the study participants is the role of the client-contractor relationship in unfair payment practices. Although the majority of participants gave mixed views on the questions asked, the one foremost thing that could perhaps be argued was that most contractors including subcontractors and suppliers would not jeopardise or damage any sort of relationship established with a client as a result of unfair payment practices.

7.6 Summary

In summary, this chapter has presented qualitative analyses from the views of experienced industry practitioners on the research problem. The views of these industry practitioners have been presented in line with the research objectives covered in Chapter 1. Additionally, findings from this chapter will be confirmed and compared to the views obtained from the preceding chapters.

The general view from the qualitative findings indicate that unfair payment practices are problematic to the tiers of construction clients. Regardless of existing contractual measures, bargaining power, the nature of the client and their position in the supply chain, the problem is deemed to be widespread. The majority of views expressed by industry practitioners are of the opinion that it is trivial for clients if an invoice is paid late, nor are any sort of unfair practices inputted, given the fact that there are apathetic interests in timely payments. Additionally there is a strong supposition that client-contractor relationships and the desire to protect them have intensified the late payment culture in construction.

CHAPTER 8

Discussion

8.1 Introduction

This chapter discusses the findings from the quantitative, qualitative and archival data analyses regarding the study. The main aim of this study is to critically evaluate the findings from the data analysed with the intention to develop a theoretical framework that will help minimise unfair payment practices in the UK construction industry in the next chapter. Firstly, this chapter discusses findings obtained from the aforementioned analyses in line with the research objectives. Further reference is made within these discussions to the literature reviewed in chapters two and three. The discussions are held in five sub-headings in relation to the research objectives set out in Chapter 1. Also, the chapter looks at the research's contributions to the body of knowledge as well as the implications of the study.

8.2 Research Findings

A common inference from the study's literature is that unfair payment practices are deep-rooted issues in the construction industry. Perhaps, the issue of unfair payment practices is not peculiar to the construction industry alone. For example, a recent report published by BEIS (2018) revealed that cosmetic and departmental stores (often referred to as high street companies such as WH Smith, Boots, Holland and Barret, etc.) have standard payment terms of 90 days and beyond plus other unfavourable practices that reflect payment practices in the construction industry.

Other distinctive features identified by the study's literature that exacerbate unfair payment practices include the fragmented and hierarchical structure common to construction supply chains. Many authors believe that fragmentation and a lengthy supply chain restrict the speed of prompt payments, which has adverse impacts on small and medium-sized businesses, individuals, families and the economy at large. For instance, 3 in 10 small contractors struggle to pay salaries and business taxes due to unfair payments (Prior 2020).

Moreover, entry barriers to the construction business were also identified as a major contributor to unfair payment practices. Indeed, financially challenged construction businesses are often allowed to operate freely in a highly competitive environment. Thus, such companies with less financial cushions(or shock absorber) are more likely to run into fdifficulties. Without doubt, a robust cash flow is essential for a construction business's survival. Moreover, cash is closely related to payments that serve as the primary source of working capital required for contractors to successfully deliver their projects and attain profits.

Another overarching finding from this study is that payments to construction contractors are like fuel that energise their businesses. Cashflow moreover is the most significant indicator of a construction business's performance. Apart from the study's literature review, the archival data analysis in table 6.5 illustrates that dwindling payments and cashflow pressures were the main reasons for a construction business's collapse. Moreover, the prevailing industry culture and disparity of bargaining position in the supply chain is often blamed for the deep-rooted unfair payment practices in construction. This factor was also highlighted in the qualitative inquiry of the study; where participants claimed that unfair payment is a trademark of the construction industry's corporate culture. For example a study participant interviewed in London claimed that "...overall, late payment is the most common issue in the construction sector ... the scale of the problem often depends on the contractor's position in the supply chain... subcontractors and suppliers are more vulnerable to late payments normally because of unfair contracts and clients' commercial domineering power..."

For a thorough understanding, the study sought to evaluate the robustness of payment procedures, the scale of unfair payment practices, contributing factors, the effects of unfair payments on a contractor's business performance and the role of the client - contractor relationship in unfair payment practices. The following subheading will discuss the study's findings based on the set objectives and themes stated above.

8.2.1 Robustness of payment procedures in various standard forms of contracts

Findings from study's literature suggested that one of the primary objectives of SFoC's is to establish a contractual relationship that includes prompt payment from the client

to the contractor. For example, current standard forms of contract commonly used in the UK include JCT 2016, NEC 4 and FIDIC. All have payment terms or provisions designed to facilitate timely payments, the right for interim payments, payless notices, withholding notices and non-payments provisions. These standard forms of contract clearly recommend interim payments, a time frame for honouring invoices (payments), the issuing of non-payments notices and interest on late payments, all designed to boost a contractor's cash flow. Findings from this study revealed that payment clauses in SFoCs are often disregarded or ignored, with clauses usually amended to encourage unfair payments to contractors. Yet, most aggrieved parties (mainly subcontractors) do not treat unfair practices as a fundamental breach of contract; though in most SFoCs it is considered a clear breach of contract.

The literature review also suggested that unfair payment practices in construction usually defy regulations, charters and codes such as HGCRA 1996, LDEDCA 2009, the prompt payment code and fair payment charter designed to curb them. Yet there is hardly a known case in law that has brought punitive measures against clients that indulge in unfair payment practices. Some study participants also suggested that contractors, subcontractors and suppliers also have their fair share of blame regarding the prevalence of unfair payment practices. This is because they often accept unethical payment arrangement such as "pay when paid", and "pay if paid" clauses in their contracts, therefore relegating fair payment regimes and the adjudication of payment related disputes to the background. This view is supported by Sharkey's et al. (2014) assertion that "often SFoCs in construction are amended to move risk to parties that have weaker bargaining power; thus, alluding to contracts of adhesion (take it or leave it) that are usually common between the main contractor acting as client to the subcontractor or supplier".

From a theoretical perspective, unfair payment practices in the construction industry can be linked to principal-agent and relational contracting theories. The Principal Agent theory as explained in Chapter 3 above can best be described as an agency relationship where the principal (clients) engages another party, the agent (or contractor), to perform a service on his or her behalf which involves the delegation of some decision-making authority to the agent (Turner 2004; Winch 2010; and Eriksson 2008). The problem that is likely to arise from such a relationship is that both 'principal' and 'agent' are seeking to maximise their own opportunity in a contractual

arrangement. Thus, there is good reason to believe that both the principal and agent will not act in the best interests of all. In such a situation, there is a tendency for the

principal (the main contractor acting as the client) not to pay his or her subcontractors and suppliers on time to boost their cashflow, in as much as the agent fails to treat the incident of unfair payment practice as a fundamental breach.

Similarly, the findings from this study are closely linked to relational contract theory put forward by Macaulay (1963). The theory stresses that generally, parties to a contract do not practice the complete contract when they perceive that their counterpart is an essential part of their business. Macaulay (1963) argued that in relational contracts, parties re-settle the contract in a direction to sustain the continuity of the business rather than terminating the contract. Perhaps this is an essential argument as to why the aggrieved contractual party (usually a subcontractor) engages in unfair payment practices with other parties in the construction industry, with neither seeking legal action or readdress. MacNeil (1986), in a review of relational contract theory, believed that co-operation is the most crucial characteristic of any contract and that one of the five primary components of contracts include "co-operation; economic exchange; planning for future; potential external sanction; and social control and administration". Moreover, MacNeil (1986) believed that most contracts were incomplete, which would give a party the opportunity to look for "loopholes" that would allow them to exploit the contract. These loopholes, as discussed by numerous authors, include late payments to contractors, disparity in rates, "pay when paid", withholding moneys due to subcontractors to enhance the main contractor's cash flow, etc.

Moreover, the literature review of unfair payment practices suggested that the nature of construction contracts, industry practices, competition, the client-contractor relationship are issues that often allow such loopholes to be perceived as normal practices. Kumaraswamy et al. (2008) stressed that relational contract theory can be used as a guide to select contracts and minimise adverse payment practices. Fundamentally, the key lesson that can thus far be learnt from both the study's findings and the theoretical review is that a contract cannot be practiced to its fullest for many reasons such as relationship considerations and its incompleteness, therefore allowing for loopholes usually common to unfair payment practices. For example, it is often a common practice in the construction industry for subcontractors to accept

longer payment terms ranging from a minimum of 45 days and a maximum of 120 days payments terms. Compelling contractors to accept longer payment terms and

many other unfair payment practices is in strict terms, a pure act of dishonesty and a fundamental breach of contract in the eye of the law.

This view is largely supported by the findings from the qualitative inquiry where most professional interviewees opined that the robustness of payment provisions within SFoCs was often impaired by legal loopholes, payer's attitudes, current payment processes, the use of adhesion contracts, vulnerability within the supply chain and client-contractor relationships. For example, one of the study participants claimed that "…most contracts and legislation have loopholes that give flexibility to payments …. Hence clients and main contractors manipulate them and withhold payments as long as they want …"

The study's quantitative inquiry did not differ from the viewpoint that SFoCs were generally ineffective in dealing with payment issues. For example, 82% of study participants strongly agreed that SFoCs payment mechanisms did not protect contractors against unfair payments. These findings suggest that SFoCs are ineffective due to many factors.

The findings also reveal that the majority of study participants that completed the questionnaire believed standard forms and payment legislations were "good as a piece of paper" because they do not address the deep-rooted unfair payment culture in the construction industry. However, the majority of clients, contractors and subcontractors still favour the use of SFoCs. Therefore the aim of this study is to design a comprehensive and sustainable payment framework that will minimise the shortcomings in current payment predicaments in the construction industry.

8.2.2 The Magnitude of Unfair Payment Practices

Objective two seeks to determine the magnitude or scale of unfair payment practices in the construction industry. To achieve this objective, the study measured the following variables: type and nature of unfair payments practices and average duration for contractors to receive payments in the UK construction industry. The study used the Late Payment Act 1998 provision, Section 4 subsection (7A) as a benchmark to measure objectives two: "anything that is a gross deviation from good commercial

practice and contrary to good faith and fair dealing; the nature of the goods or services in question and whether the purchaser has any objective reason to deviate from the

stipulated/contractual payment term". Findings from the study suggest that the majority of unfair payment practices in the literature focus on delayed or late payments as a particular problem synonymous with contractors. However, the reality is that unfair payment practices in the construction industry are not only a contractor's problem, but a wider commercial anomaly that significantly distorts the cash flow of the industry's entire supply chain.

For example, findings from the study's quantitative analysis revealed that the scale of unfair payment practices is enormous and cuts across the industry. The problem is worse in the lower levels of the construction supply chain. Key findings from the study's quantitative inquiry ranked late payment as the foremost unfair payment practice among all tiers of construction clients. Unpaid retentions and "pay when paid" were found to be prevalent practices in construction, as illustrated in figure 5.7.2.2. Moreover, the findings also revealed that there is a catalogue of hitherto-unmentioned unfair payment practices that are growing exponentially in the UK construction industry. The most prevalent of these expedient unfair payment practices include: "pay a fee to be retained in a client's supply chain list", "lower interests on delayed payments", "spurious deductions", "discounts on retention release", "supply chain bullying" and "elongated payment duration terms". Unfortunately, these unfair commercial practices are very common among main construction contractors acting as clients to subcontractors and other suppliers. There is a need to check these unwholesome practices that have significant implications for contractors' and suppliers' cash flow.

It is practically impossible to discuss unfair payment practices without critically examining the nature of unfair commercial practices among different tiers of clients. The study's documentary, quantitative and qualitative inquiries show that Tier 2 (main contractors acting as clients) are the worst perpetrators of unfair payment practices. This finding aligns with most literature in that Tier 2 clients subvert good commercial practices to enhance their business cash flow strategies. This viewpoint is supported by a recent EPR report (2019) that found "though bad payment habits are improving in many businesses; tier 2 clients seem to be far off the limelight of becoming early

payers". Hence, the proposed framework development in figure 9.1 focuses on the Tier 2 payment structure.

Furthermore, findings from the archival data analysis revealed that tier 2 clients have both poor payment performance and payment terms, with an average of 46 days payment terms, 25% of invoices not paid within the agreed terms and 90% invoices paid within 60 days. In comparison, Tier 1 clients are considered to be prompt payers, with an average of 96% of invoices settled in 30 days and 85% of invoices issued within 5 days from the date of receipt. Meanwhile, tier 3 clients are arguably considered to be the worst payers with an average of 50 days' payment terms, 26% of invoices not paid within agreed terms and 74% of invoices paid within 60 days. However, other findings revealed that tier 3 clients' have longer payment terms compared to tier 2, perhaps due to the cascade effect of delay from the upper tiers to the lower tiers within the supply chan.

A clear deduction from the study in terms of the interval for honouring invoices is that payment durations in construction remain far from the standard forms of contract's stipulated 28 days for payment. Also, findings from the quantitative and documentary analyses in terms of payment duration are closely aligned to data published by EPR (2019) that revealed the average payment duration for most construction businesses ranged from 31 to 50 days and from 51 to 90 days. Comparatively, the data also revealed that other sectors such as retail, automobile and manufacturing have better payments terms and durations for honouring invoices.

8.2.3 Contemporary Causes of Unfair Payment Practices

The review of the literature regarding causes and effects of unfair payment practices have not changed considerably in the last two decades. For example, Hughes et al. (1998) argued that causes of unfair payments are hinged on two prominent circumstance: "I cannot pay" (ability to pay) and "I would not pay" (willingness to pay) attitudes of clients. The categorisation put forward by Hughes et al. (1998) clearly explained that "I cannot pay" refers to a client's financial difficulties or constraints faced during the project's execution. The latter meanwhile is purely the client's attitude of being unwilling to pay within the agreed payment terms. For a better understanding of

the contemporary causes of unfair payment practices, the study identified 23 causes that spanned from Hughes' et al. (1998) argument. The causes of unfair payment practices were categorised into five main groups, namely cultural, industrial, technical, regulatory and others causes. Table 5.7.3.2 illustrates the five top contemporary causes of unfair payments. These causes include the use of a contractors/supplier's money to boost a client's cash flow (often referred to as cash flow strategy), the prevailing culture, the client-contractor relationship, pay 'if/or when' paid and the industry norm of 'work first get paid later'. Although the study's participants ranked these causes based on the above groupings, the majority agreed that these five causes were the most significantly contributing causes of unfair payments in modern construction businesses.

The use of cash flow strategy was ranked as the most prominent and root cause, as it received the highest mean score of 3.59 from the 23 causative factors identified by the study. Remarkably, the same factor "cash flow strategy" was also identified in the study's qualitative inquiry as the most common root cause of unfair payment practices in the construction industry. Table 6.5 showed that most of the study's participants were of the view that it is common practice for the main contractor and even clients to use unfair payments as a business strategy to boost their cash flow and financial performance.

In reality this is often common in most construction businesses that struggle financially, particularly in an industry such as construction with low working capital. Hence, most construction businesses (to a great extent main contractors) adopt measures to boost their cash flow such as inducing longer payment terms, deferring payments, withholding payments, supply chain finance, unpaid retention money, etc. Indeed, the use of unfair payment practices as a business strategy is purely an issue of "I will not pay".

Other contemporary causes identified to encourage unfair payment practices in the construction industry include the industry norm of "work first get paid later" and the client-contractor relationship. It is fair to assert that the client-contractor relationship is usually an underlying cause of unfair payment practices because most clients are aware that the business entities to whom they are indebted are most unlikely to challenge their payment prejudice behaviour because "you cannot bite the hand that feed you".

Generically, it is fair to argue that most unfair payment practices in modern construction history are largely attributed to the "would not pay" (willingness to pay) attitude of clients. Therefore, the findings from this study clearly indicate that there is a need for an alternative payment method if unfair payment practices are to be minimised in the construction industry.

8.2.4 Unfair payment practices influence a contractor's business performance

One of the key objectives of the study is to establish the impact of unfair payment practices on construction business performance. Marr and Schiuma (2003) argued that the phrase 'business performance' lacks cohesive meaning because of the vagueness of specific metrics and indicators used to evaluate business performance. In the commercial sense, business performance is measured using an array of financial metrics such as return capital employed, operating profit, debt ratio, liquidity ratio and asset utilisation ratio, etc. However, it is important to note that business performance metrics vary from one industry to another, for example cash flow and profit margin in the construction sector are often different from other industries such as oil and gas, communications, etc.

In specific terms, the study analysed four financial ratios, namely operating profit margin, return of capital employed, liquidity and debt ratio for a better understanding of the relationship between unfair payment practices and the financial performance of construction businesses. The four financial ratios were chosen because they can be used to predict business continuity, cash flow and insolvency. This is imperative because in reality, most construction businesses operate on a thin profit margin and a dwindling cash flow. Therefore, it is important to determine the impact of unfair payment practices on construction business performance.

The findings in tables 6.3.1 and 6.4.6 presented an average of five years' operating profit margin and return on capital employed for small construction contractors to be 5.71% and18.18%, respectively. Similar metrics for large construction companies were calculated and yielded a value of 3.34% and 25.73%. The disparity noed in profit margins between small and large construction companies is to be expected, because businesses in the construction sector are generally synonymous with low profit margins. Besides, this finding is in line with a recent FTI consulting report (2017) titled *"Is the Construction Industry Ripe for a New Operating Model"*. The report claims that

196

company turnovers for smaller businesses will likely constitute higher margins compared to larger ones. Similarly, the MHA report (2018) stressed that higher profit margins of 4.5% to12% were observed in smaller main contractors compared to larger construction companies, with an average profit margin hovering between 2.5% to 7.5%

Conversely, liquidity and debt ratio analyses for both large and small-sized construction companies were calculated as illustrated in table 6.3.1 and 6.4.6. The computation revealed a marginal ratio significantly less than 1.0. Arguably, construction is considered to have low liquidity and a high debt ratio because of undercapitalisation. A situation exists where businesses cannot acquire necessary funds or operate within negative working capitals with limited access to external funds. In construction, dependence on regular payments for work done is a crucial aspect in meeting a company's short-term obligations and optimising performance.

The study's quantitative inquiry showed that there is a weak relationship between the percentage agreements of the study participants' perceptions and a company's average operating profit margin, return of capital employed, liquidity and debt ratios with p-values of p = 0.01, p = 0.06, p = 0.02 and p = 0.06. Results from figures 5.8.5 (a), (b), (c) and (d) illustrated a weak relationship between the study participant's percentage agreements and a contractor's business performance. Results from figures 6.4.5a (a), (b), (c) and (d) showed a weak relationship between the time of honoring small construction contractors' invoices and business performance (.i.e. operating profit margin, return of capital employed, liquidity and debt ratios) with the following p-values: p = 0.21, p = 0.16, p = -0.03 and p = 0.04. There is a similar relationship to the figures for large construction companies that also showed a weak relationship, as illustrated in figures 6.4.6 (a), (b), (c) and (d)

A possible explanation for the weak relationships is that a contractor's business performance is often determined by various factors such as nature of business, nature of clients, business turnover, operating environment, etc. Moreover, there is a need to bear in mind that the figures used in the correlation analysis were based on the study participant's percentage agreements with a possible margin for errors, although large data was also considered, However, figure 6.4.6 (d) presented a slightly better relationship between time frame for honoring contractors' invoices and large debt

ratios with a p-value of 0.48. The analysis indicated that there is a somewhat statistical relationship between unfair payment practices and a contractor's debt ratio. Indeed, there is reason to believe that in most cases, not paying contractors regularly has a negative tendency towards debt profile.

Comparatively, the views of the study participants in the qualitative inquiry signified that unfair payment practices negatively impacted construction businesses. Interactions and other discussions with participants during the interviews showed that unfair payments affected a construction business's cash flow, a company's debt profile and working capital, etc. Moreover, some participants provided vivid accounts of how unfair payment practices in the industry had affected their businesses. Specific consequences of unfair payment practices identified by the interviewees include poor business reputation, failure to pay salaries/or overheads, loss of staff, low productivity, staff morale, failure to pay rent/mortgages, loss of credit rating, substitution of quality over price, cancellation of apprenticeship programmes and individual wellbeing. For emphasis, one interviewee claimed that ".... usually I would say it affects contractors' turnover and profits but as well as credibility, failure to pay salaries and annual bonuses, staff retention, distorts future investment into the business and tarnishes relationships with other businesses as well ...".

Overall, it is evident from the study's findings from both the quantitative and qualitative data inquiries, that unfair payment practices have both financial and non-financial impacts on business performance. However, the problem remains deep rooted in the construction industry. Therefore, this study is proposing a framework that seeks to minimise the destructive practices of unfair payment issues in the UK construction industry.

8.2.5 Role of the Client-Contractor Relationship

The study data clearly identified the client-contractor relationship as a major factor that influences deep-rooted unfair practices in the construction industry. The industry is known for both long and short-term business relationships across the supply chain. Perhaps this is due to a traditional and collaborative working culture that is often driven by a sense of securing regular contracts. Eom et al. (2015) opined that client-contractor relationships in construction were usually cordial; but with the client having a better bargaining power over the contractor. In many instances, the contractor

always does the bidding of the client. Indeed, such client-contractor relationships normally have a significant effect on the contracting parties, leading to unfair practices such as compelling the weaker party in the contract to agree to longer payment terms, non-payment of invoices and discounts for prompt payments, etc.

Findings from both the guantitative and gualitative analyses indicate the role of the client-contractor relationship significantly influences unfair payment practices in the construction industry. Results from the questionnaires and the survey show that 83% of the study's participants agreed that consideration of the client-contractor relationship would influence their decision not to initiate legal action against their client, even if there was a clear case of breach of contract. Arguably, most clients in the industry know that their supply chain cannot 'bite the hand that feeds them'. Thus, some categories of clients take advantage of subcontractors and suppliers by perpetrating unfair payment practices intentionally, because they are aware that maintaining continuous relationships is of utmost importance to many contractors and suppliers. In fact, 97% of subcontractors/suppliers often find it difficult to challenge their payment terms or practices because doing so would jeopardise their relationships and risk securing future work. For example, a study interviewee claimed that "... if a good client delays my payments for months I will not complain too much; though with adverse effects on our business; ... because we cherish continuous relationships with my clients ...in business clients are king ... we dance to their tune ..."

It is obvious from the findings, that many subcontractors and suppliers' main desire is to protect their client-contractor relationships at all costs; even to the detriment of their profits. Therefore, it is fair to argue that both contractors and suppliers are both culpable of unfair payment misdemeanours because of their behavioural attitudes and acceptance of illicit business practices.

In a nutshell, the bulk of unfair payment practice issues in the construction industry lie with the disparity in commercial power that most clients have over contractors and suppliers. This view lends credence to the principal-agent theory discussed in the preceding section in this chapter. The reality remains that the clients (principals) have a better position to negotiate, dictate and impose contractual terms on the contractors (agents). Without doubt, such commercial bargaining power usually leads to a domineering authority over contract payment terms and durations. It is apparent that

the coercive commercial bargaining powers of clients are likely to continue in the foreseeable future in construction industry; despite alternative payment mechanisms and legal regulations with the sole aim of limiting domineering clients' transactional power over contractors and suppliers. Therefore, in the succeeding chapter, the study will put forward the development of a framework that has the potential to minimize unfair payment practices in the construction industry.

CHAPTER 9

Development of a Conceptual Framework for Enhancing Payment Practices

9.1 Introduction

The aim of the study is to develop a conceptual framework for enhancing payment practices, particularly between main contractors to subcontractors and others in the supply chain. As previously discussed, the cascade payment obligation, also known as the hierarchical contractual framework, gives Tier 2 clients (main contractors) a dominant bargaining power over their supply chain. Payment in construction is usually dictated by the client or consultant team with a cascade effect of the main contractor in control of payments to subcontractors and suppliers. The procedure is that the main contractor is reimbursed for the completed works on site and in turn, the main contractor will pay their subcontractor and supplier in the project.

A common deduction from the study findings revealed that a lengthy hierarchical payment structure together with a "work first and get paid later" culture in the construction industry places subcontractors and others in the supply chain at risk of unfair payment practices such as delayed payments, "pay when paid", disparities in rates, cowboy bullying practices, non-payment of retentions, etc. The financial reality for construction businesses remains that most contractors and others in the supply chain operate on thin profit margins, low working capital and higher debts. Hence, subjecting these categories of construction businesses to unavoidable unfair payment practices makes them susceptible to insolvency risks. Indeed, construction businesses, whether large or small, are highly dependent regarding cash flow to survive. Thus, based on the findings from the study's literature review and the quantitative, qualitative and archival data inquiries, the study has put forward the development of a framework to enhance fair and prompt payments based on the concept of ring-fencing payments to contractual parties that are more vulnerable to unfair payment practices in the construction supply chain.

9.1 Functionality of the proposed framework

The structure of the proposed framework is designed to enhance fair payment practices as illustrated in figure 9.1. The framework is based on three cardinal principles, namely: (i) ring-fenced project bank accounts; (ii) an independent payment regulator designated the Special Purpose Payment System Regulator (SPPSR); (iii) periodic publication of main contractors' and other clients' payment performance. For clarity, the framework is different from existing project bank accounts (PBAs) because the framework have mechanisms to internally regulate payments (i.e. the SPPSR); minimise the domineering commercial power of Tier 2 clients and above all, the SPPSR acts as trustee or guarantor for prompt payments to subcontractors and others in the supply chain. Another advantage of the proposed framework over existing PBAs is that the proposed SPPSR will govern, manage and provide oversight in terms of distribution and prompt payment of money due to requisite contractual parties. The framework is designed to ensure that the funds deposited in the ring-fenced accounts are secured, protected and free from insolvency risks. The purpose of SPPSR is to regulate the dictatorial and domineering commercial power of Tier 2 clients (main contractors) over their suppliers that are vulnerable to unfair payment practices.

The proposed SPPSR are to be appointed by contractual parties. There must be an independent regulator likened to an arbitrator or adjudicator overseeing agreements on current construction contracting requirements. The overarching purpose of the proposed framework is to sustain the transparency of information and good payment practices for many contractors and suppliers at the receiving end in the construction supply chain. Apart from the proposed framework, the study recommends that the contractual parties should set up electronic payments such as BACs or CHAPs linked to their bank accounts, that would enable the government and the proposed SPSSR to track and monitor any unwholesome payment actions.

The traditional power of Tier 2 clients (main contractors) to vet and approve invoices submitted by subcontractors and suppliers is not removed by the proposed framework. Rather, once payment applications are submitted by subcontractors or suppliers, the main contractor audits and certifies all the interim valuations, then notifies the SPPSR to release payments due to the contracting parties.

The onus of depositing monies (or amounts due) into a ring-fenced account lies squarely on the clients, but with the consent of main contractor. The notification will contain details of the amount to be paid, retentions that need to be deducted and any pay less notices. The independent regulator will update the account with information given, then the SPPSR will disburse the payments fortnightly (every two weeks) and simultaneously to contractors and suppliers. If the parties receive less from the interim (payment) application, the monies will be held in the account and then paid in subsequent interim payment certificates. Similarly, any retention monies deducted will be held in the same account to reduce insolvency risks to the main contractor and clients. It is imperative to note that the recent UK government Department for Business, Energy and Industry Strategy (2017) claimed that an estimated £4.5 billion of subcontractors' money were held as retentions from 2015 to 2018 but were never released. Therefore, the proposed framework will help to regulate both interim (monthly) and retention payment processes if implemented properly as illustrated in figure 9.1 below.

9.3 Benefits of the proposed framework

Arguably, the benefits of the proposed framework cannot be overemphasised because it has the potentials to enhance certainty, consistency and security of payments to contractors and suppliers that are at the forefront of executing 93% of construction works on sites.

Another benefit of the framework is that it offers protection against insolvency risks in the supply chain, as payments and retentions can still be made to members of the projectteam. For instance, where the contractor or client fails to remit payments and becomes insolvent, the retained balance in the ring-fenced account would be used for other parties to the contract. This will help to reduce disputes and costs associated with latepayments such as interest payments on credit, administrative charges and time lost in chasing payments.

Moreover, with the help of the framework, parties will focus on the project more efficiently and with better value service delivery that would encourage collaboration and innovation in the industry. The Government and major private clients are targeted as key drivers to use the framework to encourage other categories of clients and contractors for its optimal use. The initiative is not compulsory to contracting parties; but the suggestion is that those that sign up for the programme will be listed under the UK's "Excellent Construction Contracting Scheme" a proposed programme that will allow major contractors and listed subcontractors to remain in the good book with the UK government. Other positive is that it has potentials to open door for prospective and lucrative government contracts for those that signup for the programme. Comparatively, the proposed framework is similar to the not- for-profit Tenancy Deposit Scheme (TDS) managed by National Housing Association and regulated by the Housing Act 2004. The TDS is a scheme that safeguards tenancydeposits from disputes between landlords/agents and tenants. Failure to comply withTDS results in severe penalties or fines issued by the Housing Act 2004.

For thorough scrutiny the propose framework was presented and validated by industry practitioners, academics and other stakeholders. In addition, the framework was presented for review at the Association of Researchers in Construction Management (ARCOM) 2019 and CIC 2020 construction conferences (see attached references for proceeding from these conferences). Below is figure 9.1 the initial conceptual framework that was developed and presented to construction professionals for validation.



Figure 9.1: Proposed Conceptual Framework for enhancing Payment Practices in the UK construction Industry

<u>Key:</u>

SPSSR- Special Purpose Payment System Regulator

Tier 1 Client - Public / private

Tier 2 Client – Main contractors

Summary of the framework:

- The SPPSR is a necessary independent body, whose key roles are to monitor and facilitate payments, rather than payments being cascaded by the main contractor to the subcontractors and the supply chain.
- The process flow for payment applications from the submission of invoices by the subcontractors/suppliers down to payment is shown by the red arrow.
- All submitted invoices for work performed shall be electronic; Tier 2 clients MUST confirm their receipt, approval or rejection within seven days.
- When Tier 2 clients approve subcontractors' invoices, the SPPSR will disburse payment to subcontractors and suppliers within two weeks.
- SPPSR is designed to be legally and financial backed by a jurisdiction or Act of Parliament required to oversee and secure prompt payments for the framework to function optimally.

9.4 Validation of the proposed framework

In addition to the use of conferences to solicit ideas about the proposed framework; the researcher designed set of questionnaire and interview questions to test practicability and validity of the proposed framework using member process checking. Lincoln and Guba (1986) described member process checking as a crucial technique for establishing the credibility of a research process. Both formal and informal member process checking was employed, thus theframework was sent to study participants and other professionals working in theconstruction industry.

During the interview, the propose conceptual framework and brief notes about its functionality was presented to various participants. They were asked to provide their views and opinions regarding the proposed framework in terms of certainty, promptness, and potentiality to enhance fair payment practices in the UK construction industry. The participants that took part in the second set of interviews consist of three quantity surveyors, two project managers, one contract manager, one managing director of a small-sized construction firm and a Professor of Construction, Economics and Management. Below are some key excerpts from the study participants' concerning their views of the proposed framework:

"...In principle this is a good idea, but I am concern with implementation in terms of clients' willingness to give up some control of cashflow. Likewise, the same practice by main contractors who critically need cash to meet suppliers and labour' demands..." (Contract Manager)

"...The framework is a brilliant idea ... but the process will require robust managing and there will be administration costs involved in ensuring that the framework requirements are met. This may increase the overall project preliminary costs in terms of the financial management of the framework ..." (Quantity Surveyor-Main contractor)

"...In the UK we have seen many schemes such as the fair payment charter and initiatives widely ignored by the industry, ... though the framework presents a potential solution taking lessons from other sectors... particularly where the issue of unethical practices have been observed ... your framework is an excellent starting point in terms of solution to the issue. However, if this is voluntary like the other payment schemes, I don't think main contractors (the major culprit) will sign up ... we need tit for tat rule (force) perhaps via primary legislation. ... for your proposal to be effective" (Project Manager-Publicclient) ".... The proposed framework is clear and has potential to resolve the current payment issue in the construction industry including return of cash retentions problem to contractors ... Ultimately, any scheme that has potentials to improve the cash flow is a positive development..." (Managing Director- Small-sized constructioncompany)

"...The issue will be enforcement, as most subcontractors are ignorant of their contractual rights, and most are afraid to exercise their rights for fear of being called for future work. adjudication costs and relationship issues with the "big boys" ... propose changes to current payment systems requires government intervention such as ombudsman for quick access payment cases resolutions..." (Quantity Surveyor-Subcontractor)

"...From what you explained so far and based on the proposed framework, two key aspect were addressed: domineering, balance power and managing the cashflow for the supply chain (suppliers and subcontractors) ... While it is clear that the power of the main contractor is needed to approve the work invoices, controlling the cash is key and sometimes essential in order to achieve project objectives..." (Professor of Construction, Economics and Management)

A clear deduction from the validation exercise (see Appendix G, interviews and contribution from two construction conferences) is that the proposed framework has potential to enhance payment practices in the construction industry if it supported or back by law or an act of parliament. However, some participants expressed concerns about the implementation of the framework regarding possible resistance and lack of enforcement my main contractors who are the major culprit of unfair payment practices.

The researcher scrutinised all feedbacks from construction professionals in the validation exercises and were deemed to constructive and valid. Afterward, the proposed framework presented in 9.1 was modified slightly to include the use of statutory or an Act of Parliament for compliance and monitoring as illustrated in figure 9.2.



Figure 9.2: Modified Conceptual Framework for enhancing Payment Practices in the UK construction Industry

Key:

SPSSR- Special Purpose Payment System Regulator Tier 1 Client – Public / private Tier 2 Client – Main contractors

Summary of the framework:

- The SPPSR is a necessary independent body, whose key roles are to monitor and facilitate payments, rather than payments being cascaded by the maincontractor to the subcontractors and the supply chain.
- The process flow for payment applications from the submission of invoices by the subcontractors/suppliers down to payment is shown by the red arrow.
- All submitted invoices for work performed shall be electronic; Tier 2 clients MUST confirm their receipt, approval or rejection within seven days.
- When Tier 2 clients approve subcontractors' invoices, the SPPSR will disburse payment to subcontractors and suppliers within two weeks.
- SPPSR is designed to be legally and financial backed by a jurisdiction or Act of Parliament required to oversee and secure prompt payments for the frameworkto function optimally.
- The proposed framework is similar to the not-for-profit Tenancy Deposit Scheme (TDS) managed by the National Housing Association and regulated by theHousing Act 2004. A detailed explanation of the framework is presented in chapter 9 of the study.

9.5 Administration of the proposed framework

Based on the feedback received, some participants raised the issue of administrative costs for the establishment and operation of the framework, because most parties may not be eager to incur additional expenses of running the proposed framework. Thus, the study suggests that to remove such cost burden on contractual parties, the interest generated from holding the monies in a ring-fenced account could be used as operating costs for the proposed framework.

For instance, under the custodial Tenancy Deposit Scheme (TDS), the interests earned from the deposit are used to administer the scheme for landlords and agents that have themselves signed up to it. The scheme stipulates that parties will decide and agree on the interest rate at the start of the tenancy period (i.e. monthly or yearly) with the bank. For example, the scheme states that 1% interest would be paid from 1st January 2018 to 31 December 2018. This measure is to ensure that such a provision would reduce the financial burden and stress incurred to TDS administers. Hence, a similar approach could be used in the framework, i.e. a situation where a ring-fenced project account is set up to allow interest charges depending on the financial institution used.

9.6 Limitations of the framework

The key limitation to the framework is how to persuade contracting parties to sign up to it. The onus is particularly on convincing Tier 2 clients (main contractors) who are the main culprits of chronic unfair payment practices. This rhetorical question is important because current construction business models allow main contractors to hold on to subcontractors' and suppliers' monies to boost their balance sheets.

Therefore, to overcome this limitation, the proposed framework suggests the inclusion of incentives such as "preferred main contractors list" and "semi-structured framework agreement" that will support or provide a contract guarantee to the main contractors. Perhaps the incentive suggestions could be piloted by the Cabinet Office, Infrastructure and Project Authority to encourage main contractors to sign up to the proposed framework. Moreover, a main contractors' rating system such as gold, silver and bronze for excellent contracting payment practices could also be introduced for prioritised contractors selected for public procurement processes by the UK

Government's Cabinet Office.

Another limitation of the proposed framework is that it overrules privity of contract and the associated data management regulations. As previously discussed in chapter 2, contracts are usually drafted with a single contractual relationship with one of the participants (main contractors) but not with all the parties in the supply chain. For instance, the current provisions prevent subcontractors from seeking direct payments from clients, even when the main contractor defaults and likewise, if the contractor adheres to good practices and makes prompt payments. The framework outweighs this limitation by taking control of the contracts with stringent legal backing that will allow payments to be directly paid to subcontractors or suppliers.

The proposed scheme is designed without the inclusion of a dispute resolution procedure within the framework. In as much as payments and other disputes are included in the existing Alternative Dispute Resolution (ADR) mechanisms, most are deemed to be complex, time consuming and are often not pursued due to the client-contractor relationship. Moreover, the appropriate use of the framework for the size and detail of a project is limited for smaller contracts that are usually carried by sole traders (i.e. sub-subcontractors and suppliers). The specific criteria for the framework are therefore determined to be high value.

CHAPTER 10

Conclusion and Recommendations

10.1 Conclusion

This chapter presents the key findings and conclusions to the study with an emphasis on how the research aim and objectives have been achieved. The main aim of the study is to develop a framework that will enhance fair payment practices in the UK construction industry. In doing so, the study established five objectives that provided a guideline for the researcher to develop a sustainable framework with the potential to enhance fair payment practice issues in the UK construction industry. In addition, the chapter also discusses limitations and suggests sustainable recommendations for the study problem.

For reference purposes, the first objective of the study was to evaluate the robustness of payment procedures in various standard forms of contracts (SFoCs) used in the construction industry. A common deduction from the study's findings and theories regarding this objective is that payment provisions in SFoCs are often modified and disregarded by parties to contracts. The study discovered that amendments to SFoCs allow clients to embed poor payment practices as a result of principal-agent and relational theories between contractor, subcontractor and the entire supply chain. Findings from both the quantitative and qualitative analyses indicate that the majority of the study's participants agreed that payment provisions in SFoCs do not help to alleviate unfair payment practices due to many factors such as payer attitudes, current payment processes, vulnerability within the supply chain and the client-contractor relationship.

The second objective was to determine the scale of unfair payment practices between tiers of construction clients. Findings from the study's archival data, qualitative and quantitative analyses revealed that unfair payments practices are an endemic problem and distort the cash flow of the entire supply chain in the industry. Late payment remains a common practice for all tiers of construction clients followed by retentions and "pay when paid" and a range of other practices. However, the quantitative findings revealed that unfair payment practices were predominant, with main construction contractors acting as clients to subcontractors and other suppliers. For instance, paying a fee to be retained in a client's supply chain list, lower interest on delayed payments, spurious deductions, discounts on retention release, supply chain bullying and elongated payment duration terms were discovered to be very common among Tier 2 clients.

Furthermore, findings from the archival analyses indicated that Tier 2 clients (main contractors) were the main drivers for poor payment performance and terms, with an average of 47 days and 86% invoices paid within 60 days. For example, the experience of Carillion Plc's demise and the practices of many others, show that the routine withholding of invoices and extending of payment terms for their supply chains are common commercial strategies employed by this category of client to boost their weak balance sheets and profit margins. Also, the findings showed that tier 3 clients are worse payers due to the cascade effect of the upper tiers in delaying invoices or not paying others on time. All in all, the study revealed that the construction industry has a reputation for longer payment terms with an average of 31 to 50 days and 51 to 90 days, respectively.

The third objective was to identify contemporary causes of unfair payment practices in the UK construction industry. The study identified 23 contemporary causes of unfair payment practices and grouped them into five main categories, namely cultural, industrial, technical, regulatory and other causes. The findings concerning this objective revealed that most of study's participants agreed some clients used unfair payment practices as a commercial (cashflow) strategy to squeeze their suppliers to boost their balances and financial performance.

Certainly, this strategy has opened a line of credit for many businesses to compete and enter the construction sector, with other bigger companies adopting similar practices. Other causes that were indicated by the study's participants includes the industry norm 'work first get paid later', client-contractor relationship and the use of adhesion contracts. Generically, it is fair to argue that the use of cashflow strategy
is spearheaded by the unwilling attitudes of clients and underpinning power of client/contractor supplychain relationship.

The fourth objective was to ascertain if unfair payments influenced a contractor's business performance. As previously discussed in Chapter 2, measuring business performance is a vague concept given the list of indicators or metrics used to evaluate the performance of a business or industry. However, the study adopted traditional accounting practices based on the use of financial ratios to measure a contractor's business performance.

Findings from the quantitative inquiry revealed a weak relationship between perceived unfair payment practices and business performance. The correlation between the study participants' agreements and companies' average operating profit margins, return of capital employed, liquidity and debt ratios illustrate a weak relationship; with p-values of p = 0.01, p = 0.06, p = 0.02 and p = 0.06. However, the correlation analysis based on archival data somewhat revealed a relationship between perceived unfair payment practices and a contractor's debt ratio significantly at p=0.48. In reality, there is a tendency for a contractor's debt to increase because of late or unfair payments, spurring the need to borrow money to survive and continue trading.

On the other hand, the study qualitative inquiry reveals that there is a snowball effect of unfair payment practices on construction businesses. Interviewees claim that unfair payment have direct consequences on contractor's profit margins, cashflow, working capital and other indirect effects such as failure to pay salaries/overheads, low productivity, loss of credit rating, stress, reputational damage, failure to pay salaries/overheads, substitution of quality over price, etc.

The fifth objective was to evaluate the role of the client-contractor relationship regarding unfair payment practices in the UK construction industry. Findings from both the quantitative and qualitative analyses showed that most study participants agreed that the client-contractor relationship significantly influenced unfair payments issues in projects. The majority of study participants affirmed that maintaining good working relationships with clients, main contractors, subcontractors or suppliers often compelled them to forgo or accept poor practices that favours their clients, particularly

Tier 2 clients (main contractors). This is mainly due to the fear of damaging commercial relationships and risks to securing future work. Findings regarding this objective lend credence to the principal-agent theory discussed in Chapter 3 of the literature review, suggesting clients (principal) have better bargaining power over their agents (sub/contractors), thus using their commercial power to perpetuate unfair payments.

In summary, the key findings from the study suggest that the bulk of unfair payment practices in the construction industry lie with Tier 2 clients (main contractors) and that there is an overwhelming perception that if unfair payment practices were curtailed within this category of clients, the entire problem would be significantly minimised. Therefore, the study puts forward a framework that could significantly enhance fair payment practices between main contractors, their subcontractors and others in the supply chain. As discussed in Chapter 9, the framework has the potential to curb the domineering commercial powers of Tier 2 clients by securing payments into a project account that would be controlled by a Special Purpose Payment System Regulator (SPSSR).

10.2 Limitations

The study is limited in several ways. For example, the study collected data about unfair payment practice from clients (public and private), consultants, main contractors and subcontractors working in the UK construction industry. However, extra efforts were made by the researcher to obtain lower tier's views (particularly sub-subcontractors, suppliers, vendors, and so forth), but was not successful due to their periodic visits on sites and difficulties in getting responses from them. Nonetheless the study considered subcontractors' views as representative of the lower tiers' opinions and perspectives on unfair payments practices.

Moreover, the companies' financial information obtained from Companies House UK was limited to 34 out 179 construction companies that were considered for the study. The difficulty was that most participants did not have their companies' accounts and profit and loss statements filed in Companies House and made available on their website. There was a similar experience during the archival analyses, whereby the

researcher obtained financial data from 62 companies including both large and medium-sized contractors. Given that there are more than 300,000 construction businesses in the UK, not all of the information was available. Moreover, the majority did not want to take part in the study due to their company's reputation and discretion over payment practices. However, the combination of companies' data from both the questionnaire and the interviews proved to be sufficient to carry out the statistical analysis regarding unfair payments and a contractors' business performance.

Another limitation of the study stemmed from the fact that a severe financial crisis and uncertainties such as Brexit and COVID-19 were experienced during the research period that significantly impacted most companies' finances. Moreover, physical validation was not possible because of Governmental restrictions on movement due to COVID-19. Therefore, due to time factors, the validation of the framework was carried out through sending postal questionnaires and emails to participants who were familiar with the study's theme. This proved to be an effective method in validating the framework and the overall findings of the research.

Moreover, it was observed during the qualitative and quantitative data collection that some participants were not sincere about their company's payment terms and true financial position. However, the research overcame this barrier by conducting cross-checking references with archival data on company payment performance and practices. This was observed during the interviews when the researcher asked about payment terms. Some participants were being economical with the truth, as they did not want to tarnish the image of their company in public or to divulge the real nature of unfair payments in the industry.

10.3 Recommendations

Based on the study's findings and overall conclusions, this research puts forward the following recommendations:

 There is a need to encourage the use of the proposed framework as a means of addressing unfair payment practices between Tier 2 clients, subcontractors and other tiers in the supply chain.

- The study recommends that there should be enough education and training programmes in contract matters for stakeholders and practitioners, to enable them to fully understand and administer different obligations and rights that may arise under payment sections in contract forms or legislation used.
- Moreover, there is the need to design a comprehensive database for contractors that capture payment practices in the UK in term of inability ("I cannot") and unwillingness ("I would not") to pay. This database will be useful to understand the underlying nature of a company's payment and cashflow position. This would be helpful for other businesses to negotiate better terms in forming contracts.
- Government and practitioners in the industry need to introduce proactive measures to strengthen existing legislations and provisions to promote fair payment practices in the construction industry. For instance, enforcement to non-compliance, automatic interest for late payers, a company's tax incentives and restricted access to public sector contracts, etc.
- There should be historical payment data of 'naming and shaming' that would enable clients and stakeholders understand a contractor's payment performance.
- The researcher recommends that further studies should be carried out to evaluate the role of the client-contractor relationship regarding unfair payment practices in the UK construction.

REFERENCES

- Abdul-Rahman, A., Munaaim, M., Danuri, M., and Berawi, M. (2008) 'Issues Affecting Payments in the Building and Construction Industry of a Rapidly Developing Economy'. *Building Engineer*, 4-10.
- Abdul-Rahman,H.,Kho,M. and Wang, C. 2014) Late payment and nonpayment encountered by contracting firms in a fast-developing economy. *Journal of Professional Issues in Engineering Education and Practice*, 140(2),
- Abdul-Rahman, H., Takim, R. and Wong, S.M. (2009) 'Financial Related Causes Contributing to Project Delay'. *Journal of Retail and Leisure Property* 8,225-238.
- ABFA (2016) 'Construction Companies Hit Hardest by Growing Waits for Payment. Delays Increase by 22% in just a Year' *Abfa News* 2016 [online] available from <u>https://www.financialdirector.co.uk/2016/08/23/construction-companies-hit-hardest-by-growing-waits-for-payment/</u> [20 August 2017]
- ABFA News (2014) 'Local Councils Lag Far Behind Government Guidance on Supplier Payment'. *ABFA News* [online] available from <<u>http://www.abfa.org.uk/news/81/Local-councils-lag-far-behind-Government-guidance-on-supplier-payment</u>> [20 August 2017]
- Abowitz, D. A. and Toole, T. M. (2009) 'Mixed Method Research: Fundamental Issues of Design, Validity, and Reliability in Construction Research'. *Journal of Construction Engineering and Management* 136 (1),108-116.
- ACCA (2015) 'Ending late payment.Part1:Taking Stock available from <<u>https://www.accaglobal.com/content/dam/acca/global/PDF-technical/small-business/pol-tp-elp-</u> <u>3reflections.pdf</u>> [20 August 2017]
- Ali,N., Ali, N., and Ameer,A (2005) 'Construction Industry Payment and Adjudication Act: Reducing Payment-Default and Increasing Dispute Resolution Efficiency in Construction'. International Forum Construction Industry Payment Act and Adjudication.
- Akinsiku,O.E.and Ajayi,O.M.,(2016).Effects of delayed payment of contractors on construction project delivery in Nigeria. The Construction, Building and Real Estate Research Conference of the Royal Institution of Chartered Surveyors.
- Amaratunga, D., Baldry, D., Sarshar, M. and Newton, R., (2002) 'Quantitative and qualitative research in the built environment: application of "mixed" research approach'. Work study. Vol.51 No.1 pp.17-31.
- Ameer, N. A. N. (ed.) (2006) Construction Industry Payment and Adjudication Act, Reducing Payment Default and Increasing Dispute Resolution Efficiency in '*International Forum on Construction Industry Payment Act and Adjudication*' held 13&14 at Kuala Lumpur Convention Centre. Kuala Lumpur, Malaysia.
- Amoako,K B(2011) The effects of delayed payment on cashflow forecasting of Ghanian road contractors. (Master's Thesis) Kwame Nkrumah University of Science and Technology,Krust
- Ansah. K. (2011) 'Causes and Effects of Delayed Payments by Clients on Construction Projects in Ghana'. *Journal of Construction Project Management and Innovation*,1(1),27-45.

- Arewa, A.O. and Farrell, P. (2015) 'The culture of construction organisation: the epitome of institutionalised corruption'. *Construction Economic and Buildings* 15(3), 5-71.
- Arocena,P. and Nú nez, I.(2010) 'An empirical analysis of the effectiveness of occupational health and safety management systems in SMEs'. *International Small Business Journal* 28(4),398-419.
- Atrill, P. and McLaney, E.J., (2008) *Accounting and finance for non-specialists*. Pearson Education Limited Publication.
- Ashworth, A. (2006) *Contractual Procedures in the Construction Industry.* 5th ed. Essex, Pearson Education.
- Asset Based Finance Association (2016) *Late Payment: An analysis by sector* [online] available from<u>https://www.abfa.org.uk/news/92/ Late-Payment:-An-analysis-by-sector</u> [28 August 2017]
- Australian Procurement and Construction Council (1996) National Action on Security of Payment in the Construction Industry[online] . available from <u>http://www.apcc.gov.au/ALLAPCC/APCC%20PUB_National%20Action%20on%20Security%20</u> <u>of%20Payment%20in%20the%20Construction%20Indudustry%201996.pdf</u> [20 July 2018]
- Ayudha,B.I.N.(2012) *Factors causing delay in payment of Residential projects in Thailand* in FIG Working Week 2012, Knowing to manage the territory, protect the environment, evaluate the cultural heritage Rome Italy,6-10,1-11.
- Azman,M.N., Dzulkalnine, N., Hamid, Z. A., Kamar, K. A. M., and Nawi,M.N.M (2013) *Payment Scenario in the Malaysian Construction Industry Prior to CIPAA*, in Law Dispute Resolution Conference 56.
- BACS (2019) 'UK SMES face debt Burden of £23.4billion'. Press Release [online] November. available from <u>https://www.bacs.co.uk/NewsCentre/Press</u> [22 December 2019]
- Banwell, H (1964) *The placing and management of contracts for building and civil engineering works.* HMSO,London.
- Baum,R.,J.Frese,M.,Baron,R.A and Katz,J.A.,(2007) 'Entrepreneurship as an area of psychology study: An introduction'. *The Psychology of Entrepreneurship1*.18

Barber, D (2012) Bayesian Reasoning and Machine Learning. Cambridge: Cambridge University Press

Blaike, N (2000) Desigining Social Research, 2nd ed. Polity, Cambridge.

- BBC news (2013) 'Cameron to Consult on Late Payment to Small *Business*'. BBC [online] 14 October available from <u>https://www.bbc.co.uk/news/business-24517594</u> [20 January 2018]
- Betts, M. and Lansley, P. (1993) 'A Review of the First Ten Years'. *Construction Management* and Economics 11 (4), 221-245.
- BERR (2008a) SME Statistics for the UK and Regions 2007. Enterprise Directorate Analytical Unit. Department for Business Enterprise and Regulatory Reform (BERR).

- Bogdan, R. C. and Biklen, S. K. (1998) *Qualitative Research in education: An Introduction to theories and methods.* 3rd ed. Boston,MA:Allyn and Bacon.
- Bon, R. (2000) *Economic Structure and Maturity: Collected Papers in Input-Output Modelling and Applications*: Ashgate Pub Limited.
- Bon, R., Birgonul, T.and Ozdogan, I. (1999) 'An Input-Output Analysis of the Turkish Construction Sector, 1973-1990: A Note'. *Construction Management & Economics* 17 (5), 543-551.
- Boon, J and Prigg, C. (2012). Evolution of quantity surveying practice in the use of BIM–the New Zealand experience. In *Proceedings of the CIB International Conference on Management and Innovation for a Sustainable Built Environment* (pp. 84-98).
- Brealey,R.A.,Myers,S.C and Marcus,A.J (2010) *Fundamentals of Corporate Finance*. 6th ed Newyork:McGraw Hill.
- Briggs, A. R., Coleman, M., and Morrison, M. (2012) *Research Methods in Educational Leadership & Management*. Sage Publications.
- British Woodworking Federation (2018) *Retention Payments in the Construction Industry*. [online] 19 January available from <u>https://www.bwf.org.uk/wp-content/uploads/british-woodworking-federation-response-to-consultation-on-retention-payments-in-the-construction-industry.pdf</u> [17 March 2019]
- Briscoe,G.,Dainty,A.R and Millet,S.,(2001)'Construction supply chain partnership:skills, knowledge and attitudinal requirements'. *European Journal of Purchasing and Supply Management* 7(4),243-255.
- Brook, M. (2016) *Estimating and Tendering for Construction Work*. 5th ed. Taylor and Francis.
- Bell,E., and Bryman,A., (2007) 'The ethics of management research: an exploratory content analysis'. *British Journal Management* 18(1) 63-77.
- Bryman, A. and Bell, E. (2007) Business research methods. Oxford: Oxford University press.
- Bryman, A. and Bell, E. (2015) Business Research Methods. Oxford: University Press.
- Burell, G. and Morgan, G.,(2012) Sociological paradigms and organisational analysis: Elements of the sociology of corporate life. Routledge.
- Bresnen, M. and Marshall, N., 2000. Building partnerships: case studies of clientcontractor collaboration in the UK construction industry. *Construction management and economics*, *18*(7), pp.819-832.
- Brewer, J.D and Miller, R.L. (2003) *The AZ of social research: a dictionary of key social science research concepts*. Sage Publication.
- Burgess,T.F. (2001) A general introduction to design of questionnaires for survey research. University of Leeds.
- Burrell, G., & Morgan, G(1979) Sociological Paradigms and Organizational Analysis, Heinemann.
- Cabinet Office (2012a) Government Construction, A Guide to the Implementation of Project Bank Accounts (PBAs) in Construction for Government Clients: Cabinet Office

Cabinet Office (2011) Government Construction Strategy 2011. London: HM Cabinet Office

Carmichael, D.G. (2002). Disputes and international projects. CRC Press.

- Chalker,M. and Loosemore, M., (2016) 'Trust and productivity in Australian construction projects: a subcontractor perspective. Engineering'. Journal of *Construction and Architectural Management* 23(2),192-210.
- Chappell, D., Powell-Smith, V., Marshall, D., and Cavender, S. (2008): *Building Contract Dictionary*. John Wiley and Sons.
- Cavanagh,S.(1997)'Content analysis: concepts, methods and applications. Nurse Researcher 4,5-16.

Chan, P.C., Chan, W.M., Fan, L.C.N., Lam, T.I. and Yeung, J.F.Y. (2004). A comparative study of project partnering practices in Hong Kong. In *CII-HK Conference* (pp. 65-75).

- Chen, J. J. (1998) 'The Characteristics and Current Status of China's Construction Industry'. *Construction Management and Economics* 16 (6), 711-719.
- Cherry, M (2018) Carillion's small suppliers must be paid. [online] 15 January available from <u>https://www.fsb.org.uk/resources-page/carillion-s-small-suppliers-must-be-paid.html</u> [22 February 2018
- Chilisa, B. and Kawulich, B. (2012) Selecting a Research Approach: Paradigm, Methodology and Methods'. *Doing Social Research, A Global Context.* London: McGraw Hill.
- Chiang, Y. (2009) Subcontracting and its ramifications: A survey of the building industry in Hong Kong. International Journal of Project Management, 27(1), pp.80-88.
- CIC (2020) "Unfair Payment Issues in Construction: Re-thinking Alternative Payment Method for Tier -1 Contractor to Subcontractor"[online]. available from: http://dx.doi.org/10.29117/cic.2020.0012.

Construction Contracts Acts (2013)

Cohen, L., Manion, L., and Morrison, K. (2002) *Research Methods in Education*. Routledge.

Collis, J., Hussey, R., Crowther, D., Lancaster, G., Saunders, M., Lewis, P., Thornhill, A., Bryman, A., Bell, E., and Gill, J. (2003) 'Business Research Methods'. *Chongqing, China: University Press. Cosier, RA and Harvey, M.(1998), the Hidden Strengths in Family Business: Functional Conflict, Family Business Review* 11 (1), 75-79.

Collis, J. and Hussey, R. (2013) *Business Research: A Practical Guide for Undergraduate and Postgraduate Students*. Macmillan International Higher Education.

- Coppola,F (2018) Carillion's Failure: The Many Questions that Need Answers. Online availablefrom <u>https://www.forbes.com/sites/francescoppola/2018/01/16/carillions-failure-the-many-questions-that-need-answers/?sh=7f2e3f727dce</u> [16 May 2019]
- Construction Excellence Board (2016) A Review of Payments within the UK Construction Industry by Ce Funding and Finance Group-April 2016

- Construction Enquirer (2019) 22 disaster in a year:why firms worth £1.2bn collapsed. [online] available from:<u>https://www.constructionnews.</u> [27 June 2019]
- Construction News (2019) Dozen top contractors owed £1bn in retentions. [online] available from:<u>https://www.constructionenquirer.com/2017/02/15/dozen-top-contractors-owed-1bn-in-</u> retentions/. [27 June 2019]
- Construction Index (2018) *UK Construction's Small Business Now Number More than Million* [online] available from <<u>https://www.theconstructionindex.co.uk/news/view/uk-</u> <u>constructions-small-businesses-now-number-more-than-a-million</u>> [06/02/2018].
- Conell, W. (2014) The Economic Impact of Late Payment [online] European Economy paper 531.availabefrom<u>https://ec.europa.eu/economy_finance/publications/economic_paper/2014/pdf</u> /ecp531_en.pdf [27 April 2018]
- Cornford, T. and Smithson, S. (2005) *Project Research in Information Systems: A Student's Guide*. Macmillan International Higher Education.
- Cooke, B. and Jepson, W.B. (1986) Cost and Financial Control for Construction Firms. London:Macmillan Educational Ltd. Pp. 25-26, 41-4

Construction Supply Chain Payment Charter

Construction Excellence (2004): Promoting teamwork through partnering workshops

- Cotter, P(,2005) Late payment: Why it pays to know your customers, Experian's Business Information Division.
- Clough, R.H., Sears, G.A., Sears, S.K., Segner, R.O. and Rounds, J.L., (2015). *Construction contracting: A practical guide to company management*. John Wiley & Sons.
- Churchill Jr, G.A., (1979) A paradigm for developing better measures of marketing constructs. *Journal of marketing research*, *16*(1), pp.64-73.
- Creswell, J.W. and Creswell, J.(2003) *Research design* (pp. 155-179). Thousand Oaks, CA: Sage publications.
- Creswell,J. W and Creswell J.D (2018) *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches.* Fifth edn; Thousand oaks CA: SAGE Publications.
- Creswell, J., and Plano Clark, V. (2007) *Designing and Conducting Mixed Methods Research*. Thousand Oaks, CA: Sage.
- Creswell, J.W.,and Tashakkori, A.(2007) *Differing Perspectives on Mixed Methods*. Thousands Oaks CA;Sage.
- Creswell, J. W and Clark, P (2018) Designing and conducting mixed methods Research. Thousand Oaks, CA: Sage.
- Crossan, F. (2003) 'Research Philosophy: Towards an Understanding'. *Nurse Researcher* (*through 2013*) 11 (1), 46-55.

Crotty, M. (1998) The Foundations of Social Research: Meaning and Perspective in the

Research Process. Sage

- Dainty, A.(2008) 'Methodological Pluralism in Construction Management Research'. Advanced Research Methods in the Built Environment 1, 1-13
- Dainty, A., Green,S., and Bagilhole, B. (2007) *People and Culture in Construction*: A Reader. Routledge.
- Danuri, M. M., Munaaim, M. C., Rahman, H. A., and Hanid, M. (2006) *Late and Non-Payment Issues in the Malaysian Construction Industry-Contractor's Perspective.* in *International Conference on Construction, Culture, Innovation and Management (CCIM)* held 2006
- Davis, R (1991) Construction Insolvency. London: Chancery Law Publishing Ltd.
- Deloitte Research Report (2018) *SME Digital Payments*. The Paytech Revolution Series 2018: [online] available :https://www2.deloitte.com/content/dam/Deloitte/au/Documents/financial-services/deloitte-aufs-sme-digital-payments-270218.pdf [12 March 2018]
- Department for Business Energy and Industrial Strategy (2017) 2011 Changes to Part 2 of the Housing Grants, Construction and Regeneration Act 1996 [online] available from<u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_dat</u> a/file/655879/2017.10.29 PIR Consultation FINAL.PDF [22 December 2017]
- Department for Business Energy and Industrial Strategy (2018) *Small businesses and productivity* [online] available from <u>https://publications.parliament.uk/pa/cm201719/</u> [22 June 2019)
- Department for Business Energy and Industrial Strategy (2019) *Longitudinal Small Business Survey: SME Employers (Business with 1-249 employees)* [online].available from<u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_dat</u> <u>a/file/889656/LSBS_2019 employers.pdf</u> [22 December 2019)
- Department for Business Energy and Industrial Strategy (2019) *Retention Payments in the Construction Industry: A Consultation on the practice of Cash Retention Under Construction Contracts.* [online] available from <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/fil</u> <u>e/868630/retention-payments-in-the-construction-industry-summary-of-responses.pdf</u> [22 January 2020]
- Department for Business, Innovation and Skills (2013) *Strategy Sets Foundations for UK Construction Growth* [online] available from <<u>https://www.gov.uk/government/news/strategy-</u> <u>sets-foundations-for-uk-construction-growth</u>> [22 July 2017]
- Department for Business, Innovation and Skills (2015) *Challenging Grossly Unfair Payment Terms and Practices* [online] available from<u>https://www.cicm.com/wp-content/uploads/2015/06/bis-15-248-challenging-grossly-unfair-payment-terms-and-practices-summary-of-responses-JUNE-2015.pdf</u> [22 July 2017]
- Department for Business, Innovation and Skills (2015) *Duty to Report on Payment practices and performance- Guidance to reporting on payment practices and performance* [online]available<u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/831507/payment-practices-performance-reportingrequirements.pdf [12 September 2019]</u>

- Department for Business, Innovation and Skills (2013) *Supply Chain Analysis into the Construction Industry - A Report for the Construction Industrial Strategy*: [online] available from<u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_dat</u> <u>a/file/252026/bis-13-1168-supply-chain-analysis-into-the-construction-industry-report-for-theconstruction-industrial-strategy.pdf</u> [10 October 2017]
- Denscombe,M.(2017) *The good research guide: for small-scale social research projects.* 6th ed. McGraw-Hill Education.
- Diamond,D.W.(1984) 'Financial intermediation and delegated monitoring'. *The Review of economic studies*, 51(3),393-414
- Doloi, H. (2009) 'Analysis of pre-qualification Criteria in Contractor Selection and their Impacts on Project Success'. *Construction Management and Economics* 27 (12), 1245-1263.
- Donkor, S. (2011) Determinants of Business Failure: The Perspective of Smes Building Contractors in the Ghanaian Construction Industry. [Online] Published thesis available from <u>http://ir.knust.edu.gh/bitstream/123456789/4695/1/SAMUEL%20DONKOR.pdf</u>[10 November 2017]
- Donna, M.M(1998) Research Methods in Education and Psychology: Integrating Diversity with Quantitative & Qualitative Approaches. Sage Publications.
- Dornyei, Z.,(2007) Research Methods in Applied Linguistics. NewYork: Oxford University Press.
- Dubois, A and Gadde, L.E(2002) 'Systematic combining: An abductive approach to case research'. *Journal of Business Research* 55(7):553-560.
- Duffy, M.E.(1987) 'Methodological Triangulation: A Vehicle for Merging Quantitative and Qualitative Research Methods'. *Image: The Journal of Nursing Scholarship* 19 (3), 130-133.
- Easterby-Smith, M. T. and Thorpe, R. (2002) 'R. and Lowe, A.(2002)'. *Management Research: An Introduction* 2,342.
- Easterby-Smith, M., Thorpe, R., and Jackson, P. R. (2012) *Management Research*. Sage.
- El-adaway, I., Fawzy, S., Burrell, H., and Akroush, N. (2017) 'Studying Payment Provisions Under National and International Standard Forms of Contracts'. *Journal of Legal Affairs and Dispute Resolution in Engineering and Construction* 9 (2), 04516011.
- Elo,S., and Kynga,H.,(2008) ' The qualitative content analysis process'. Journal of Advanced Nursing 62(1)107-115.
- Eom, S.J.,Kim, S. C. and Jang, W.S. (2015) 'Paradigm Shift in Main Contractor-Subcontractor Partnerships with an e-Procurement Framework'. *KSCE Journal of Civil Engineering* 19 (7), 1951-1961.
- Eriksson, P. E. (2008) 'Procurement Effects on Coopetition in Client-Contractor Relationships'. *Journal of Construction Engineering and Management* 34 (2),103-111.
- Euler Hermes (2015) Euler Hermes: UK late payments hit two-year high at 2015 year-end [online] Press release available from <u>https://www.eulerhermes.com</u> [27 December 2019].

- Euler Hermes (2017) Insolvency and Payment. Available at: <u>http://www.eulerhermes.com/economic-research/themes/insolvency-payment-risk/Pages/default.aspx</u> [11 December 2017].
- Euginie,L.(2006) Curing the ills of non-payment in the construction industry- the Singapore Experience .*In 8th Surveyors' Congress*,Kuala lumpur.ISM.
- European Payment Report (2019). Intrum (online) available from <u>https://www.intrum.com/publications/european-payment-report/european-payment-report-2019/</u>[19 November 2019].
- European Commission Payment Directive 2011/7 EU- Combating late payment in commercial transactions. Published by Official Journal of the European Union.available from https://www.eubusiness.com/topics/finance/late-payment-directive / [14 December 2017].
- European Payment Report (2016). Intrum (online) available from <u>https://www.intrum.com/publications/publications-archive/</u>[19 November 2019].
- Federation of Small Business (2016) Time to Act the Economic Impact of Poor Payment Practice [online] available from <u>https://www.fsb.org.uk/resource-report/time-to-act.html</u> [15 December 2017].
- Federation of Small Business (2019) Chain Reaction: Improving the Supply Chain Experience for Smaller Firms [online] available from <u>https://www.fsb.org.uk/resource-report/chain-reaction-improving-the-supply-chain-experience-for-small-firms.html</u> [17 August 2019].
- Federation Master Builders (2017) Three Quarter of UK Builders Under Threat from "Cow Clients" Online .available from <u>https://www.fmb.org.uk/</u> [7 September 2019].
- Federation Master Builders (2018) Nearly 80% of builders call for licensing schemen to crush the cowboys,says FMB" Online .available from_ [7 September 2019]. <u>https://www.fmb.org.uk/resource/nearly-80-of-builders-call-for-licensing-scheme-to-crush-the-cowboys-says-fmb.html</u> [10 July 2018]
- Federation Small Business (2018) Nine Out of Ten Public Sector Suppliers Hit by Late Payment. [online] availablefrom<<u>https://www.fsb.org.uk/media-centre/press-releases/nine-out-of-ten-public-sector-suppliers-hit-by-late-payments</u>> [27 January 2019]
- Fellows,R. and Liu,A.M.M (2008) Research Methods in Education and Psychology: Integrating Diversity with Quantitative & Qualitative Approaches. Sage Publications.
- Field, B. and Ofori, G. (1988) 'Construction and Economic Development: A Case Study'. *Third World Planning Review* 10 (1),41.
- Fong, P. S. and Choi, S. K. (2000) 'Final Contractor Selection using the Analytical Hierarchy Process'. *Construction Management and Economics* 18 (5),547-557.
- Forgues, D, Iordanova, I., Valdivesio, Fand Staub-French, S (2012) Rethinking the cost estimating process through 5D BIM: A case study. In Construction Research Congress 2012: Construction Challenges in a Flat World (pp. 778-786).

- Franco-Santos, M., Kennerly, M., Micheli, P., Martinez, V., Mason, S., Marr, B., Gray, D., and Neely, A. (2007) 'Towards a Definition of a Business Performance Measurement System'. *International Journal of Operation and Production* 27(8),784-801.
- Frimpong, Y., Oluwoye, J. and Crawford, L., 2003. Causes of delay and cost overruns in construction of groundwater projects in a developing countries; Ghana as a case study. *International Journal of project management*, *21*(5), pp.321-326.
- FTI Consulting (2017) Is the Construction Industry Ripe for a New Operating Model . [online] December .available from<u>https://www.fticonsulting-emea.com/</u> [27 January 2019]
- 'Gerrad N' (2018) 'Construction Insolvencies Soar After Carillion Liquidation''. *Construction Manager* [online] October.available from https://www.constructionmanager [21 February 2019].

Gill, J. and Johnson, P. (2010) Research Methods for Managers: Sage.

Gibson,A. (2000)\$1.5M contract payout delay.New Zeland Herald.available from <u>https://www.nzherald.co.nz/</u> [8 May 2018]

- Grant,B.M. and Giddings, L.S.(2002) 'Making Sense of Methodologies: A Paradigm Framework for the Novice Researcher'. *Contemporary Nurse* 13 (1), 10-28.
- Grant, P. (2019) 'SME Bosses Go without Wages due to Late Payment Abuse'. *Construction Enquirer* (Online).available from <u>https://www.constructionenquirer.com/</u> [27 June 2019]
- Gluck, G.,(1979) Standard form contracts: The contract theory reconsidered. *International and Comparative Law Quarterly* 72-90.
- Griffiths, R., Lord, W., and Coggins, J. (2017) 'Project Bank Accounts: The Second Wave of Security of Payment?'. *Journal of Financial Management of Property and Construction* 22 (3), 322-338.
- Gruneberg, S. and Ive, G. J. (2000) *The Economics of the Modern Construction Firm*: Springer.

Guba, E.G.,(1990) 'The paradigm dialog'.San Franciso,CA; Sage Publications.

Hair, J.F., Money, A.H., Samouel, P. and Page, M., (2007) Research methods for business. Education+ Training.

Halpin, D.W and Woodhead, R.W, (1998) Construction Management. Wiley, NewYork.

Harris, F and McCaffer, R (2003) *Modern Construction Management*. 5th ed.Oxford.

Hasmori M F, Ismail I and Said I (2012) "Issues of Late and Non-Payment Among Contractors in Malaysia", 3rd *International Conference on Business and Economic Research Proceeding*, 12-13 March 2012, Golden Flower Hotel, Bandung, Indonesia.

Haugh, H. (2012) 'The Importance of Theory in Social Enterprise Research'. *Social Enterprise Journal* 8 (1), 7-15.

Hamzah, A., Chen, W., Mohd-Danuri, M.S., & Che-Munaaim, M.E. (2014). Cause, effect, and reaction for late-payment and non-payment issues in Malaysian building sector. Journal of Business Management and Innovations, 1, 8 – 20.

Hart, D. and Holmstrm, B., (1986) The theory of contracts.

Hillebrandt, P. M. (1985) Economic Theory and the Construction Industry.: Springer.

- Hinze, J. and Tracey, A.(1994)' The Contrator-Subcontractor Relationship: The Subcontractor's View'. *Journal of Construction Engineering and Management 120 (2) 247-287.*
- Holt, G. D. (2013) 'Construction Business Failure: Conceptual Synthesis of Causal Agents'. *Construction Innovation* 13 (1), 50-76.
- 'House of Commons' (2014) *Paying Government Suppliers on Time*. London: National Audit Office.
- Housing Grants, Construction and Regeneration Act (HGCRA) 1996.London: Telford 1996.
- Howorth, C. and Wilson, N., 1998. Late payment and the small firm: an examination of case studies. *Journal of Small Business and Enterprise Development*.
- Hughes, W., Hillebrant, P and Murdoch, J (1998) *Financial Protection in the UK building Industry: Bond, Retention and Guarantees*. London: E & FN Spon.
- Hughes, W., Hillebrandt, P. and Murdoch, J. (2000) The impact of contract duration on the cost of cash retention. *Construction Management and Economics*, 18 (1). pp. 11-14.
- Hughes, W., Champion, R., and Murdoch, J. (2015) *Construction Contracts: Law and Management*. Routledge.
- Hughes, W., Hillebrandt, P.M., Greenwood, D. and Kwawu, W., (2006) Procurement in the construction industry: the impact and cost of alternative market and supply processes. Routledge.
- Humphreys, P., Mathew, J. and Kumaraswamy, M. (2003) 'Pre-construction project partnering: from adversarial to collaborative relationships'. *Supply Chain Management*. 8(2) 166-178.
- Ihuah, P. W. and Eaton, D.(2013) 'The Pragmatic Research Approach: A Framework for Sustainable Management of Public Housing Estates in Nigeria'. *Journal of US-China Public Administration* 10 (10), 933-944.
- Jabar, M. A., Sidi, F., Selamat, M. H., Ghani, A. A. A., and Ibrahim, H. (2009) 'An Investigation into Methods and Concepts of Qualitative Research in Information System Research'. *Computer and Information Science* 2 (4), 47.
- James,S. (2019) *Government Announces Proposed New Powers to Tackle Late Payment.* [2019] online June. available from<u>https://cpa.co.uk/government-announces-proposed-new-powers-to-tackle-late-payments/</u> [23 September 2019].
- Jiang,A.,(2012) 'Negotiating construction contracts through practical cashflow planning and analysis model'. *International Journal of Construction Management* 12(2)23-33.
- Jin, W., Mohan, K., and Soo, G. (2008) 'Payment Problems and Regulatory Responses in the Construction Industry: Mainland China Perspective'. Journal of Professional Issues in

Engineering Education and Practice 134(4) 399-407.

- Johnson, R.B and Onwuegbuzie, A.J and Turner, L.A (2007) 'Toward a definition of mixed methods research'. *Journal of Mixed Methods Research* (1)112-133.
- Johnson, R.B. and Onwuegbuzie, A.J.,(2004) Mixed methods research: A research paradigm whose time has come. *Educational researcher*, *33*(7), *pp.14-26*.

Joint Contracts Tribunal (JCT SBC/Q 2016 - Standard Building Contract with Quantities.

- Judi, S. S. and Rashid, R. A. (2010) 'Contractor's Right of Action for Late Or Non-Payment by the Employer'. *Journal of Surveying, Construction and Property* 1(1).
- Kaid, L. L. and Wadsworth, A. J.(1989) 'Content Analysis'. *Measurement of Communication Behavior* 197-217.
- Kamal, E. M. and Flanagan, R. (2014) 'Key Characteristics of Rural Construction SMEs'. *Journal of Construction in Developing Countries* 19 (2).
- Kagioglou, M. Cooper, R and Aouad, G., (2001) 'Performance management in construction a conceptual framework'. Construction Management and Economics 19 85-95.
- Kenyatta, M.O.(2016) 'Sources of Payment Problems in the Construction Industry of Kenya: A Perspective of the National Construction Authority'. *International Journal of Engineering and Advanced Technology* 5(9) 36-42.
- Kenley, R.(2002) Milking the cash cow. *The New Zealand Herald*. [online] Available at: <u>http://www.nzherald.co.nz/business/news/article</u>.
- Kelly Tolhurst (2018)- Creating a responsible payment culture: A call for evidence in tackling late payments. Published by Departement for Business ,Energy and Industrial Strategy.
- Kessler, F(1943) The Contracts of Adhesion–Some Thoughts about Freedom of Contract Role of Compulsion in Economic Transactions'. Columbia Law Review, 43, p.629.
- Ketokivi,M. and Mantere,S.,(2010) 'Two strategies for inductive reasoning in organizational research'. *Academy of management review* 35(2) 314-333.
- Kilgallon, B. (2013). The Late Show: Project Bank Accounts. RICS Construction Journal, pp. 14-17
- Klenke, K. (2016) *Qualitative Research in the Study of Leadership*: Emerald Group Publishing Limited.
- Kokkola, T. (2010) The payments system:Payments,securities and derivatives,and the role of the Eurosystem. European Central Bank.available from<u>https://www.ecb.europa.</u>[27 May 2019].
- Kometa, S. T., Olomolaiye, P. 0 and Frank, C. H. (1995a) Quantifying Client-generated Risk by Project Consultants. *Construction Management and Economics*, 12, pp. 433-

443.

- Kothari, C. R. (2004) *Research Methodology Methods and Techniques*. 2nd ed. New Delhi: New Age International Publishers Ltd.
- Kuhn, T. (1962) 'The Structure of Scientific Revolutions. Chicago: University'. Press, Chicago.
- Kumaraswamy, M.M., Anvuur, A.M. and Mahesh, G.,(2008) 'Contractual framework and cooperative relationship'. *Collaborative relationships in construction: developing frameworks and networks* 78-104.
- Kuzmanic, M. (2009) 'Validity in qualitative research: Interview and the appearance of truth thorugh dialogue'. *Horizons of Psychology* 18(2) 39-50.
- Latham, M. (1994) Constructing the Team: Joint Review of Procurement and Contractual Arrangements in the United Kingdom Construction Industry. London: HMSO.
- Laryea,S and Hughes,W. (2006) The price of risk in construction projects. In Proceedings of 22nd Annual ARCOM Conference, Association of Researchers in Construction Management,UK,Vol.2,pp 553-561
- Lean, C.S.(2001) 'Empirical Tests to Discern Linkages between Construction and Other Economic Sectors in Singapore'. *Construction Management and Economics* 19 (4),355-363.
- Li, Y. and Li (2015) Expatriate Manager's Adaption and Knowledge Acquisition. Springer.
- Lincoln,Y.S., Lynham,S.A and Guba,E.G.,(2011) 'Paradigmatic controversies,contradications and emerging confluences, revisited'. *The Sage handbook of qualitative research* (4) 97-128
- Lincoln,Y.S and Guba,E.G (1986) But is it rigorous? Trustworthiness and authenticity in naturalistic evaluation. In *New directions for program evaluation* 30, 73-84.San Franciso,CA: Josey-Bass.
- Local Democracy, Economic Development and Construction Act (2009).
- Love,P.E.D and Holt,G.D (2000) 'Construction business performance measurement: the SPMalternative'. Business Process Management 6(5) 408-416.
- Loxley, J. (2018) *The Collapse of P3 Giant Carillion and its Implication*. [online] available from <u>https://www.policyalternatives.ca/</u> [10 February 2019].
- Lu, S., Sexton, M. G., and Abbot, C. (2008) *Key Characteristics of Small Construction Firms: A United Kingdom Perspective.* 'In Proceedings: CIBW065/055 Joint International Symposium: Transformation through Construction' held 2008
- Lucey, T. (2009) Costing. London: Cengage Learning
- Macaulay, S. (1963) 'Non-Contractual Relations in Business: A Preliminary Study'. *American* Sociological Association 1 (28), 55-67

- Mack, N., Woodsong, C., MacQueen, K. M., Guest, G., and Namey, E. (2005) 'Qualitative Research Methods: A Data Collectors Field Guide.
- Mackenzie, N. and Knipe, S. (2006) 'Research Dilemmas: Paradigms, Methods and Methodology'. *Issues in Educational Research* 16 (2), 193-205.
- Macneil, I. (1986) 'Exchange Revisited: Individual Utility and Social Solidarity'. *Ethnics an International Journal of Social, Political and Legal Philosophy* 96 (3), 567-593.
- Malalgoda, C., Amaratunga, D., and Haigh, R. (2014) 'Challenges in Creating a Disaster Resilient Built Environment'. *Procedia Economics and Finance* 18, 736-744.
- Marr, B. and Schiuma, G (2003) 'Business Performance Measurement Past, Present, and Future '. *Management Decision* 41 (8), 680-7.
- Mason, J. (2016) Construction Law: From Beginner to Practitioner. Routledge.

Masterman, J (2002). An introduction to Building Procurement Systems. 2nd edn. London: E & FN Spon.

- Maylor, H., Blackmon, K., and Huemann, M. (2016) *Researching Business and Management*. Macmillan International Higher Education.
- Mbachu,J.,(2011) 'Sources of contractor's payment risks and cashflow problem in the New Zealand construction industry:Project team's perceptions of the risks and mitigation measures'. *Construction Management and Economics* 29(10),1027-1041.

McCann,S (1999) 'Time to Pay Subcontractor'. New Law Journal Practitioner, 1078.

- McLernon, T. and McClements, S. (ed.) (2015) 'Comparing and Evaluating the Contractual Provisions of Two Analogous Construction Contracts for the Management of Cost.'. *RICS* Cobra.
- Mertens, D M. (2005). Research and evaluation in education and psychology: Integrating diversity with quantitative, qualitative and mixed methods (2nd ed.). Thousand Oaks, CA: Sage.
- Mertens,D.M (2010) Research and evaluation in education and psychology: integrating diversity with quantitative, qualitative and mixed methods. 3rd ed. Thousand Oaks,CA;Sage.
- Miles,M.B,Huberman,A.M,andSaldaña,J.,(2014)Qualitative data analysis. A Methods Sourcebook.3 edn. Thousands Oaks,CA; Sage Publications

Miller, R. L., and Brewer, J. D. (2003). A-Z of Social Research. London: Longman.

Mitchell,D (2012) 5D BIM: Creating cost certainty and better buildings.RICS COBRA, pp.1-9.

MHA (2018) *UK Construction Sector 2018*. [online]. available from <u>https://mha-uk.co.uk/insight/knowledge/2018-uk-construction-sector-report/</u> [18/07/2019].

Mkansi, M. (2012) 'Research Philosophy Debates and Classifications: Students' Dilemma'. *Journal of Business Research Methods* 10(2) 132-140.

Moore, C.C., Mosley, D.C. and Slagle, M., (1992) Partnering: guidelines for win-win project management. Project Management Institute.

- Morgan, D. L. (2007) 'Paradigms Lost and Pragmatism Regained: Methodological Implications of Combining Qualitative and Quantitative Methods'. *Journal of Mixed Methods Research* 1 (1), 48-76
- Munaaim,M.C.,Dauurl,M.M.andAbdul-Rahman,H.,(2007) Is late or non-payment a significant problem to Malaysian contractors. *Journal of Design and Built Environment*, *3*(1).

Murray, M. (2003) Construction Reports 1944-98. Oxford : Blackwell Science.

Murdoch,J and Hughes,W. (2000) Construction contracts: Law and management. (3red).London:Spon Press

Myers, M. D. (2013) Qualitative Research in Business and Management.Sage.

National Federation of Builders (2018) *Withholding Subcontractor Payments is Unethical.* available from <<u>https://www.builders.org.uk/news/withholding-subcontractor-payments-is-unethical/</u>> [24/02/2018]

Nachmias, D. and Nachmias, C (2008) Research Methods in the Social Sciences. Macmillian .

- National Audit Office (2005) Improving Public Services through better construction. A report by the Comptroller and Auditor General Ordered by the House of Commons to be printed on 14 March 2005.
- National Audit Office (2018) Cabinet Office Investigation into the government's handling of the collapse of Carillion. A report by the Comptroller and Auditor General Ordered by the House of Commons to be printed on 5 June, pp. 22 32
- 'National Specialist Contractors (2014)- Written evidence submitted by the National Federation of Roofing Contractors. available from <<u>https://publications.parliament.uk/pa/cm201415/cmpublic/smallbusiness/memo/sb41.htm</u> > [18/07/2017].
- 'National Specialist Contractors (2014) survey- £250 million of retentions is owed to specialists. available from ">https://spectormagazine.co.uk/%EF%BF%A1250-million-of-retentions-is-owed-to-specialists/ [18/07/2017].
- National Construction Association of Sri Lanka, (NSCL 2008) Report on Payment Delays in Sri Lankan Construction Industry. National Construction Association of Sri Lanka, Battaramulla, Sri Lanka
- National Construction Contracts and Law Report (2018) online.available from https://www.thenbs.com/knowledge/national-construction-contracts-and-law-report-2018 [18/01/2019]

NEC4 Engineering and Construction Contract (ECC)

- Neuman, W. L(2009) Social research methods; Qualitative and quantitative approaches. 7th ed;Boston;Ally and Bacon.
- Office for National Statistic (2019) Construction Output in Great Britain.[online]. available from <u>https://www.ons.gov.uk/businessindustryandtrade/constructionindustry/bulletins/constructionout</u> <u>putingreatbritain/september2019</u> [22/11/2019].
- Office of Government Commerce (2007) *Guide to Best 'Fair Payment Practices*. London: Office of Government Commerce.
- Okeyo, M. P., Rambo, C. M., and Odundo, P. A. (2015) 'Effects of Delayed Payment of Contractors on the Completion of Infrastructural Projects: A Case of Sondu-Miriu Hydropower Project, Kisumu County, Kenya'. *Chinese Business Review* 14 (7), 325-336.
- O'Leary,Z (2004) The essential guide to doing research. London;Sage.
- Pathirage, C., Amaratunga, R., and Haigh, R. 'The Role of Philosophical Context in the Development of Theory: Towards'. *Construction Management and Economics* 11, 221-245.
- Paul,S. and Wilson,N (2006) 'Trade Credit Supply: an Emprical Investigation of Companies Level Data'. *Journal of Accounting, Business and Management Construction* 13,85-113.
- Patton, M.Q., (1990) Qualitative evaluation and research methods. SAGE Publications, inc.
- Patton, M.O (2002) *Qualitative research and evaluation methods*.3rd ed Thousand; Oaks CA; Sage.
- Peter, L and Arewa, A(2018) 'Potentiality of Emerging Technologies to Minimise Late-Payments Quandary in Construction'. *In: Gorse and Neilson, C, J (Eds) Proceedings of the 34th Annual ACROM Conference*. held 3-5 September at Belfast, UK: Association of Researchers in Construction Management.
- Pettigrew,R.(2005) *Payment Under Construction Contracts Legislation*.Thomas Telford Publishing.
- Peat,J., Mellis,C.,William,K and Xuan,W.(2002) Health Science Research: A Handbook of Quantitative Methods.London:Sage.
- Pheng Low, S., Liu, J., and Sio, S. (2010) 'Business Continuity Management in Large Construction Companies in Singapore'. *Disaster Prevention and Management: An International Journal* 19 (2), 219-232.

Pietroforte, R., Bon, R., and Gregori, T. (2000) 'Regional Development and Construction in

Italy: An Input-Output Analysis, 1959-1992'. *Construction Management & Economics* 18 (2), 151-159.

- Powell,G.(2012) Construction contract preparation and management: from concept to completion, Basingstoke: Palgrave Macmillan
- Plimmer, G. (2013) 'Carillion Attacked Over Subcontractor Payments'. *The Financial Times* [online] available from:<u>https://www.ft.com/</u> [2 August 2019]
- Price, D. (2018) *Build UK Exposes Poor Payment Record of Main Contractors*. online January [2018] available from <<u>https://www.theconstructionindex.co.uk/news/view/build-uk-exposes-poor-payment-record-of-main-contractors</u>> [19 November 2019]
- Price, D. (2019) *Carillion hit and cashflow squeeze forces Hawk Plant under*. Construction News online January available from <u>https://www.constructionnews.co.uk</u> [22 November 2019].
- Prior,G (2020) *SME Bosses go without wages due to late payment abuse.* Construction Enquirer [online] available from <u>https://www.constructionenquirer.com/2019/11/28/sme-bosses-go-without-wages-due-to-late-payment-abuse/</u> [22 July 2020].
- Public Contracts Regulations (2015)
- Punch,K (2013) Introduction to social research: Quantitative and qualitative approaches. Sage.
- Pye Tait Consulting Report (2017) Retentions in the Construction Industry.BEIS Research Paper 17 available from;<u>https://assets.publishing.service.gov.uk/</u> [2 May 2018].
- QU,S.Q.,and Dumay,J. (2011) ' The qualitative research interview'. *Qualitative Research in Accounting and Management* 8(3),238-264.
- Quesada-Pineda,H.J (2019) Analysis of Financial Statement using Ratios [online].available from <u>https://vtechworks.lib.vt.edu/b.pdf?sequence=1&isAllowed=y</u> [19 November 2019]
- R.A. Stewart, Mohamed, S., and Marosszeky, M. (2004) 'An Empirical Investigation into the Link between Information Technology Implementation Barriers and Coping Strategies in the Australian Construction Industry'. *Construction Innovation* 4 (3), 155-171.
- Rachel Reeves (2018) BEIS Committee Chair writes to companies on payment practices [online].available from <u>https://committees.parliament.uk/committee/365/business-energy-and-industrial-strategy-committee/news/97548/beis-committee-chair-writes-to-companies-on-payment-practices/</u> [10 August 2018].
- Raj, B. V.A. and Kothai, P. (2014) 'Improving the Labour Productivity through Other Resources in Construction Field'. *International Journal of Engineering Research and General Science* 2 (2), 205-213.
- Ramachandra, T. and Rotimi, J.O.B., (2011). The nature of payment problems in the New Zealand construction industry. *Australasian Journal of Construction Economics and*

Building, 11(2), 22-33.

- Ramachandra, T. and Rotimi, J.O.,(2015) 'The construction contracts act and its implcations for payment defaults in New Zealand'. Journal of Legal Affairs and Dispute Resolution in Engineering and Construction 8(1).
- Ramachandra, T. (2013) Exploring feasible solutions to payment problems in the construction industry in New Zealand. Auckland University of Technology.AuckaInd.AUT.Published Thesis online.
- Rameezdeen, R. (2007) 'Image of the Construction Industry'. Proceedings CIB Priority Theme: Revaluing Construction, *Aw065'Organization and Management of Construction 'Perspective, CIB Publication* 313, 76-87.
- Rameezdeen, R. and Ramachandra, T.(2008) 'Construction Linkages in a Developing Economy: The Case of Sri Lanka'. *Construction Management and Economics* 26 (5), 499-506.

Reeves K (2003) 'Pay Up'. JUBM Construction News and Views 1: 4-6.

Richard Beresford (2018) Construction UK Magazine.

RICS guidance note (2015) Interim valuations and payment.1st edition.

Rhodes,C. and Ward,M (2019) *Business Statistics*. Briefing Paper 06512[online].available from:<u>https://commonslibrary.</u> [19 January 2020].

RSM (2017) Construction Sector Access to Finance. London: RSM International Association.

Russell, D., (2007) *Building supply chain collaboration: a typology of collaborative approaches*. The International Journal of Logistics Management.

- Saunders, M., Lewis, P., and Thornhill, A. (2009) 'Understanding Research Philosophies and Approaches'. *Research Methods for Business Students* 4, 106-135.
- Saunders, M., Lewis, P. and Thornhill, A. (2016) Research Methods for Business Students. 7th Edition, Pearson, Harlow.

Saunders, M. N. (2019) Research Methods for Business Students. 8th ed. Pearson Education.

- Seymour, D., Crook, D., and Rooke, J. (1997) 'The Role of Theory in Construction Management: A Call for Debate'. *Construction Management & Economics* 15 (1), 117-119.
- Sarah Mcmonagle (2017) Three Quarter of UK Builders Under Threat from "Cow Clients'. available from <u>https://www.fmb.org.uk/about-the-fmb/newsroom/three-quarters-of-uk-builders-under-threat-from-cowboy-clients/</u> [19 August 2018].
- Shavelson,R.J., and Cronbach,L.J (2004) 'My current thoughts on coefficient alpha and successor procedures'. *Educational and psychological Measurement* 64(3),391-418.

- Sharkey, J., Bell, M., and Jocic, W. and Marginean R. (2014) *Standard Forms of Contract in the Australian Construction Industry*. University of Melbourne.
- Sherif,E. and Kaka, A.(2003). Factors influencing the selection of payment systems in construction projects. In *Proceedings of the 19th Annual Association of Researchers in Construction Management (ARCOM) Conference, Putney, VT, USA* (pp. 3-5).
- Singh,J.Y and Yadav,D.P, (2013) 'Return on Capital Employed-A Tool for Analyzing Profitability of Companies'. *International Journal of Techno-Management Research* 1,01.
- Singh, S. and Lakanathan, G. (1992) Computer-based cash flow model, In Proceedings of the 36th AnnualTtransactions of the American Association of Cost Engineers - AACE, AACE, WV, USA, No. R.5.1-R.5.14
- Sin,A.S.T (2006) 'Payment issues-the present dilemmas of Malaysian construction industry (Master's Thesis). Universiti Teknologi Malaysia, Johor Bahru, Malaysia. available from http://eprints.utm.my/2313/1/AngSuSinMBD2006TTT.pdf [19 August 2017].
- Smith, T. M. & Reece, J. S. (1999) 'The relationship of strategy, fit, productivity and business performance in a service setting'. *Journal of Operations Management* 17 (2) 145-161.
- Smith, M. J. (1998) Social Science in Question: Towards a Post disciplinary Framework. Sage
- Somkekh,B and Lewin,C(2005) *Research methods in the social sciences*. Thousand Oaks. Sage.
- Sood, S (2016) Scourge of late payment still persists in construction year late payment got a little less payments.[online] available at: <u>https://www.constructionnews.co.uk/news/archive-news/2016-the-year-late-payment-got-a-little-less-late-03-03-2016/</u> [12 October 2017]
- Supardi, A. Adnan, H. & Mohammed, M. F. (2011). Security of payment regime in construction industry: Are Malaysian sub-contractors ready? The Built and Human Environment Review, 4 (1),122-137.
- Suzannah Nichols (2016) 2016:the year late payment got a little less payments.[online] available at: <u>https://www.constructionnews.co.uk/news/archive-news/2016-the-year-late-payment-got-a-little-less-late-03-03-2016/</u> [12 October 2017]
- Sykes, W. (1990) 'Validity and Reliability in Qualitative Market Research: A Review of the Literature.' *Journal of the Market Research Society.*
- Tashakkori, A., and Teddlie, C. (2003) *Handbook of mixed methods in social and behavioural research*. London: Cassell
- The Department of Environment ,Transport and Regions (DETR) (1998a) *Combating Cowboy Builders: A Consultation Paper*. London:HMSO
- The Insolvency Service Company (2019) *Insolvency Stastics*, Q4 October to December 2019. UK [online] available from<u>https://www.gov.uk/government/statistics/company-insolvency-statistics-october-to-december-2019</u> [12 January 2020].

The Late Payment of Commercial Debts Regulations 2013

The Scheme for Construction Contracts (England Wales) Regulation 1998.

The Small Business, Enterprise and Employment Act (2015)

The Public Contract Regulation (2015)

Tan, Y, Xue, B. and Cheung, Y.T.(2017) Relationships between main contractors and subcontractors and their impacts on main contractor competitiveness: An empirical study in Hong Kong. *Journal of Construction Engineering and Management*, *143*(7), p.05017007.

Todd,J(1979) Mixing Qualitative and Quantitative Methods: Triangulation in Action. *Journal of Administrative Science Quarterly* Vol.24,No.4 pp 602-611.

Tony D., (2017) 'Late Payment Damaging SME Suppliers in the Construction Sector'. *Infrastructure Intelligence* 19 July.

- TSheets (2018) Construction Companies Hurry Up and Wait ... to Get. [online] available from <<u>https://blog.tsheets.com/2018/business-help/construction-cash-flow-survey</u>> [24/05/2018]
- Tungsten Late Payments Continue to Blight Construction Firms ; Quarter of SMEs in the Sector at Risk of being Unable to Continue Trading [online] available from <<u>https://www.tungsten-network.com/media/16603016/15-10-09-late-payments-continue-toblight-construction-firms-final.pdf</u>> [12/09/2017]
- TUC report 'What lessons can we learn from Carillion and what changes do we need to make'. April [online]. available from<u>https://www.tuc.org.uk/sites/default/files/Lessons from</u> Carillion report.pdf [15/08/ 2018]
- Turner, J.R., (2004) 'Farsighted project contract management: incomplete in its entirety'. *Construction Management and Economic* 22(1) 75-83.
- Trochim W. M., and Donnelly, J. P. (2008) *Research methods knowledge base*. 3rd ed. Mason, OH: Cengange Learning.
- Van Maanen J (1983) Qualitative Methodology. London: Sage.
- Ward,S.C.,Curtis, B. and Chapman, C.B. (1991) 'Objectives and performance in construction projects' *Construction Management and Economics* 9, 343–53.
- Williams, C. (2007) 'Research Methods'. *Journal of Business and Economic Research* 5 (3), 65-72.
- Winch, G. (2010) Managing Construction Projects, Oxford: Blackwell Science.
- Winter, G.,(2000) 'A Comparative Discussion of the Notion of 'Validity' in Qualitative and Quantitative'. *The Qualitative Report* 4(3),1-14
- Wheeler H (2016) More than half of construction subcontractors forced to accept contract terms through fear loosing business.[online].available

https://www.bibbyfinancialservices.com/about-us/news-andinsights/news/2016/subcontractors-forced-to-accept-contracts

- Wong, C. H., Holt, G. D., and Cooper, P. A. (2000) 'Lowest Price Or Value? Investigation of UK Construction Clients' Tender Selection Process'. *Construction Management and Economics* 18 (7), 767-774.
- Wu, J., Kumaraswamy, M., & Soo, G. (2008). Payment problems and regulatory responses in the construction industry: Mainland China perspective. *Journal of Professional Issues in Engineering Education & Practice*, 134(4), 399-407.
- Yao, A. (2014). Effects of delayed payment on projects and stakeholders: A case study of Keta Municipal Assembly. (Unpublished M.Sc. thesis). Kwame Nkrumah University of Science and Technology, West Africa
- Ye, K. M. and Rahman, A. (2010) 'Risk of Late Payment in the Malaysian Construction Industry'. *International Scholarly and Scientific Research and Innovation* 4 (5), 81-89
- Yin, R. K.(2003) Case Study Research. 3rd edn. London, England: Sage Publications
- Zaghloul,R.and Hartman, F.,(2003) Construction contracts: the cost of mistrust. *International journal of project management*, 21(6), pp.419-424

APPENDIX A – Questionnaire Survey 1

Please read each question carefully and tick the appropriate box that best reflects your opinion, perception and judgement concerning unfair payments practices in the UK construction industry. Please endeavour to answer every question honestly and do not offer biased responses to portray your company or the construction industry off in a good light. Your anonymity and confidentiality are assured.

In the context of unfair payment practices in construction please provide your judgement regarding extent to which you agree or disagree with the following statements:

- 1. Unfair payment practices is a persistence problem to construction contractors
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 2. Unfair payment practices is rampant because of "I cannot" pay attitude of clients Very strongly agree
 - Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 3. Unfair payment practices exists due to "I would not" pay attitude of clients.
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 4. Clients strategic plan to boost cashflow exacerbate unfair payments
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- Payment provisions in standard forms of contracts (SFoCs) helps to minimise Unfair payment occurrence Very strongly agree

Strongly agree

Agree

Disagree Strongly disagree Very strongly disagree Unsure

- 6. SFoCs (JCT/NEC/FIDIC) have sufficient mechanisms to protect contractors' from unfair payment
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 7. 28 days payment period stipulated by various SFoCs help to minimise unfair payments Very strongly agree Strongly agree Agree Disagree Strongly disagree
 - Very strongly disagree
 - Unsure
- 8. Payments procedures stipulated within SFoCs are not robust enough to minimise unfair payment
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 9. Contractors are reluctant to use penalties stipulated in regulations, SFoCs and chartersconcerning unfair payments because of fear that will damage their business relationship
 - Very strongly agree
 - Strongly agree
 - Agree
 - Disagree
 - Strongly disagree
 - Very strongly disagree
 - Unsure
- 10. Unfair payment problems varies between tiers of construction clients
 - Very strongly agree
 - Strongly agree

 - Agree Disagree
 - Strongly disagree
 - Very strongly disagree
 - Unsure

- 11. Unfair payment practices are frequent among Tier 1 clients (.i.e client to main contractor)
 - Very strongly agree
 - Strongly agree
 - Agree
 - Disagree
 - Strongly disagree
 - Very strongly disagree
 - Unsure
- 12. Unfair payment practices are frequent among Tier 2 clients (.i.e main contractor tosubcontractors)
 - Very strongly agree
 - Strongly Agree
 - Agree
 - Disagree
 - Strongly disagree
 - Very strongly disagree
 - Unsure
- 13. Unfair payments practice are frequent among Tier 3 clients (.i.e subcontractors to subsubcontractors/ or supplier)
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 14. Unfair payments influences overall contractors' financial performance
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 15. Unfair payments significantly influence contractors' profitability
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 16. Unfair payment is a threat to contractors' business survival
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree

Very strongly disagree Unsure

- 17. Unfair payments do not significantly influence contractor
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 18. Emerging digital technologies (such as BIM, Agresso, et) have potentials to minimise unfair payments compared to traditional payment procedures in the construction industry
 - Very strongly agree Strongly agree Agree Disagree Strongly disagree Very strongly disagree Unsure
- 19. Could you indicate on average how long does it takes for contractor/subcontractors toreceive payment from clients.
 - o 15 to 30 days
 - 30 to 50 days
 - 50 to 90 days
 - o 90 to 120 days
 - Above 120 day
- 20. Please indicate type(s) of projects with common (frequent) cases of unfair payment.
 - Privately funded projects.
 - Government funded projects.
 - Public-private partnership projects.
- 21. From your experience rate the extent of your agreement to the following causes of unfair payment practices using; 5= very strongly agree, 4=strongly agree, 3=agree, 2= Disagree,1= Strongly disagree, 0 =Very Strongly disagree and was assigned unsure

S/N	Cultural -related causes	5	4	3	2	1
1	Client-contractor relationship					
2	Pay when paid					
3	I "would not" pay attitude of client					
4	I "cannot" pay attitude of client					
5	Prevailing culture					
S/N	Industrial-related causes	5	4	3	2	1
1	Cow bullying					
2	Corruption					
3	Supply chain finance					
4	Low barriers to entry and exist					
5	Industry norm 'work first get paid later"					

S/N	Technical -related causes	5	4	3	2	1
1	Defective works					
2	Delay in approval of works					
3	Errors in submitting claims and valuation work					
4	Insufficient documentation and information					
S/N	Regulatory -related causes	5	4	3	2	1
1	Complex payment legislation					
2	Ambiguities within standard forms of payment provisions					
3	Multi-tiered hierarchical structure of contractual framework					
4	Complex payment legislation					
5	Widespread use of adhesion contracts 'take or leave it'					
S/N	Other -related causes	5	4	3	2	1
1	Lack of transparency					
2	Exploiting fierce competition among					
	subcontractors					
3	Undue commercial interest					
4	Administration inefficiencies					
5	Cashflow strategy					

22. From the list below; please rate effects of unfair payments on your projects -on scale of1 to 5, (with 1= lowest and 5= highest):

EFFECTS	1	2	3	4	5
Creates cash flow difficulties					
Diminishes organizational reputation					
Affects the administration and management of					
company					
Results bullying to construction firms					
Increases credit crunches					
Result to liquidation or bankruptcy of a project					

Section B; General Information of Participants

- A. Please indicate the number of years you have been working in the construction industry.
 - 1-5 years 6-10 years 11-15 years 16-20 years 21-25 years
 - More than 25 years
- B. Please indicate your profession al designatory
 - Project Manager Quantity Surveyor Architect Engineer Contractor Subcontractor Client
- C. Please indicate nature of business your construction company is involved in; Residential building Commercial building Property development
 - Infrastructure Heavy construction and others
- D. Please indicate the very approximately the type of contract that your company is involved with;
 - JCT forms
 - FIDIC forms
 - NEC forms
 - o Other Standard forms of contracts
 - Bespoke forms
- E. If your member of profession body, please indicate the designator letters that you hold; you are free to tick more than one box if appropriate
 - o ACIOB
 - o MCIOB
 - o MRICS
 - o MRIBA
 - o Others

Thanks for filling out this questionnaire

APPENDIX C- Interview

The purpose of the interview is to seek your opinion about unfair payment practices in the construction industry.

Please answer the following questions based on your experience in the context of unfair payments to construction contractors.

- 1. What is your view about chronic unfair payment problem in the UK construction industry?
- 2. What are your thoughts about client's interest concerning unfair payment issues? Do you think they are sympathetic or deliberate?
- 3. Do you believe that clients are indifference to unfair payments practices in construction?
- 4. How will you describe clients' attitude to unfair payments problems in the construction industry?
- 5. Most literatures suggest that current payments practices in the construction industry (such as payment provision in standard forms of contractors, project bank account and other administrative measures) are ineffective in resolving unfair payments practices in the construction industry. Do you agree?
- Certainly, unfair payment affects contractor's cash flow. Apart from that, can you identify other aspect of business performance that unfair-payments and its practices affect most? e.g. profit, credit rating, cost of doing business,.etc
- 7. Unfair payments seem to vary between different tiers of construction clients' in the industry. From your experience which category of contractors are worst hit by unfair payment problems?
- 8. Apart from cash flow problems tell me how unfair payment affect your business negatively
- 9. Can you give some examples of how project attributes such as size, scope of works, contract sum, nature of clients, etc influences unfair payment?

APPENDIX D – PARTICIPANT INFORMATION SHEET

You are being invited to take part in research of development of a framework to minimise unfair payment practice in the UK construction industry. Laura Lazaro Peter at Coventry University is leading this research. Before you decide to take part, it is important you understand why the research is being conducted and what it will involve. Please take time to read the following information carefully.

What is the purpose of the study?

The aim(purpose) of the study is to develop a theoretical framework to minimise unfair payment practices to subcontractors and suppliers in the UK construction industry. Recent UK government report identified late payment, "pay when paid" culture, withholding retentions, disparagingly rate of items, imposition of rates, and exclusion of provisional remedy as common unfair payment practices bedevilling the UK construction industry.

The study objectives are:

- 1. To evaluate robustness of payment procedures in various standard forms of contract used in the UK construction industry.
- 2. To determine the scale of unfair payment practices between tiers of construction clients.
- 3. To identify contemporary causes of unfair payment in the UK construction industry.
- 4. To ascertain if unfair payment practices influence contractors' business performance.
- 5. To develop a theoretical framework that will help minimise unfair payment practices in the UK construction industry.

Why have I been chosen to take part?

You are invited to participate in this study because you have experience of industry's payment practices.

What are the benefits of taking part?

By sharing your experiences with us, you will be helping Laura Lazaro Peter and Coventry University to better understand the payment practices in the UK construction sector.

Are there any risks associated with taking part?

This study has been reviewed and approved through Coventry University's formal research ethics procedure. There are no significant risks associated with participation.

Do I have to take part?

No – it is entirely up to you. If you do decide to take part, please keep this Information Sheet and complete the Informed Consent Form to show that you understand your rights in relation to the research, and that you are happy to participate. Please note down your participant number (which is on the Consent Form) and provide this to the lead researcher if you seek to withdraw from the study at a later date. You are free to withdraw your information from the project data set at any time. You should note that your data may be used in the production of formal research outputs (e.g. journal articles, conference papers, theses and reports) prior to this date and so you are advised to contact the university at the earliest opportunity should you wish to withdraw from the study. To withdraw, please contact the lead researcher (contact details are provided below).Please also contact the Research Support Office research.eec@coventry.ac.uk and +44 (0)24 77658278. You do not need to give a reason. A decision to withdraw, or not to take part, will not affect you in any way.

What will happen if I decide to take part?

You will be asked a number of questions regarding construction industry and payment practices, terms, performance, etc. The questionnaire or interview will take place in a safe environment at a time that is convenient to you. Ideally, we would like to audio record your responses (and will require your consent for this), so the location should be in a fairly quiet area. The questionnaire or interview should take around 15- 30 minutes to complete.

Data Protection and Confidentiality

Your data will be processed in accordance with the General Data Protection Regulation 2016 (GDPR) and the Data Protection Act 2018. All information collected about you will be kept strictly confidential. Unless they are fully anonymised in our records, your data will be referred to by a unique participant number rather than by name. If you consent to being audio recorded, all recordings will be destroyed once they have been transcribed. Your data will only be viewed by the researcher/research team. All electronic data will be stored on a password-protected computer file. All paper records will be stored in a locked filing cabinet. Your consent information will be kept separately from your responses in order to minimise risk in the event of a data breach. The lead researcher will take responsibility for data destruction and all collected data will be destroyed on or before

Data Protection Rights

Coventry University is a Data Controller for the information you provide. You have the right to access information held about you. Your right of access can be exercised in accordance with the General Data Protection Regulation and the Data Protection Act 2018. You also have other rights including rights of correction, erasure, objection, and data portability. For more details, including the right to lodge a complaint with the Information Commissioner's Office, please visit <u>www.ico.org.uk.</u> Questions, comments and requests about your personal data can also be sent to the University Data Protection Officer - enquiry.iqu@coventry.ac.uk

What will happen with the results of this study?

The results of this study may be summarised in published articles, reports and presentations. Quotes or key findings will always be made anonymous in any formal outputs unless we have your prior and explicit written permission to attribute them to you by name.

Making a Complaint

If you are unhappy with any aspect of this research, please first contact the lead researcher or If you still have concerns and wish to make a formal complaint, please write to director os studies as seen below:

Dr Andrew Arewa

Director of Studies (DOS)

Faculty of Engineering & Computing John Laing Building

Room 133

Tel: 02477657710

E-mail: andrew.arewa@coventry.ac.uk

Laura Lazaro Peter Doctoral Researcher Coventry University Coventry CV1 5FB

Email: peterl@uni.coventry.ac.uk

In your letter please provide information about the research project, specify the name of the researcher and detail the nature of your complaint

List of large (main) contractors' business performance in the UK construction industry $% \left({{{\rm{D}}_{{\rm{D}}}}_{{\rm{D}}}} \right)$

-	_																																															_						_
														kezp protep protep protep															Aesp postp 021ote													Acqu protap urvetada (b Sec)												kezp protop dörsöfe
L			25		20		25		}	35		35		96		25			8	- 1	2	5		1	35		1	5	1945		35			10	1	35		25		1	5	60		25		22			25		25		25	trains
1	/k	later	Openiquet	kar b	oer Gestigst	kse	later Open	igget koe	lorer	beinget	kar la	ver iheri	igpt koe		(protopol)	Subjected	ine 1	braigent	latificted.	ine 0	ziget i	anteriot	ior 0	stigat 3	anticipt	Roar 0	prippt a	nhäistet ko		intes	Intille	kar b	status	1281 Int	Grecket	ntials 4	ine fare	ka jesti	in kar	inclusions	izite kor		isians.	iskas k	or isi	elie isika	a koar	al site	isks i	or Talab	isks ir	a folate	laks ker	
Г	(Company)	243	85	17 0	145	331	3627 1	18	3819	15	52 :	5 3	14 13	318	Xi	71.1	1.576	8	10	25	5	Đ	100	34	971	185	34	972 198	55 58	223	10	1ESSW	SE1 1	84 1338	8 221	2 15	358 2	27 28	10601	921 I	25 21800	10	23	38 15	846 5	5 22	19235	20	20 15	20 8	18 18P	25 5	5 13563	14
r	1 Campany 1	85	15	13 6	ES 4	1.5	115	3 43	310	50	164 1	95 2	5 29	510	8	15	LOEE	2	27	12	9	5	SEE	5	31	12455	5 }	11 156	5 53	9	S	12987	5 }	6 (13866)	6 Z	3 15	100	1 1	36782	3 1	1 15	65	1	8 13	162 I	3 1 5	1884	0	a 33	534 35	8 10	12 21	SK LEEF	13
ĥ	i Campany 1	3821	E1	15 3	K9 123	01	1 1361	12 53	151	103	63 1	873 N	EI 50	50	ES	83	15.235	55	315	10	£1	25	38	18	Ξē	554	81	22 22	5 29	365	35	1879-67	22 4	62 11250	4 85	481 11	335 3	E 193	17818	355 3	E 1108	1 15	23	38 13	168 E	3 35	1285	題	34 32	BE 191	351 127		25 17555	15
ĥ	(Canpany 4	1281	285	68 3	45 93	18	1245	14 23	21	<u>1</u>	136 2	21 5	ES 29	12	26	36	he!	10	255	10	EI	H.I	115	54	361	1111	25	824 IS	5 33	211	EI	15853	90 1	ki lime	1 52	E 11	EUE E	54 181	10054	95 j	1 1975	5 12	21	254 15	55 1	E 192	1830	933	285 112	935 ISI	32 137	164 17U	58 17507	B
ĥ	:Conpays	18268	KINE	10 17	827 3821	18	32145 1	80 19	19213	485	28 19	828 39	R2 18	310	23	7£1	12.575	15	342	135		389	53	35	524	113	25	EER (S	R 39	킨	3	10536	E	SK 1982	3 1	E 13	1000	1 2	19355	-EE 1	SE LINES	10	885	399 15	5355 33	8 99	1582	TH	45 15	es e	43 187	35 32	8 1875	E
F	(Company)	120	134	64 1	8 25	171	100	10 112	1925	92	53 1	109 19	19	tø.	憲	596	1565	265	861	25	22	8235	978	93	36.95	330	38	9878 118	62 25	15	360	11964	965 3	(81 LEE)	EN	867 12	XB U	2E	198365	0.0	E INS	63	35	28 12	101 D	4 31E	15398	30	181 11	98 30	30 137	125 103	RE LIGHT	18
r	1 Damping 1	\$107	355	68 <u>9</u>	85 155	140	958 1	70 29	200	201	12 3	251 8	E 19	327	Ξ	16	1507	SS	30	25	35	3	15.38	E	3	1.SE	=	E3 18	3 83	12	3	LEATIN	=	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	E 24	34 15	19967 1	E 35	ISSE.	5 1	1 1000	< L5	35	25 3	103	1 52	INSTE	11	157 15	38 99	28 15	45 25	58 1987	11
r	i Campoy I	10.000	5545	1541 314	18 3517	1131	2438 14	87 63	19923	2/571	19 15	598 53	103 115	18	255	EI	1307	21	305	12	9	35	12.55	36	3825	121	E]	305 12	3 35	25	22	182723	15 1	50 1068	5	383 10	10H S	3 33	1850	23 E	10 10752	4 68	335	423 15	560 II	2 304	122575	100	12 H	165 235	E25 187	HE 36	X75 18867	11
r	1 Campany 9	13780	9583	ER 19	688 1596	18	FEE 7	72 15	13/19	34811	134 13	265 38	18 15	138	122	22	51.356	XE	350	8E	121	3EE	15	28	1575	182	36	198 518	E 38	220	1991	LETER	22 1	DE LEDE	i liei	302 19	BE 3	15	ISBEE	92 3	100	2 (3	4EE	522 13	136 38	6 615	1.683	138	355 (1	96 FE	293 142	12 18	570 (450)	B
Г	1 Carpany 11	180/35	230	201 344	113 2825	147 1	45% I	81 25	183	134	10 12	785 XI	22 27	210	251	28	5 6 6 7	129	33	135	3	33	100	305	953	183	168	90 11	5 22	525	269	14SOE	238 2	RE 1585	888	SRS 14	RES 1	91 55	12568	83 1	1000	68	78	R 18	58 5	8 8	1953	10	66 (E	738 124	363 155	88 EB	100 10001	H
Г	1 Serpeny 11	3817	£4	14 2	15 43	135	IBJ I	3 12	293	35	1134 3	53 E	E 173	52	61	261	1382	ei i	10	21	10	367	586	15	365	1.003	2	76375 (LB)	72 25	1300	8300	153974	500 8	00 1150	1900	esen 18	100	10 100	11867	6000 (12	10.67	18	951	258 13	1875 8	2 25	LNEE	ъ	15 0	E3 33	121 137	671 TH	18 1189	15
Ľ	1 Campany 12	267	3085	28 S	18 115	15	984 9	48 48	1538	9171	1127 14	E 28	12 28	64	115	294	3135	25	98	13	90	191	SBS	20	21	157	22	96 23	3 11	534	165	18235	BE 1	12345	5 57E	907 18	EBE 2	22 383	19856	83 8	1081	i 11	925	151 IS	E36 11	3 311	1986	38	6E 11	253 325	5782 1857	25 43	SE LEES	E
Ľ	1 Campany 11	101	2501	1134 2	872 180	585	78001 8	20 43	5530	\$70	43 15	870 29	200 2.27	55	30	4015	Sisten	23	80	25	ED	SE	200	\$11	HSI	258	ED	301 13	15 53	19E	360	16351	100 3	E 1367	1450	351 54	- H	21 49	135265	381 4	10 (1276)	13	35	33 13	38. 3	R BR	1.18255	51	BS 11	225 538	IEI 19	165 863	BRI URB	12
Ľ	3 Campany 14	1895	553	14 11	QE 1346	1288	10211 5	40 SF	2575	-	15 11	295 275	20 12	ਿ	33	25	1.665	23	127	22	36	361	430	81	38	382	757	1741 1121	12 12	3238	438	196362	æ []	ES 1056	200	352 13	36 3	83 JBJ	Deep	365 3	15 (1888	15	580	881 13	792 II	50 1 60	1JEDE	5586	300 0	865 HH	130	134 610	971 18221	5
E	S Campony 15	199675	48	18 8	37% 14830	810	1891 3	EI 18	SEES	435	28 1	BQ 3	D4 112	58	85	254	SHEES.	785	278	545	1991	33		6	1755	い沢	15	38 39	12 12	165EU	4060	(ESE) I	900 (0	80 353	1921	4600 19	5267 BB	101	120348	6722 38	511 12034	15	929	90H 15	88 8	is est	LTGETE	SEE .	HR 13	825 636	98 197	185 2.2	201 100700	19
L	S Campony Tri	14570	430	18 8	410	180	5370 3	HE 222	1932	301	234 11	180 25	80 28	28	80	121	1956	3EI	50	122	30	51	SH	361	251	13E	30	181 2.2	5 22	631	330	18505	30 (i	1758	2280	450 15	800 8	a 33	11056	381 3	1957	13	HES	651 1	ER 8	5 50	12256	88	25 (il	558 ES	88 15	125 233	85 1561	13
L	3 Carpon 17	1981	250	12 21	180 (230	58 1	3000 2	III 43	5530	570	43 1	50 j 2	22	12	30	400	ines .		81	=		SE	200	31	351	158		30 13	5 55	1500		1636		DD (3677	1 1000	257000 54			I SERIE	3000 10		8	335	254 II	ISE G	8 50	1627		38. 15	55 83		25 382	580 100704	
L	3 Campany 11	SERVER I	37500	15 901	8400 287201	217 9	21000 31	800 137	18 (BOD	19200	26 83	100 385	5700 155	137	1710	20701	3.55	28	391	122		9931		121	3541	чœ	360 (SERIE SE	E 28	1611	2290	1356574 2		30 18CE	12980	SGU 13	1815 ET	an (1997	1236	27381 58	EE \$168	212	285	251 15	35 5	U 190	19957	=	276 22	96 (M	100	10 10	SE 1708	
L	3 Campany 19	557276	18281	13 (2)	578581	623 9	1678 23	188 55	181564	40595	48 12	221	105 23	5:0	-	EES.	1000	ESI	4234	15	100	3875	25	375	SUET	156	25	JBH 19	18	336	655	LENEE		28 1888	12345	B428 15	40 0	65 055	1 1000	308 3	50 (1963	13	435	\$FEIE 13	772 OF	10 1007	113979	(ESE	161	81	BER IN	16 855	1 BEER (1977)	Б
L	3 Carpoy 31	TELNO	30100	19 128	1401 1219001	150 1	29801 27	100 223	12018	1750	18 28	730 258	500 170	19	300	260	1987	38	32570	=		:780	538	:78	398	STEE	2500	SIEI 28	5 53	673311	3220		60 (S	1365	SEDAD	43500 13	1910 (A	80 3570	L SEEK	BUSDI 35	1456	13	XES	ER IS	325 B	H 200	13256	359	1000	RE 25	100	158 XISO	19300 14561	в
E	3 2010132	\$48000	15000	19 93	13000	14 1	13000 20	100 220	153000	13000	13 87	1000 670	1001 0.75	14	3200	2580	1536	INU	1003	22	100	1002	1923	200	5100	1225	900	2001 23	8 25	600	33200	120005	SIII (2	200 13599	2 36200	8200 13	BEF E	10 720	1 39557	EZIII 18	183	15	200	27000 12	SE E	00	1 18576	36000	3500 (C	267 2680	155	25 2:00	200 19809	19
Т	Three D	10,0000	TERM	12	and thene	128 0	10000	ten cat	S rargent	10400	10.00	con i sac	man ten	525	1000	10711	berra i	con 1	1000	had -	1 mm	100011	ingen 1	CHINE .	10070	barn	£	TON IST	10.00	2.02	100 3	100000	en 1 1	m lano	1 100	100 10	ne:1 a	a 1 m	INCOM.	2001 1	n kurn	10	10/201	2021 12	-	and server	diam.	Longer L.	1000 \$12	DD MOTO	Stern Lar	and spec	Correct Service	12

Appendix F List of small and medium sized contractors (SME's) business performance in the UK construction industry

			ange ander					hesp protop					anab Matab						Arap postap
		94	digati (2010					45 M					faction 1						Administra Security
							21										215	201	
Stal Destigation Loss Destigation	kase linner Base linner Anse	r Tanaw Gentleput Anner	Oprolog polit Havindentinis Anwer	opelegent lookaintais sow	Opening politic investigation in Associ	Specific Progetting	Opening polition killer Sale Asse		instantinity low	traction practicities more	intististis as interior	ally now treated precision	tor tota	alle takes are take	uniter factors have	talaite talan ine	talaitie taitus ine	tolidite tollars a	
		1																	
20mmert 272000 24000 140 250725 125071	38 SECT 2018 19 00784 1007 420	274275 8569 25	22 200 6522 25	580. E28 25	INSE 1540 13	20271 6368E 343	1 E565 9082 5:1		858136 (78725) EH	10x70x 982088 138	1943KD 88303 02 940273 10	1207 121 1079677 822034	12 11 200	00 37100 645 372	265 38528 18	1290 1226 14	2008 3876 18	1 2040 122765	2 18
Support Control of the Article State	12 10200 1000 10 10200 0000 10	POSCOL MADE LC		3703 2640 2.6	9426 22628 3.3	200 500 75	1 1621 3601 23	33	amore there are	204000 208000 125	284001 282001 18 295000 210	200 12 92200 40900	12 12 12		12 10 10 10 10 10 10 10 10 10 10 10 10 10	200 000 12	2028 80201 04	oom letter	2 02
Arrest many came in course and	an amount and an amount of	Ampen come to			000 000 10				armed coarry 1 and	and a second to the	Annual Annual on Annual an	the second second		The second state of the	and share and		solution and and	and and	in anima
Sheeped control opents of support spaces	the formation and the formation of the	10020 0000 000 000	10 100 100 20	1000 1000 011	100 1000 10	000 1000 10	100 BES 10	1.1	and the second second	THEN THEN IN	and the second of another and	201 10 27420 38700	12 18 20	and another that the	ALC: 10100 100	cone carta in	NUMBER OF STREET	Trans areas	D AX
damanya 222007 22103 131 210703 147823	48 282549 58278 428 222828 5672 428	1738417 20763 13 3	10 10:00 12:00 14	1020 1020 112	3678 6568 2.6	1075 10761 10	333 1925 55	1 25	108207 1070281 1.11	110815 100046 610	10120 10001 12 BRKD 03	224 121 34226 12156	(1) (1) (2)	100 2200 LTD 122	11 9692 415	FEED 125861 EV	NEED BEACT OF	82286 98855	3 0.3
\$20000 20000 20000 10 070000 20000	18 TRENDE IZ2000 4/3 SEARED 5/1000 4/8	Sarace 20000 200	Market and	CORR GREET X.0	£301 G600 2.5	schild annun su	1 1000 1000 20	2 52	25-CODE 14000001 0.00	742000 882000 647	140300 017000 0.0 963000 210	100 10 549000 018000	LAS EE THE	and many of the	200 20000 16	NORM CONTRACTOR	30000 074000 02	hanne prisent	2 08
Singers (23800 124001 5.31 08000 36000	10.81 0732001 48200 5.51 7220801 208001 2.76	44ET01 10000 7.31 8	19 2000 30200 (5	NEE 2000 10	4601 18000 14	2000 58200 1.8	3000 3000 200	15	2/8800 (20100 147	455801 \$34800 £79	1011001 201000 120 307000 34	200 10 22000 970700	601 1.00 SE	LEE alien Les 307	NUE LINUE UZ	7020 4820 13	JANZANE ANDANE DAT	INCOM ANTER 1	2 19
Samped 273800 2000 10 20400 2040	13 27070 26700 12 2702700 12000 1.0	Exect 30766 2.20	13 1000 1000 13	2000 2000 40	3628 6628 55	5140 67403 L3	30700 40900 52	4.0	500908 78H800 1.16	EE2001 578200 1.43	12555 10255 12 102765 23	DDC 140 373400 547800	14 12 25	im signe can as	10 1001	1000 1000 00	HARE THEN IT	Sector Sectors	2 19
150mpay-0627407000 240000 8.00 2740300 224500	17 22340 12100 12 22600 9800 4.0	2002000 2000 10	15 2020 1220 22	202 1092 125	\$200 SSND 1.5	1000 10000 60	268 2698 18	12	NCB00 880800 122	\$7,2201 90500 121	102200 10100 13 5000 20	200 125 32900 585400	121 115 240	an and an an	1011 10111 101	SOUR SHEET DO	XXXX 20080 100	\$7988 25788 I	3 16
152mpay (* 60427 38042 487 94509 38221	42 50/207 20620 422 40/082 23620 420	3454 0181 17 4	CX 18:0 226 2X	252 852 2.5	30821 30825 2:6	265 655 88	228 528 25		100 000 1000220 116	SZERN SHEEK 112	18750 (2001 19 0008 0	E29 119 DE270 10288	17 12 20	102 2022 102 50	683 2016 10	262 255 15	19638 (1961) 16	226 555	£ 18
CRAMPY D 322255 18252 234 40522 2688	725 45828 25077 538 457822 (2688 140	24020 5034 227	CI 180 081 18	268 862 83	241 2484 242	SRSE 44271 2.3	1 1221 1280 52	33	57.0KD 209.50 144	100640 10522 CH	162627 127667 0.0 279700 10	227 141 50420 25521	12 12 20	GR F100 10 X0	1993 1993	\$250 2560 12	1974 9160 18	2083 8621	R 13
Charge C Belle 4001 1 11 10700 F100	43 94000 1500 15 94900 1000 10	\$728801 727488 \$ 450 4	CO 600 100 123	\$200 DOD 3.5	FUE SCORE 13	2008 2008 19	THE SHOE IS		20000 (254800) 0.09	207001 5 00	Server Second on Access 31	100 00 2000 2000		III IIII 123 (23	NE GEE DE	CHIE 1000 13	JARDE JOHNE 13	TORN STOR	
States and the second states	10 0000 10 0000 1000 100	20000 2000 3 CD	388 388 123	2000 2000 220		1 MERE 134000 13	2000 120000 10	-	TANK PARAMETERS	270000 100000 000	Property and the property of	THE DE STREET		100 1000 100 100	and other in	acting carries and	CORDER JORDE DE	John Lance	
Charge of Science and Low Science and Science	un Amou Amou 117 Second Summer 419	Strang and Strang	100 SALES 14	1000 SHEE 51	1000 SHED 11	DEED SOUTH IN	2500 Add Ad	33	COMPANY AND ANY	ACCRED DEPENDING LESS	Addition and and an array of a	AND AN ADDRESS ADDRESS	11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	and store of the state	COLUMN IN	NAME AND DO	STOR SHEET OF	The later	3 016
Changed and have been been over	en temper any landername enter au	10000 010 110 0	14 140 100 10	22 22 22	202 202 21	105 665 15	200 200 10		SALARY AND A VIE	58801 JOSOF 118	CONTRACTOR AND ADDRESS OF ADDRESS ADDR	D4 17 1038 1020	10 10 10	100 2000 CH 100	NO. 2021 13	NUMBER OF STREET, STRE	BADS OTHER 13	THE PARTY	
100mm-0 201 175 4.5 000 175	C 17 30 10 90 C 30	0 1 10	12 (5 (2) 14	10 60 53	30 30 10	4 9 19	2 34 42	55	491 932 112	100 100 107	47 10 17 20	10 10 100 004	13 1 18 99	and the last	EN ANY 13		on on 12	1000 (100	E 1%
182man/19 (72868 90286 8.0 (78827 3882)	28 194946 1900 (19 1904T) 4707 LB	100001 2000 10 2	21 164 1931	AND 1993 121	290 690 23	097 698 12	284 1019 12	25	5605 394800 000	17675 29401 620	20292 49607 0.7 47549 49	122 125 CODE 38781	03 00 92	20 9408 642 10	RE 2000 19	1000 (2016 DA	1000 2000 1X	2000 ZHAN	2 06
28anaw30 (10) 114 64 100 20	17 982 50 102 55 80 10	303 30 03	11 28 226 43	26 26 2.0	20 20 61	8 28 68	20 20 10	2.5	221 245 686	205 249 034	216 228 02 288 2	N 58 125 221	(1) (1) (2)	ER NOR DE CE	SED 5805 13	0521 6220 04	CRED WHEE DE	2002 6702	19 19
280mpmp20 5007 302 620 5535 5536	341 5561 1671 230 2008 34x 7.2	1021 121 131	42 25 236 23	255 870 240	101 1008 101	201 43 23	B 35 25	2 3.6	2238 1092 007	1076 3870 000	124 3869 112 122 3	ER EN 275 2759	601 BED 500	de 2270s 12 28	tal digit tal	OWNER ADDRESS LED	10000 10200 102	1983	3 03
190mpag2517022 57675 5.4 24162 37970	111 22908 14027 128 18927 2371 7.6	1122303 13401 125	LT 1925 1386 2.0	1223 1225 115	2252 1252 6.3	252 386 53	1 2901 25781 52		10107 10000 08	27123 83348 681	2828 2002 00 12875 24	105 154 12821 22530	125 128 168	EB 24290 16 131	758 56398 122	MARC 2022 10	38670 182581 184	1 12764 128256 1	2 0.6
282mpmy22 25700 2640 154 154000 1586	12 17223 775 12 2276 3421 12	12230 200 10 1	11 88 223 15	286 228 52	259 2696 63	2001 2000 24	1 203 SHE X:1	: 10	6280 12874 648	10120 12046 651	G483 (3923 04 7050 13	23 12 10754 2125	CH LS 30	10 222 12 12	27m 2705 1.42	323 636 13	JHR IST IS	20200 20200 1	2 62
301anang36 56502 2008 301 96102 2029	247 SERVE 2008 115 SERVE 128 128	127826 3675 252 3	16 Sec 255 45	363 HEI 34	282 255 43	28 252 13	105 845 23	1.1	17888 189875 611	12027 2005 143	825 4810 12 822 3	DC 121 1775 4050	620 640 200	NAK SATS IN IN	KER KER LER	202 666 55	HOLE AND DR	INSU DOS I	8 18
250mmy2529694 2808 154 26085 2826	15 19233 4256 27 197424 1969 120	121825 1001 135	22 223 224 23	305 865 86	658 6363 62	383 873 25	SR: 325 21	2.6	52542 65082 68	7545 7545 130	3277 3297 18 12854 10	10 10 101 101	130 235 230	66 8200 18 76	66: 6807 LB	2923 2656 20	252 202 28	1 605 903 I	8 00
342mparg/35 900 01 2000 1.0 90000 (1945	128 87280 6569 423 87829 38280 4.3	1 EMITE 1924 100 7	14 25E 650 13	2005 2000 614	259 257 25	1 MAR 200 MA	1 282 646 23	ER.	229/20 678225 628	NOTH 422817 625	8465 27390 040 2026 12	122 EH 386E 9424	63 EQ 154	EI 508 12 12	2001 20057 123	366 388 10	OHE ESE 13	096 293	2 15
752mmg/2 2020E 1271E 127 276718 20204	731 2739542 10731 148 198576 12382 1.54	19854 0285 63 5	81 224 242	258: 286 28	250 2565 G	206 206 18	226 255 (3		22828 34562 694	20201 22015 012	2004 2058 00 12280 13	10 18 0250 1705	682 6 81 676	ME 2000 12 35	102 2028 135	650 355 E	466a (2278) 62	1 1016 19803	2 13
352mpmy35 22576 20076 3.71 355600 40541	158 229615 25578 1587 1592787 159673 3240	2 212 2121 2121 212	SEC SER 33	6361 37562 5.9	HIN Risk \$12	1260 2351 24	1 153 2450 240	: 3:3	10830 122856 037	EDDE SCUL CH	43102 52000 02 20045 27	65 IN 2394 2625	CH 20 1	15 163 1F 15	A25 2856 184	261 1228 16	286 288 05	629 (56)	A 08
262mag/3(2026 0021 14 2021 2426	12 30% 408 14 2120 302 13	120000 HINENS 214	2 april 12548 16	3256 5823 12	686 6856 52	800 800 23	2021 2026 13		40710 (10754) CH	HET BACH CH	SCOL ACCEL OF SCOVERS	224 22 61942 2100		10 100 10 10	SIS SHEE LE	102 ALL 10	3090 5054 1 08	1786 12788	
1220042 26 23 18 10 10	10 1010 220 23 400 244 43	4621 200 128	10 10 10	34 20 10	25 253 5	20 101 00	346 360 169	6 63	140 120 120	12 28 126	2 2 19 19 19	10 13 1990 1990	10 10 20		NOL APPER LUX	THE THEFT IN	CORD ARMON CON	258a 192a	4 14
There is a series of the serie	THE PARTY AND THE PARTY AND THE	There are a series of the seri	THE SECTION AND ADDRESS OF ADDRES	AND 1980 10	200 200 23	000 000 10	260 805 12	12	Den Denn Cor	13913 279981 626	2020 4001 0.1 41545 40	DP ID GID JUNE	104 1 100 1 Ho		100 0000 100	and some in	and and in	The later	* 08
Three to any training the second state	The Amount of the The Topology and the	Annes Tanna Law				1000 0000 00			con long long	cost cost cost	Contract of the second of the	THE SHARE SHARE	A 14 1 14 1 14 14	THE ADDRESS 14. 1	and share 1.15	starts a character and	print among the	Trans I store	in anima
Manageria and the second secon	All sectors succession and another sector and	STUDE LINES CN	AN IN THE LA	10 10 11	108 208 C	286 2288 23	3843 3203 224	2.3	1949 1920 036	1296 1204 036	ATURY ANTER OF TAXABLE OF	ALT DE STORES TOURS	10 10 10	100 1000 10 10	101 6036 128	terms total tot	1000 1001 10	10000 11000	8 68
These of these states in the states where	The second them in the second state of the	2 Taure 1 1000 100		107 100 10	1005 1 2005 1 20	1000 0000 10	555 000 12		contain a statement of state	THERE STOLET AND	TTON ANTON ON ATTACAN	10 10 100 100 1000		10 mort 15 1 50	art 100 110	1000 AUGUST 110	STR. 1008 1.05	state 1 motor	
30mm/3 157 0 10 975 35	28 082 00 122 000 10 120	1 67 2 13	48 21 26 13	14 25 14	31 576 16	21 22 13	5 15 12		GR 85 62	276 129 146	225 215 236 337 3	10 202 202	125 238 120	10 2000 16 M	ALC: 1997	1710 SED 12	1000 1000 100	000 1000	2 15
These 3 10070 (0000 (0) (7500 2002	108 10822 2070 110 19628 1082 102	1000 1020 22	10 2026 20270 63	2022 2027 2.8	2951 2955 6.3	2005 0002 34	20 536 10	20	22772 43409 625	10103 01438 025	2871 47081 1.9 228 22	10 10 2101 2304	C10 E11 900	N2 2007 12 (R	852 2099 140	2002 12220 122	2012 20167 10	266 262	2 12
200mpay36 04542 20000 517 010020 92010	241 20220 48222 612 288562 20827 125	100234 10021 110 7	15 348 400 LJ	10 100 100 100 100 100 100 100 100 100	4055 32355 NB	2867 6523 22	220 285 55	1 23	23055 90790 625	2234 2223 66	#1250 002070 010 947854 01	100 100 5280 10071	12 LN 20	138 2352 12 27	7527 26376 LBL	3390 38678 DR	20020 20001 00	37821 2486	£ 0.8
202000 10201 10201 10201 10200 102000	NAC TRESS CARDO LAS CARDOS LAS	1 0 0000 0000 000 000 000 000 000 000 0	KX ::SH: 6885 52	2010 CH21 56	880 359 3.5	2876 23825 23	2223 2238 244	1.1	14570 30274 640	COP11 25592 625	7122 28125 028 12228 13	12 13 4322 22256	64 121 25	NC 3306 10 13	20% ISBN 136	752 2825 15	208 205 13	642 3636	3 0.3
40anany4 22014 2007 12 2008 2010	13 27555 9828 43 28272 767 10	1700701 2021 127 4	434 2002/ 22090 615	2003 20025 431	203 2026 13	7960 29676 718	301 2107 252	11	12828 12225 678	7292 1923 68	2542 15679 08 19400 16	GE 173 17213 12487	00 13 10	10. 100 10 10	621 FBX 142	604 982 16	2007 1000 107	208 1000	2 05

Appendix G- Quantitative data analysis

DISTORATINE STRAFF1	 Unitiv payment is a persistence problem to Construction contractors 	o Urfai paymentisramp at bezu se d''n cann d'' pay 2 abit u de of clients	2 Unfair payment exists due to "1 would not" pay asst under of clients.	Clerts strategic plan to boost cain flow excertate unfair payment problems.	Parment provisions in £ ander diamis of contracts (SoCi) helps to minimize unfair payment occumence	D SFOCs(LCT/NEC/FIDIC) have sufficient mechanisms to a protect contractory from unifair payment problem	2. 28 days payment period signalesed by various SPoCS 4. helpt ominimize unfair payment problem	D Payment procedures & pulated within SFo Garenot. 2 in counterrough tominimise unfair payment.	Untractions are retruction use pensations and unace in D regulations, SFoC and child resson membing unfair 2 reamment because of fear that it will damage their constructions of the structure of the struct	business relationship Durfair payment problems writesbetween bies of contractors	D Unfair payment problems are frequent among Tier 1 contractor (j.e. dirent to main contractor)	2 Utfair payment problems ar eff aquent among The 2 5 contractor (J. e. mán contractor to sub on tractors)	b Unfair payment producers are fraguent among Tier 3 2 contractor (i.e. subcontract ors to other 2 subcontractors	2 Unfair payment influences over all contractors? In ancial performance.	2 Urfair payments sprificently infuences contractors to profit ability	2 Unfair payment is a thread to contractor of business survival.	2 Urfair parmentós notsignificantlyinfluence contractor profitability.	D Emerging digital technologies have potternial sto. Eminimi sel alto poyments scorrip ared to o'r diffornal payment procedur esint he construction in du stry	Sum of response sfrom study participants
Q from P1 Q from P2	5	5	3 4	5	2	2	2	4	5	5	3	5	3	5	5	4	0	3	66 56
Q from P3 Q from P4	5	3	4	5	1	1	2	5	5	4	3	5	2	5	5	4	0	0 2	59 46
Q from P5 Q from P6	4	2	2	-	3	3	3	2	5	5	2	4	2	4	4	4	2	3	57 44
Q from P7 Q from P8	3	2 5	2	2	4	4	2 5	2	5	5	1	4	1 2	4	3	3	2	1 3	50 65
Q from P9 Q from P10	2	2	2	2	3	3	3	2	3	3	2	4	2	4	3	3	2	1	42
Q from P11 Q from P12 Q from P13	4	2 4	2	4	4	4	4	2	4	4	2	3	2	4	4	4	2	1	56
Q from P14 Q from P15	2 5	2 2	- 3	2 4	3	3	3	1 2	3	3	- 3	3	- 4	3	3	3	2	- 2	36 50
Q from P16 Q from P17	4	3	3	3	3	2	2	2	3	3	3	4	3	3	3	3	2		49 51
Q from P18 Q from P19	3	2	2	-	3	2	3	2	5	4	3	2	-	3	3	2	2	2	45
Q from P20 Q from P21 Q from P22	5	3	3	5	2	2	2	3	5	5	4	4	4	4	3	4	2	2	61 54
Q from P23 Q from P24	5 4	5	3	5 4	2 4	2 4	2 4	4 2	5	5 4	4 2	2	3	5 4	4	5 4	2	-	63 58
Q from P25 Q from P26	5	4	5	5	1 2	1 2	2	4	5	5	2	4	3	3 4	3	5	1 2	- 1	58 49
Q from P27 Q from P28	4	2	4	2	2	1 2	2	3	4	4	4	2	4	5	4	4	1	1 3	53 64
Q from P29 Q from P30	5	3	4	5	1	2	2	5	5	4	3	4	2	4	5	4	2	1	55 60
Q from P31 Q from P32 Q from P33	4	3	4	5	2	2	2	3	5	4	2	5	2 4	4	3	3	1	0	54
Q from P34 Q from P35	5	5	3	5	2	2	2	4	5	5	3	2	4 3	5	5	4	0	3	64 52
Q from P36 Q from P37	5	3	4	5	1 2	1 2	2	5	5	4	3	5	3	4	5	4	0	3	62 62
Q from P38 Q from P39	3	5	4	3	2	2	1 2	4	4	5	3	4	2	4	3	4	1	3	48
Q from P41 Q from P42	3 5 2	3	4	3 5 2	1	1	2	5	4	4	3 2	5	2	4 5 2	3 5 3	4	2	3	62 36
Q from P43 Q from P44	3 4	1	2	3	3	3	2	2	3	4	3	2	3	3	2	3	2	1	41 52
Q from P45 Q from P46	3	2	4	1 3	2	2	2	2	4	4	3	4	1	5	4	4	1 2	2	49
Q from P47 Q from P48 Q from P49	4 5	2 4	2 5 7	4 5 7	4	4	4	2 4	4	4 5	2	4	3	4	4	4 5	2		59 59 57
Q from P50 Q from P51	5	5	3	5	2	2	2	4	5	3 4	3	1 5	3	4	4	5	0	3	59
Q from P52 Q from P53	5	3	5	5	2	2	2	3 2	5	4	4	1	4	3	3	3	2	3	59 35
Q from P54 Q from P55	3	2	4	4	2	2	2	3	5	3	4	3	4	3	3	3	2	3	50
Q from P56 Q from P57 Q from P58	4 5 3	3	4 5 5	5	4	2 2 2	3 2 2	3	4	2 4 3	3	4	2 3	3	3	3	2 2 2	1	55 56 48
Q from P59 Q from P60	3	2	4	4	2	1 2	1 2	4	4	5	5	5	0	3	3	4	1	1	52
Q from P61 Q from P62	4	4	1	4	3	2	2	3	3	4	3	4	3	4	4	4	2	2	56 57
Q from P63 Q from P64	4	4	2	4	2	2	2	4	3	4	4	4	4 3	4	4	4	0	1 2	56 54
Q from P65 Q from P66	2	2	2	3	3	3	3	3	4	3	2	4	2	3	3	3	2	2	46
Q from P68 Q from P68	4 5 4	4	5	5	1	1	2	4	5	5	2	4	3	3	3	5	1 2	1	58
Q from P70 Q from P71	3	2	4	1 3	2	2	2	3	4	4	3	4		5	4	4 2	1 2	3	51 45
Q from P72 Q from P73	5	5	3 4	5	2	2	2	4 5	5	5	4	2	4	5	5	4	0	2	64 56
Q from P74 Q from P75	5	3	4	5	1 4	1 4	2	5	5	4	3	5	2	5	5	4	2	3	62 58
Q from P76 Q from P77 Q from P78	3	2	2	2	3	1 3	2 3	2	4	4	2	2	3	3	2	3	2	4	58 47 51
Q from P79 Q from P80	5	5	3	5	2	2	2	4	5	5	3	5	3	5	5	4	0	3	66 56
Q from P81 Q from P82	5	3	4	5	1 3	1 2	2	5	5	4	3	5	2	5	5	4	0	3	62 52
Q from P83 Q from P84	4	2	2	4	4	4	4	3	4	4	2	3	4 5	4	4 3	4	1		60 59
Q from P85 Q from P86	4	2	5	4	3	3	3	4	4	4	4	4	3	4	4	4	1	0	55
Q from P88 Q from P89	2	2	2	3	1	1	0	3	4	5	2	4	2	3	3	5	2	2	46
Q from P90 Q from P91	4	3	3	3	3	3	3	2	3	3	3	3	3	3	3 4	3	2	3	53 51
Q from P92 Q from P93	2	2	2	3	1	1	0	3	4	5	2	4	2 2	3	3	5	2	2	46 47
Q from P94 Q from P95	4	2	4	1	2	3	3	2	4	4	3	4	-	3	4	4	2	3	53
Q from P97 Q from P98	3	3	3	3	4	2	2	3	3	3	3	3	4	4	2	5	2	3	55 45
Q from P99 Q from P100	2 4	2	2	3 4	1 4	1 4	0	3	4	5	2	4	2 4	3 4	3 4	5	2	2	46 61
Q from P101 Q from P102	5	3	5	5	1	1 2	2	4	5	5	3	4	2	3	3	5	1	- 2	60 48
Q from P103 Q from P104	4	3	3	3	3	3	2	2	3	3	3	3	3	3	3	3	2	3	41 53 50
Q from P105 Q from P106	3	2	2	3	3	2	3	3	5	4	3	2	-	3	2	2	2	0	44
Q from P108 Q from P109	3 5	3	4	3 5	3	2	2	5	4	4	3	5	3 2	4	3	3	2	3	56 62
Q from P110 Q from P111	4	3	-	4	3	1	3	4	3	4	2	3		4	4	3	2	-	39
Q from P112 Q from P113 Q from P114	4 3 4	2	2	4 3 3	3	2 3	2 2 3	3 2 7	3	4	2	3 4	2 2 3	3	2	3	2		46
Q from P115 Q from P116	3	2	4	1	2	2	2	3	4	4	3	4		4	4	4	1 2	3	50 45
Q from P117 Q from P118	2	2	2	3	1	1 2	0	3	4	5	2	4	2	3	3 4	5	2	2	46 58
Q from P119 Q from P120	3 4	3	5	4	2	2	2	3	5	4 3	4	3	3	4	3	4	2	1 0	54 57 45
Q from P121 Q from P122 Q from P123	4	5	2	4 5 5	2	2	2	4	4	3	3	3	- 3	4	4	4	1	0	52
Q from P124 Q from P125	3 5	3	4	3 5	3	2	2	5	4	4	3	5	3	4	3 5	3	2	- 3	56 63
Q from P126 Q from P127 Q from P127	4	4	3	5	2	2	2	4	5	3	2	1	3	4	4	4	1 2		58 39
Q from P128 Q from P129 Q from P130	4 3 2	2 2 2	4 2 2	4 3 3	3	2 3	2	2	3	4	2 3 2	3 2 4	2	3	3	3	2	- 2	48 42 46
Q from P131 Q from P132	4	3	3	3	3	3	3	2	3	3	3	3	3	3	3	3	2	3	53
Q from P133 Q from P134	3	2	2	3	3	2	3	3	5	4	3	2	0	3	3	2	2	0	45 55
Q from P135 Q from P136	5	3	5	5	2	2	2	4	4	4	2	5	2	4	4	4	1	0	58
Q from P137 Q from P138 Q from P139	5 4 5	3	4 5	4 5	1	1	1	4	5	4	2	5	2	4	4	4	2	1	56
Q from P140 Q from P141	5	5	3	5	2	2	2	4 5	5	5 4	3	2	3	5	5	4	0	3	63 56
Q from P142 Q from P143	5 4	3	4	5	1 4	1 2	2	5	5	4	3	5	2	5 4	5	4	1	3	63 55
Q from P144 Q from P145	3	3	5	4	2	2	2	4	5	4	4	3	3	4	3	5	2	1 2	55
Q from P146 Q from P147	4	2	4	4	2	2	2	3	5	4	2	4	-	4	3	3	1	1 3	59 47
Q from P146 Q from P149 Q from P150	3	2	4	1	3 2 4	3 2 3	3 2 2	3	4	4	3	4	- 4	5	3 4 2	4	1 3	-	48
Q from P151 Q from P152	2	0	0	0	4	4	3	1 2	3	3	1 3	3	-	4	2	2	3	-	35 43
Q from P153 Q from P154	3	2	4	3	2	2	2	4	4	5	2	3	4	3	2	3 4	3	0	51 50
Q from P155 Q from P156	4	3	3	4	2	2	2	3	4	3	4	4	2	4	3	4	1 2		53
Q from P157 Q from P158 Q from P159	4 4 3	4 5 2	4 5 5	5	2 2	2	4 2 3	4	4	4	3 2 3	4	2	4	4	4	2	0	60 46
Q from P160 Q from P161	3	2	5	3	2	2	2	3	3	- 2	3	4	-	3	3	5	2		25 53
Q from P162 Q from P163	3	2	4	3 4	4	3	2	3	3	5	2	3	4	3	2	4	3	1	54 60
Q from P164 Q from P165	2 4 7	2	2 5 7	3	2	3	3	2	3	4	2	4	3	4	4	4	2	3	45 58 57
Q from P167 Q from P168	4	3	4	3	2	1 3	1 2	3	3	3	1 2	4	2 4	4	3	4	2	4	51
Q from P169 Q from P170	4	2	2	2	4	3	3	2	3	3	1	3	1	4	- 2	3	-	4	43 32
Q from P171 Q from P172	3 5	3	4	2	3	2	- 2	3	3	3	3	3	3	3	3 5	3	2	3	49 60
Q from P173 Q from P174 Q from P175	4	4		5	2	2 3	2 3	4	5	3	2	1 3		4	4	4 3	1 2	2	58 39 55
Q from P176 Q from P177	4	- 2	3	2	4	4	4	2	3	3	1 2	3	-	2	2	3	3	2	42
Q from P178 Q from P179	3	2	5	4	2	2	2	4	5	4	4	2	3	4	2	4	3	0	55 50
each question Variance	4	3	3 1.26	4	2	2	2	3	4	4	3	4	3	4	3 0.73	4	1	2	53
Appendix I Questionnaire Validation Survey

Development of a Framework to Minimise Unfair Payment Practices in the UK construction

Industry

The aim of the study is to develop a theoretical framework to minimise unfair payment practices to subcontractors and suppliers in the UK construction industry. Recent UK government report identified late payment, "pay when paid" culture, withholding retentions, disparagingly rate of items, imposition of rates, and exclusion of provisional remedy as common unfair payment practices bedevilling the UK construction industry.

Objectives

The study objectives are:

- 1. To evaluate robustness of payment procedures in various standard forms of contract used in the UK construction industry.
- 2. To determine the scale of unfair payment practices between tiers of construction clients.
- 3. To identify contemporary causes of unfair payment in the UK construction industry.
- 4. To ascertain if unfair payment practices influence contractors' business performance.
- 5. To develop a theoretical framework that will help minimise unfair payment practices in the UK construction industry.

Brief insight of research methods

This study adopted concurrent mixed methods design for better understanding of the research problem. Total of 179 structured questionnaires were considered for quantitative data analysis and 19 interviews were conducted with seasoned stakeholders including clients, contractors, quantity surveyors, commercial managers, contract managers, commercial lawyers, managing directors of construction firms and business development manager. Furthermore, archival data and case studies were obtained from reliable sources such as Build UK and Companies House website concerning payment and business performance for both clients and contractors that are doing business in the UK.

Summary of findings to the Study

Key findings from the study show that Tier 2 clients' (main contractors) accounted for 82% of unfair payments practices, while 13% and 5% cases were linked to Tier 3 and Tier 1 clients respectively. Indeed Tier 2 has and does exert strong commercial influence over payments to construction supply chain. Other findings reveal that the industry is beset with host of unfair practices that range from late payments, withholding of retentions, pay when paid, cowboy bullying, spurious deductions, discounts for prompt payments, supply chain finance, pay to stay- fees and etc

The study also discovered that payment provisions in standard forms of contract are often ignored. In most cases the immediate effect of unfair payment practices on contractors is that it retards their cash flow; with resultant consequence on financial performance. Generically, the effect of unfair payments practices on construction supply chain is like a snowball effect; with direct consequence on business profit margin, incessant insolvencies and indirect effects leading to mental illness, stress, business reputational damage, etc.

Subsequently, the study developed a framework to minimise unfair payments practices to construction supply chain. The framework has potentials to encourage prompt payment of invoices from Tier 1main clients (private/public) to subcontractors and other suppliers by reducing excessive

payment bureaucracy and dominance of main contractors. Figure 1 illustrate proposed framework designed to ring-fenced project account; to be backed by a charter and protected by legislation. The project account is to be controlled by a Special purpose payment system regulator (SPPSR)



Figure 1: Proposed Framework to Minimise Unfair Payment Practices in the UK construction Industry

Key:

SPSSR- Special Purpose Payment System Regulator

Tier 1 Client – Public / private

Tier 2 Client – Main contractors

Summary of framework:

- The SPPSR need to be independent and their key role is to monitor and facilitate payments, rather than being cascaded by main contractor to subcontractors and supply chain.
- The process flow for payment application from submission of invoices by subcontractors/supplier down to payment is shown by red arrow.
- All submitted invoices for work performed shall be electronic; Tier 2 clients MUST confirm receipt, approval or rejection within seven days.
- When Tier 2 approved subcontractors invoices SPSSR will disburse payment to subcontractors and suppliers with two weeks.
- Legal and financial backup by a jurisdiction or act of parliament is required to oversee and secure prompt payment for the framework to function optimally.
- The proposed framework is similar to the non-for-profit Tenancy Deposit Scheme (TDS) managed by National Housing Association and regulated by Housing Act 2004. Detailed explanation of the framework are presented in chapter 7 of the study.

Questionnaire for framework Validation-Academic

The purpose of questionnaire below is to validate developed framework illustrated above.

In the context of attached the study aim, objectives and findings stated above please read each question carefully and put a tick (v) on the column that best describes your agreement with these statements below:

		Very strongly agree	Strongly agree	Agree	Disagree	Strongly disagree	Very strongly disagree	Unsure
1	Overall intention of the framework is good							
2	The framework has potentials to minimise unfair practices to subcontractors and suppliers							
	With appropriate legislative backing the framework; has potentials to minimise							
3	unfair payment practices							
	The framework provide insight concerning alternative payment methods to							
4	subcontractors and supplier in the construction industry							
_	If implemented properly the framework has potentials to enhance fair							
5	payment practices.							
6	The framework presents a new concept to paying construction supply chain							
7	The framework is logical							
8	The framework has potentials to minimise other unfair payment practices							
~	The framework has potentials to speed up prompt payment of invoices to							
9	subcontractors and suppliers within two weeks.							
10	As a construction professional I will recommend the framework to others							
11	The framework is capable of minimising late payment culture in the UK							
12	The framework can encouragement fair payment practices in construction							
13	The framework is too complicated and I will not recommend it to others.							
14	Perhaps Tier 2 clients will not sign up to use the framework							
15	The framework layout is easy to understand							
· · ·	Please use spaces below to express your views about the study key findir	ngs ar	d the	e dev	elope	ed fra	mewoi	·k:
	Thank you!							